

- Page intentionally left blank -

Niels Franssen

4134834 n.franssen@tudelft.nl Oost-Indië Plaats 2, 2611BR, Delft +31646685184

Graduation information



P5 report	
11/5/2019	
First mentor:	Dr. ir. L.H.M.J. (Louis) Lousberg
Second mentor:	Dr. ir. A. (Alexander) Koutamanis
Third mentor:	Dr. ir. R. (Ruud) Binnekamp
External examiner:	Dr. ir. A. (Arjen) Meijer
Scientific domain:	Design & Construction Management
Track:	Management in the Built Environment
MSc:	Architecture, Urbanism and Building Sciences
Faculty:	Faculty of Architecture and the Built Environment, Delft University of Technology
Graduation Company:	Alba Concepts
	r
First internship mentor:	J. (Jim) Teunizen
Second internship mentor:	J. (Jip) van Grinsven





PREFACE

The circular economy is one of the hottest topics in the current construction industry. In my time as a student at the Delft University of Technology I noticed the growth of the topic circularity within the curriculum, in both the design focused bachelor of the Faculty of Architecture and the Built Environment, but also within the master track Management in the Built Environment. I was privileged to execute a research concerning circularity at Alba Concepts that gave me the opportunity to combine the research field with practice. Even though circularity is developed into a buzzword, my graduation project has reached a state where circularity within the construction sector has been made tangible. As a researcher, I have tried to be a part of the development of the construction industry, to help reaching the Sustainable Development Goals concerning circular economy, determined by the Paris Agreement. With the help of many people, it resulted in this graduation thesis:

Urban mining in design and construction processes; a study on implementing urban mining of existing building elements in the local housing for improving the construction industry towards a circular economy.

After a period of 8 years, my time as a student at the Faculty of Architecture in the Built Environment comes to an end. It has been an amazing journey, a period with incredible experiences and so many people I have met and will stay in touch with in the future. During my bachelors I took a gap year to become the chairman of the board of the study association D.B.S.G. Stylos, what I experienced as one the most important years of my personal development. It made me realize my fascination within the world of architecture is not within designing, but collaborating with people to get projects realized. This was my main reason to start the Management in the Built Environment track. During my master, collaboration and project management kept me interested. This made me decide to finalize my time as a student with a research that could connect me to practice and in addition would give me the opportunity to develop a tool, which potentially could make an impact within the construction industry.

The last year has been a roller-coaster, defined by the many people that I would like to thank for their help. First I would like to thank my graduation supervisors. I would like to thank my first mentor Louis Lousberg for always being helpful, critical and to the point. Because of Louis I was able to start my graduation earlier compared to my fellow students. I would like to thank Alexander Koutamanis for the incredible amount of background information and knowledge in the topics of urban mining and circularity, which always gave me an insight from a different perspective. I would also like to thank Ruud Binnekamp, for helping me developing and construction the operational model, but also for the fun and relaxed discussions that we had at the Civil Engineering Faculty. I could have never completed this thesis without my mentors.

The start of the graduation took place at Royal HaskoningDHV, where I had the possibility to join a pilot of the Joint Interdisciplinary Project (JIP), in which the TU Delft collaborated with external companies from practice. I would like to thank my mentor Koen van Viegen and my fellow students Brian Reinders and Pratul Nema for the development of the research questions. I would like to thank Birgit de Bruin for connecting me to the JIP and Vincent Gruis, Fred Hobma, Hans Hellendoorn and Roberto Cavallo for helping me organizing the JIP in a way it could be implemented within my graduation. I hope I can repay the effort through sharing my experiences with other students.

I would like to thank all my colleagues at Alba Concepts for all the moments they helped me develop my research. Next to the interesting discussions, I was privileged to join many group activities, drinks and events together. I would especially like to thank my main supervisors Jim Teunizen and Jip van Grinsven, next to colleagues Stijn van Enckevort, Sven Bögels and Mike van Vliet, which gave me insight in the practice field of reusable building elements costs and factors. I hope I have repaid their effort by being a useful colleague, for example by giving the new office in 's Hertogenbosch a green touch by providing it with all the new plants.

During my graduation, I gained a lot of support from my colleagues at the communication department at the Faculty of Architecture and the Built Environment. I would like to thank Sue van de Giessen, Annet Zorge, Lotte Dijkstra and Nicolet Mansveld-Olsthoorn for always listening to my progress and for providing a personal working spot for me at their office. I would also like to thank Linda van Keeken for her support during this year and Arnout Sabbe for helping me with the collaboration with the Cinderela project.

Finally I would like to thank all my fellow students from Architecture, my soccer team, my neighbours, other friends and my parents for supporting me throughout this project, taking my mind of off the graduation at times as well as helping me remember what is important in life. I especially want to thank Sophie van Bochove for sticking up with me the entire second half of the project, helping me finalizing my master thesis and always being there for me.

My time as a student has come to an end. It has been a wonderful student career and I look forward to what the future will bring me. I hope you will enjoy reading my master thesis.

Niels Franssen Delft, 05 November 2019

ABSTRACT

In most of the current starting construction projects, most stakeholders use virgin materials for building elements. Considering climate change, building materials are getting scarce, it is necessary to reduce the number of virgin building materials and improve circularity in the construction industry. The problem is the absence of a middle man that could research the data of existing usable materials and connect this information to stakeholders of starting dwelling projects. This raises the research question: How can an operational model link the supply of existing building materials with the demand for new construction projects in order to reduce the use of virgin materials and thereby improve circularity in the construction industry? The goal of this research is to provide a useful tool for the construction industry to join the transition towards a circular economy. Based on three levels of scale, an operational model is developed that gives a comparison between the materials costs of existing building elements and virgin building elements for a starting dwelling project. This comparison results in insight and an overview for clients, such as municipalities or housing corporations into the costs of potential dwellings built with existing building materials. The operational tool also gives insight in whether or not reusable building materials provide a feasible business case, considering a framework based on the clients input.

Keywords

Urban mining, existing building materials, virgin building materials, operational model, projectbased, region-based, enterprise-based, feasibility

SUMMARY

I. INTRODUCTION

In the current starting construction projects, most of the materials used for the dwellings are gathered from processing factories. The use of virgin materials is common among regular construction projects and has been used in most of the projects in the past. Considering climate change, building materials are getting scarce (Kleinjan, 2017), it is necessary to reduce the number of virgin building materials and improve circularity in the construction industry. This thesis researches the possibilities of increasing the circularity, by reducing the usage of virgin building materials in construction projects and help the construction industry with the transition towards a circular economy (CE). The building materials that can be gathered by urban mining (UM) must be considered while initiating new projects. The feasibility of the process by adopting reusable materials into the design compared to virgin building elements is also of importance. The market has to be researched, whether or not there is a demand for this type of building materials or this type of system for contractors. It should be researched whether or not it is feasibly possible to adapt reusing materials to make it more costefficient, or more interesting according to the Sustainable Development Goals (SDG's) of the United Nations (Davy & Guar, 2018) to receive funding. Creating a business case from the perspective of the middle man should be taken into account as well. This is just a small list of the topics within this research, which in overall tries to clarify the challenges and research how the transition towards a CE could be stimulated in the construction sector.

II. RESEARCH QUESTION

The main objective of this thesis is to support the construction industry with the transition towards a circular economy, by decreasing the usage of virgin building materials. This raises the following main research question:

How can an operational model link the supply of existing building materials with the demand for new construction projects in order to reduce the use of virgin materials and thereby improve circularity in the construction industry?

Figure 0.1 shows a conceptual model of the research at hand. The term UM is supported by the collection of the data. The data arises from the supply that consists of case studies coming from a project-, enterprise- or region-based level context. This supply provides neglected dwellings that



consists of building elements. These buildings elements are the reusable materials that trough UM will be used for new construction projects. A comparison of feasibility shows if the circularity of the building sector could be improved.

This conceptual model, combined with the main research question provides the following subquestions for this thesis:

- *I.* What specifications are necessary for a case study to be implemented as input for the operational model?
- 2. What is the current demand in the construction industry for new housing projects?
- *3.* How can a comparison between the overview of the cost of virgin and existing building elements for new dwellings provide a feasible business opportunity?

III. METHODOLOGY

Figure 0.2 shows an overview of the setup of the research. This is the starting point of the exploratory research, where the research questions have been developed during the orientating period of the Joint Interdisciplinary Project (JIP). The setup of the research is separated into two different research paths. The left path focussed on the supply side, which consists datasets divided into three levels of scale: project-based, region-based or enterprise-based. The right path of the research setup model focussed on the demand side of the research question that exists of the conducted literature study and expert discussions with stakeholders from practice. It also discovered the correct demand for the program requirements. These two separate paths provided input for the operational model. Eventually, the results of the operational model give feedback to the research questions.



Figure 0.2 Research setup (own ill.)

IV. OPERATIONAL MODEL

Besides the literature research and the expert discussions, a part of this graduation research contains designing an operational model. Operational research deals with operation-related problems, specific research that starts with a typical 'how to' question. It deals with design problems, research about new artefacts that do not yet exist. The goal is to improve a certain subject or topic through designing the operational model and improving the model by running it, again and again, creating a detailed final design (Barendse, Binnekamp, de Graaf, van Gunsteren, & van Loon, 2012). This research provides a model, which uses linear programming to determine what the most optimal solution is, concerning the research goal. Within this research, the model calculates what combinations of dwellings maximizes the use of already available materials, directly minimizing the use of virgin building materials (Binnekamp, 2018). The difference in costs between the dwellings designed with reusable building materials is compared with dwellings built

completely with virgin materials. Eventually, this generates results, creating overview and an advice for stakeholders whether or not executing reusing materials will improve the circularity of their project. The model is adaptable to different contexts, by upscaling or downscaling the setup and adjusting the variables and constraints. A schematic overview of the model is presented in Figure 0.3.



V. SUPPLY

Datasets from case studies are used as input for the operational model, to determine the supply side of the model and making the connection to real-life problems. The input, consisting of datasets for the model, are provided by multiple different companies. These datasets are divided over the different levels of scale and collected within the schematic overview. The different levels of scale provided the following cases:

- Project-based:
 - o Bo-Ex apartments Utrecht
 - o Project Doen! Stationsarea Arnhem
 - o Stadswinkel Tilburg
 - o De Alliantie
- Region-based:
 - o EME-platform
- Enterprise-based
 - o Cinderela
 - o Cirdax / Cirkelstad / New Horizon

The different cases that are researched as input to test hypothetical operational models turned out to be a bit of a struggle. Most of the potential partners lacked corporation, due to privacy regulations. It resulted in a much smaller amount of datasets that were possible to use for the operational model than expected. The consequences were that this research changed the focus towards a realistic output instead of the input. The operational model should be developed in a sense that it is ready to operate, which means that if there are datasets becoming available to use, the model should be designed in a way that data could be directly implemented.

VI. Demand

The Rijksinstituut voor Volksgezondheid en Milieu (2016) stated the challenge that it is important to design buildings in such way that all of the materials in them are suitable for high-quality reuse. However, the long life of building structures – 50 to 100 years – makes it difficult to determine how the materials will be dealt with in several decades' time. So it makes sense to analyse the current standard dwellings and determine which materials have the potential to be reused. This will be input for the operational model.

The RVO (Rijksdienst voor Ondernemend Nederland) provided, in collaboration with Royal HaskoningDHV, six different types of standardized dwellings that nationwide can be used as reference dwellings. These documents exist of all the practical information about the dwellings. Within these six types of standardized dwellings, three have been chosen to be analysed, which are the standard row-house, the semi-detached dwelling and the completely detached dwelling. The dwellings are analysed on a set of topics related to the materials used within the building, which are of use to determine the right quantities of materials which have the possibility to be reused within the dwellings. Dwellings contain multiple building materials and elements, but not every element is fit for reuse. There are certain factors that determine whether or not materials could have a second life cycle that directly reacts to the detachability of building materials, which influences the quality of the material when it is detached or wrecked. The factors that decide how much materials are being influenced by accessibility of connection, detachability of connection and other specifications for materials are designed by Durmisevic (2006).

A set of six different types of materials show the highest possibility of reusability. The choice on these six types of materials is based on an analysis of the present building materials in dwellings in collaboration with a colleague of Alba Concepts. The type of materials that are used for the elements within the dwellings are based on existing details, which are based on the 'SBR-Referentiedetails' for dwellings (Nieman, 2015). These details consist of exact information on what type of material is used for every particular part of every element, which directly connects to the following six different materials:

- Floors: Channel plate floor on the ground level (150mm thickness) and the first or second level (200mm thickness);
- Exterior wall: Prefab concrete walls (Residential partitioning wall) and Clickbricks (Façade);
- Interior wall: Prefab panels (Ytong or Faay panels);
- **Roof:** Ceramic roof tiles;
- Doors: Corridor doors, Façade doors (front) and French door (garden);
- Windows: Meranti frame.

VII. Feasibility

The clients that are interested in reusing building elements should have a clear view on the cost overview of virgin and reusable materials. This comparison gives insight whether or not using reusable building materials is a feasible business case. Figure 0.4 shows the entire overview of the investment costs for a construction project (Gemeente Amsterdam et al., 2017). In this research, the calculation focusses only on the material costs of the MAMO overview (MAMO: Materiaal, Arbeid, Materieel en Onder aanneming).



Investment costs

Figure 0.4 Investment costs of a construction project (Gemeente Amsterdam et al., 2017)

The virgin building material costs are determined, based on practice and prices used by contractors in current traditional projects. Next, based on input generated from the current innovation on circular economy of the construction industry and discussion sessions with experts, a substantiated calculation on the costs of reused building materials is presented. This is shown in Figure 0.5. The costs for reusable materials per square meters $[€/m^2]$ consists of the sum of the deconstruction factor, the transportation factor and the material validation factors, which are the external influences multiplied by the life span and the LI (losmaakbaarheidsindex), developed by Alba Concepts.

An important notice is that this formula is based on the assumption and interpretation of different factors, which are coming from experts that are active in the field of circular construction industry, but the formulas haven't been applied to cases yet. Every single part of the calculation should be tested and substantiated before the formula can be applied in practice. When it is eventually being used, the test cases show which factors have the most influence in the outcome of the costs, also directly putting the reliability of this factor in perspective.



Reusable materials total costs / m2

VIII. Results

The gathering of information on the supply and demand side of the research setup resulted in the execution of a hypothetical project-based model, which generates the output of an example operational model. The results consists of the following subjects:

- Reused building materials.
- Reference dwellings to be built.
- Financial comparison between reused and virgin building materials.

The results of the comparison of costs in the project-based model are visible in Table 0.1. The table shows that, based on the number of reusable materials available, the three dwelling types as reference, the distance of the sources of the materials, the budget and the demand of dwellings by the client, it would be 14% cheaper to use reusable building materials compared to virgin building materials. However, a lot of different assumptions are made to achieve this result and the framework used for the operational model creates a large number of interesting topics for discussion.

IX. Discussion

The results of the project-based level calculation show that it is, in fact, cheaper to use reusable materials for new dwellings to develop. But, there are many factors that should be taken into consideration when stating the fact that reusable building materials are cheaper than virgin materials. The construction industry is far more complex than the costs for just the building materials. To transform the entire structure of the construction industry, far more extensive research should take place and even when it is researched, due to the large scale and the many stakeholders within the construction industry, it would take years, even decades to transform the system. However, for this research, certain assumptions made within the scope of the reuse of building materials are discussed. Four main topics of discussion are provided, each related to the scope of the research and substantiated with arguments to take into account when stating the results mentioned in the former chapter.



Table 0.1 Project-based model output (own table)

There is a lack of transparency in the availability of reusable building materials. Even though there is so much material out there, ready to be reused, there is no clear overview of what the exact numbers, dimensions, quality and other specifications are. The problem is the competition between the different suppliers, which creates a barrier between demand and supply. The irony is that multiple organisations who are leaders in this transition are all about collaborating, sharing, transparency and other motivational speeches, but in reality, the construction industry is far from an integrated collaborating structure.



Transparancy of datasets



Validation of feasibility

Involvement of stakeholders Figure 0.6 Discussion topics (own ill.)

The second topic of discussion is related to the actual materials that will be reused. The six types of building elements are divided into a set of materials which show potential to be reused or related to reference projects that showed whether or not the material could be reused. However, the number of materials are chosen within a certain scope, while in fact much more of the building elements of dwellings should be reused. In this research, the final calculation shows the comparison between the costs of the reusable and virgin building materials, which can be reused. However, for an entire dwelling to be realized, much more materials are necessary. Looking at the layers of Brand (1994), the other layers, such as 'services', 'space plan' and 'stuff' are not taken into account in this calculation. If more information and techniques are used to deconstruct and preserve the other building elements, the list of to be used materials will be larger, thus creating more dwellings with almost completely reused materials.

To calculate an entire project of creating dwellings with reused building materials, much more aspects of costs should be taken into account to create a complete, well-substantiated comparison of costs and the entire budget. Also, the virgin material costs overview is based on sources online, but these source of costs is different for each material. The formula designed for reusable materials is probably the most interesting point of discussion. This calculation is not yet being used in practice and is an interpretation of different investigations of commodity valuations and factors of detachability and deconstruction. To actually validate the formulas, real-life projects with correct input are necessary to see if it would generate a realistic outcome.

Projects such as dwellings constructed with reusable structural materials will never take off unless there is a significant amount of stakeholders involved within the process. If the initiative comes from a municipal department, partners in the possession of datasets of building materials are necessary, but also companies who are willing to take on construction projects with reusable materials. Without multiple stakeholders with ambitions for circularity who join the project during the initiative phase, the goals set to be achieved will be harder to reach.

X. Conclusion

The main research question of this research is:

How can an operational model link the supply of existing building materials with the demand for new construction projects in order to reduce the use of virgin materials and thereby improve circularity in the construction industry?

The main conclusion:

By developing an operational model which calculates the **difference in costs** between the reused and virgin building materials, showing that the reusable materials are **cheaper** compared to the virgin materials, based on the given **framework**.

This question generated the research goal of reducing the usage of virgin building materials in starting construction projects. Looking at the results of the project-based level operational model, it shows in the comparison in feasibility that the costs for reusable materials are cheaper compared to the virgin building materials. This resulted in an overview and insight for the hypothetical client of the project, which shows it is more interesting to invest in reusable materials compared to virgin materials. This could lead to a reduction in the usage of virgin building materials, which was the goal of the main research. Also, the final deliverable, which is the operational model, proves its functionality and showed insight within the solution space. It is developed with the possibility to implement different datasets, multiple variables and other factors. The qualities that the operational model presents show the value of the model for the construction industry.

With presenting this conclusion, a set of comments should be made to substantiate the statement listed above and the quality's that are related to the operational model. In the discussion chapter, the framework in which the results were executed is based on a set of assumptions that lead to a positive outcome of the comparison. This means that every context leads to a different result, so the numbers presented are purely to provide insight for companies or municipalities who would consider reusing building materials. To execute the project, much more research and additional variables are necessary to create a closed framework with a realistic outcome.

TABLE OF CONTENTS

PREFA	CE	3
ABSTR	RACT	5
SUMM	ARY	6
TABLE	OF CONTENTS	16
LIST O	F FIGURES	20
LIST O	F TABLES	22
1. INTE	RODUCTION	24
1.1 1.1.1 1.1.2 1.1.3 1.1.4 1.1.5 1.2 1.2.1 1.2.3 1 1.2.4 1.2.5 1.2.6	Research BackgroundThe current state of the building sectorCircular economyJoint Interdisciplinary ProjectEnergy transitionSustainable Development GoalsProblem statementSocietal relevanceScientific relevanceResearch question.2.3.1Sub-question 1.2.3.2Sub-question 2.2.3.3Sub-question 3Research goalResearch deliverablesResearch organisation	25 25 26 27 30 31 33 34 35 36 37 37 37 38 38 39 39
1.3	Conclusion	40
2. LITE	RATURE STUDY	42
2.1 2.1.1 2.1.2 2.1.3	Urban Mining Definition Challenges Conclusion	43 43 44 44
2.2 2.3	Material flows in the CE Level of scale	45 47
		47

2.3.1	Project-based level	47
2.3.2	Region-based level	49
2.3.3	Enterprise-based level	50
2.4	Material management systems	50
2.4.1	Missing of the 'man in the middle'	51
2.4.2	Madaster	52
2.4.3	Cirdax – Re-Use Materials	52
2.4.4	Excess Material Exchange (EME)	53
2.4.5	Conclusion	54
2.5	Definition of building elements	55
2.5.1	Building levels	55
2.5.2	Building Circularity Index – practice model	57
2.5.3	Conclusion	59
2.6	Housing demand in the Netherlands	60
2.6.1	Growth of population in the Netherlands	60
2.6.2	Growth of dwellings prognosis	62
2.7	Conclusion	63
3. METH	IODOLOGY	66
31	Research setun	67
3.1.1	Introduction	67
3.1.2	Exploratory research	69
3.1.3	Research through design	70
3.1.4	Study limitations	73
3.2	Operational model	73
3.2.1	Setup	74
3.2.2	Input specifications	76
3.2	2.2.1 Design space	76
3.2	2.2.2 Exogenous and endogenous variables	77
3.2	2.2.3 Constraints, goals and objectives	77
3.2.3	Model specifications	78
3.2	2.3.1 Uperations research formula	78
J.,	2.3.2 Linear programming general mathematical model	8U
3./ 22/	2.3.3 What SBESL! add-in for Microsoft Excet	0U 01
ວ.2.4 ຊໍ	2 / 1 Stakabaldare	01 82
3.	2.4.1 Statemoties 2.4.2 Comparison of feasibility factor	82
0.7		02
3.3	Supply	83
J.J. J.J.	Dalasels Sliuliule Projant.hasad casas	04 07
J.J.Z יך	1 10jete Jaseb Lases 3 2 1 Ro-Fy anartmente Iltrecht	00 Q7
3.	3 2 7 Project Doenl Stationarea Arnhem	07 87
3.1	3.2.3 Stadswinkel Tilburg	88
3.3	3.2.4 De Alliantie	89

3.3.3 3.3.4 3.3 3.3.4 3.3 3.3.5	Region-based cases 3.3.1 EME-Platform Enterprise-based cases 3.4.1 Cinderela 3.4.2 Cirdax / Cirkelstad / New Horizon Conclusion	89 89 91 91 92 93
3.4 3.4.1 3.4.2 3.4.3	Demand Dwelling prototypes references RVO Reusable building materials Participating stakeholders	93 94 96 98
3.5 3.5.1 3.5.2 3.5 3.5 3.5 3.5	FeasibilityCosts virgin building materialsCosts reusable building materials5.2.1Deconstruction5.2.2Transportation5.2.3Material validation5.2.4Conclusion	99 100 101 102 103 104 106
3.6 3.6.1 3.6.2 3.6.3	Execution Project-based model Region-based model Enterprise-based model	106 107 107 107
4. FINDI	INGS	110
		110
4.1 .1 4.1.2 4.1.3	Main findings Materials Dwellings Feasibility	111 111 112 113
4.1 4.1.1 4.1.2 4.1.3 5. DISC	Main findings Materials Dwellings Feasibility USSION	111 111 112 113 118
4.1 4.1.1 4.1.2 4.1.3 5. DISC 5.1	Main findings Materials Dwellings Feasibility USSION Transparency of datasets	111 111 112 113 118 119
4.1 4.1.2 4.1.3 5. DISC 5.1 5.2	Main findings Materials Dwellings Feasibility USSION Transparency of datasets Reusability of building materials	111 111 112 113 118 119 119
4.1 4.1.2 4.1.3 5. DISC 5.1 5.2 5.3	Main findings Materials Dwellings Feasibility USSION Transparency of datasets Reusability of building materials Validation of feasibility	111 111 112 113 118 119 119 120
4.1 4.1.2 4.1.3 5. DISC 5.1 5.2 5.3 5.4	Main findings Materials Dwellings Feasibility USSION Transparency of datasets Reusability of building materials Validation of feasibility Involvement stakeholders	111 111 112 113 118 119 119 120 122
4.1 4.1.2 4.1.3 5. DISC 5.1 5.2 5.3 5.4 6. CONC	Main findings Materials Dwellings Feasibility USSION Transparency of datasets Reusability of building materials Validation of feasibility Involvement stakeholders	111 111 112 113 118 119 119 120 122 124
4.1 4.1.2 4.1.3 5. DISC 5.1 5.2 5.3 5.4 6. CONC 6.1	Main findings Materials Dwellings Feasibility USSION Transparency of datasets Reusability of building materials Validation of feasibility Involvement stakeholders CLUSION Sub-question 1 feedback	111 111 112 113 118 119 119 120 122 122 124 125
4.1 4.1.2 4.1.3 5. DISC 5.1 5.2 5.3 5.4 6. CONC 6.1 6.2	Main findings Materials Dwellings Feasibility USSION Transparency of datasets Reusability of building materials Validation of feasibility Involvement stakeholders CLUSION Sub-question 1 feedback Sub-question 2 feedback	111 111 112 113 118 119 119 120 122 124 125 125

6.4	Main research question feedback	126
7. REC	OMMENDATIONS	130
8. EVA	LUATION	134
9. REF	LECTION	138
10. BI	BLIOGRAPHY	142
11. AP	PENDICES	150
11.1	Appendix A: Data single dwelling 'VIIA Projecten'	151
11.2 11.2 11.2 11.2 11.2 11.2	Appendix B: Operational model example.1Main clients and goals.2Mathematical elements.3Model input.4Model description.5Model output	152 152 153 154 156 157
11.3	Appendix C: Hypothetical constraints input regional-based model	158
11.4	Appendix D: Example dataset structure Alba Concepts	160
11.5	Appendix E: Overview materials standard row-house analysis	162
11.6	Appendix F: Overview materials semi-detached house analysis	163
11.7	Appendix G: Overview materials detached house analysis	164
11.8	Appendix H: Disassembly factor weights and attribute weights	165
11.9	Appendix I: Overview costs virgin building materials	166
11.10	Appendix J: Overview deconstruction / transportation costs reusable building materials	167
11.11	Appendix K: Overview of material validation reusable building materials	168
11.12	Appendix L: Project-based model input	169
11.13	Appendix M: Project-based model output	171

LIST OF FIGURES

FIGURE 0.1 CONCEPTUAL MODEL RESEARCH QUESTION (OWN ILL.)	6
FIGURE 0.2 RESEARCH SETUP (OWN ILL.)	8
FIGURE 0.3 SCHEMATIC OVERVIEW DECISION MAKING MODEL (OWN ILL.)	9
FIGURE 0.4 INVESTMENT COSTS OF A CONSTRUCTION PROJECT (GEMEENTE AMSTERDAM ET AL., 2017)	11
FIGURE 0.5 REUSABLE MATERIALS TOTAL [€/M²] (OWN ILL.)	12
FIGURE 0.6 DISCUSSION TOPICS (OWN ILL.)	14
FIGURE 1.1 COMBINED OVERVIEW OF TRANSITION LE TO CE MODELS (OWN ILL.)	25
FIGURE 1.2 DIVISION OF ELEMENTS WITHIN DESIGN PROCESS TOOL (FRANSSEN ET AL., 2018)	28
FIGURE 1.3 URBAN MINING DATA MANAGEMENT TOOL (FRANSSEN, NEMA NAGENDRA KUMAR, & REINDERS, 2018)	29
FIGURE 1.4 MATERIAL FLOW & LOGISTICS FLOW (FRANSSEN ET AL., 2018)	30
FIGURE 1.5 SUSTAINABLE DEVELOPMENT GOALS (UN, 2018)	31
FIGURE 1.6 LINK SDG'S TO CIRCULAR ECONOMY (THELEN ET AL., 2018)	33
FIGURE 1.7 PROBLEM STATEMENT (OWN ILL.)	34
FIGURE 1.8 CONCEPTUAL MODEL RESEARCH QUESTION (OWN ILL.)	37
FIGURE 2.1 LADDER VAN LANSINK (LANSINK, 1979)	45
FIGURE 2.2 MATERIAL FLOWS IN A CIRCULAR ECONOMY [SOURCE: EPEA & RETURNITY PARTNERS] (R. J. GELDERMANS, 20	16)45
FIGURE 2.3 THE CIRCULAR ECONOMY—AN INDUSTRIAL SYSTEM THAT IS RESTORATIVE BY DESIGN (EMF, 2013)	46
FIGURE 2.4 APPLYING CIRCULAR ECONOMY PRINCIPLES TO THE BUILT ENVIRONMENT (CHESHIRE, 2016)	46
FIGURE 2.5 LEVELS OF SCALE (OWN ILL.)	47
FIGURE 2.6 THE GREEN HOUSE (DUURZAAM GEBOUWD, 2018)	48
FIGURE 2.7 EARTHQUAKE DAMAGED DWELLING AT GRONINGEN (KNAG, 2016)	49
FIGURE 2.8 COFFEE LOGSTM, PRODUCT BY BIO-BEAN (HARRISON ET AL., 2016)	50
FIGURE 2.9 AIRBNB / UBER / MARKTPLAATS	51
FIGURE 2.10 MADASTER LOGO (MADASTER, 2019)	52
FIGURE 2.11 OVERVIEW CIRDAX TOOLS (RE USE MATERIALS, 2019)	53
FIGURE 2.12 EXCESS MATERIALS EXCHANGE LOGO AND MODULES (DAMEN & VAN MAAREN, 2018)	54
FIGURE 2.13 THEORY OF MATERIAL LEVELS (DURMISEVIC, 2006)	55
FIGURE 2.14 SHEARING LAYERS (BRAND, 1994)	56
FIGURE 2.15 FIXED, PARTIALLY FIXED AND OPEN STRUCTURE ON DIFFERENT BUILDING LEVELS (DURMISEVIC, 2006)	57
FIGURE 2.16 CONCEPTUAL MODEL OF THE BUILDING CIRCULARITY INDICATOR ASSESSMENT MODEL (VERBERNE, 2016)	58
FIGURE 2.17 PRODUCT CIRCULARITY INDEX & ELEMENT CIRCULARITY INDEX (ALBA CONCEPTS, 2018)	58
FIGURE 2.18 BUILDING CIRCULARITY INDEX DRIVERS & CONDITIONS (ALBA CONCEPTS, 2018)	59
FIGURE 2.19 BUILDING CIRCULARITY INDEX CALCULATION STEPS (ALBA CONCEPTS, 2018)	59
FIGURE 2.20 BEVOLKINGSGROEI PER GEMEENTE, 2015-2030 (CENTRAAL BUREAU VOOR DE STATISTIEK, 2016)	61
FIGURE 3.1 RESEARCH SETUP SMALL (OWN ILL.)	67
FIGURE 3.2 RESEARCH SETUP (OWN ILL.)	69
FIGURE 3.3 CYCLICAL PROCESS OF THE DEVELOPMENT OF IDEA'S, BASED ON VAN DEN AKKER, BANNAN, KELLY, NIEVEEN, J	AND
PLOMP (2013)	71
FIGURE 3.4 STEPS OF AN EMPIRICAL RESEARCH PROCESS (KUMAR, 2011)	72
FIGURE 3.5 STEPS OF A DESIGN PROCESS (DYM, LITTLE, & ORWIN, 2004)	72
FIGURE 3.6 SCHEMATIC OVERVIEW DECISION MAKING MODEL (OWN ILL.)	74
FIGURE 3.7 SCHEMATIC OVERVIEW THREE PHASES OF THE OPERATIONAL MODEL (OWN ILL.)	75
FIGURE 3.8 SCHEMATIC OVERVIEW INPUT (OWN ILL.)	76
FIGURE 3.9 SCHEMATIC OVERVIEW MODEL (OWN ILL.)	78
FIGURE 3.10 MATHEMATICAL OVERVIEW ENDOGENOUS VARIABLES (OWN ILL.)	79
FIGURE 3.11 OPERATIONAL MODEL SCHEMATIC REPRESENTATION 'WHAT'SBEST!' ADD-IN MICROSOFT EXCEL (OWN ILL.)	81
FIGURE 3.13 SCHEMATIC OVERVIEW OUTPUT (OWN ILL.)	82
FIGURE 3.14 CINDERELA PROJECT CONFERENCE AMSTERDAM (PRANJIĆ & MLADENOVIĆ, 2018)	83

FIGURE 3.15 CIRCULAIRE ECONOMIE FESTIVAL NIJMEGEN (VONK, DE JAGER, & DIRKX, 2019)	84
FIGURE 3.16 SHEARING LAYERS (BRAND, 1994)	84
FIGURE 3.17 DATABASE STRUCTURE (OWN ILL.)	85
FIGURE 3.18 OVERVIEW OF PROJECT-BASED CASES IN THE NETHERLANDS (OWN ILL.)	87
FIGURE 3.19 BO-EX (BO-EX, N.D.)	87
FIGURE 3.20 RIJKSOVERHEID (BOER, 2018)	87
FIGURE 3.21 W/E ADVISEURS (W/E ADVISEURS, N.D.)	88
FIGURE 3.22 DE ALLIANTIE (DE ALLIANTIE, N.D.)	89
FIGURE 3.23 ENTERPRISE-BASED REGION: AMSTERDAM METROPOLITAN AREA (OWN ILL.)	90
FIGURE 3.24 EME-PLATFORM POSSIBILITIES IN INNOVATION, BASED ON DAMEN AND VAN MAAREN (2018)	90
FIGURE 3.25 CINDERELA (ZAG, 2016)	91
FIGURE 3.26 CINDERELA H2020 COMPLETE CDW FLOW ORIGINATING FROM THE AMA IN 2016 (B. GELDERMANS ET AL., 20	19)92
FIGURE 3.29 CIRDAX (RE USE MATERIALS, 2019)	93
FIGURE 3.29 CIRKELSTAD (BÜCH ET AL., N.D.)	93
FIGURE 3.29 NEW HORIZON (VERKUIJLEN, 2015)	93
FIGURE 3.30 SCHEMATIC DWELLING ANALYSIS OVERVIEW OF EXCEL MODEL (OWN ILL.)	94
FIGURE 3.31 STANDARD-HOUSE DWELLING, SEMI-DETACHED HOUSE, DETACHED HOUSE (KRIJNEN, 2015D, 2015E, 2015F)	95
FIGURE 3.32 SHEARING LAYERS (BRAND, 1994)	96
FIGURE 3.33 SIX DIFFERENT BUILDING ELEMENTS WITH HIGH POTENTIAL FOR REUSABILITY (OWN ILL.)	97
FIGURE 3.34 INVESTMENT COSTS OF A CONSTRUCTION PROJECT (GEMEENTE AMSTERDAM ET AL., 2017)	99
FIGURE 3.35 DIRECT CONSTRUCTION COSTS (MAMO) (GEMEENTE AMSTERDAM, CONCEPTS, & COPPER8, 2017)	100
FIGURE 3.36 COBOUW (COBOUW, N.D.)	100
FIGURE 3.37 VIRGIN BUILDING MATERIALS COSTS SCHEMATIC OVERVIEW OF EXCEL TABLE LOCATED IN APPENDIX 11.9 (0)	WN ILL.)
	101
FIGURE 3.38 DISCUSSION SESSION EXPERTS ALBA CONCEPTS 26-08-2019 (OWN IMAGE)	102
FIGURE 3.39 DECONSTRUCTION COSTS FORMULA, TOTAL DECONSTRUCTION COSTS FORMULA (MADASTER ET AL., 2018)	103
FIGURE 3.40 ADDITIONAL TRANSPORTATION COSTS FORMULA (MADASTER, TNO, & ALBA CONCEPTS, 2018)	104
FIGURE 3.41 MATERIAL VALIDATION FACTOR REUSABLE BUILDING MATERIALS (OWN ILL.)	104
FIGURE 3.42 VALUATION MATERIALS OVER TIME (VAN ENCKEVORT, 2019)	105
FIGURE 3.43 REUSABLE MATERIALS TOTAL [€/M²] (OWN ILL.)	106
FIGURE 4.1 OUTPUT SPECIFICATIONS OPERATIONAL MODEL (OWN ILL.)	111
FIGURE 4.2 OUTPUT AMOUNT OF DWELLINGS (OWN ILL.)	113
FIGURE 5.1 DISCUSSION TOPICS (OWN ILL.)	118
FIGURE 5.2 INVESTMENT COSTS OF A CONSTRUCTION PROJECT (GEMEENTE AMSTERDAM ET AL., 2017)	121
FIGURE 11.1 DATA SINGLE DWELLING (HASKONINGDHV, 2018)	151
FIGURE 11.2 MAIN STAKEHOLDERS, ROYAL HASKUNINGDHV (CONTRACTOR) & MUNICIPALITY OF GRONINGEN (CLIENT)	152
FIGURE 11.3 RESEARCH GOAL (OWN ILL.).	153
FIGURE 11.4 BUILDING ELEMENTS (OWN ILL.)	154
FIGURE 11.5 RECUVERED MATERIALS (OWN ILL.)	154
FIGURE 11.6 DWELLING TYPES AND INFORMATION (FUNDA).	154
FIGURE 11.7 CONCEPTUAL DEFINITION OF DWELLING DESIGN (OWN ILL.)	155

LIST OF TABLES

TABLE 0.1 PROJECT-BASED MODEL OUTPUT (OWN TABLE)	. 13
TABLE 2.1 PROGNOSIS OF THE HOUSEHOLD SIZE IN THE NETHERLANDS IN THE PERIOD 2017 - 2030 (RIJKSOVERHEID, 2018) 60
TABLE 2.2 DEVELOPMENT OF HOUSEHOLDS IN SIZE AND AGE, IN THE PERIOD 2017 - 2030 (RIJKSOVERHEID, 2018)	. 61
TABLE 2.3 PROGNOSIS ON INCREASE IN DWELLINGS NEEDS, DWELLING STOCK AND PLAN CAPACITY PER PROVINCE IN THE	PERIOD
2017 – 2030 (RIJKSOVERHEID, 2018)	. 62
TABLE 2.4 PROGNOSIS ON DEVELOPMENT OF THE HOUSING STOCK, IN THE PERIOD 2017 – 2030 (RIJKSOVERHEID, 2018)	. 63
TABLE 4.1 OUTPUT MATERIALS PROJECT-BASED MODEL (OWN TABLE)	112
TABLE 4.2 PROJECT-BASED MODEL OUTPUT (OWN TABLE)	114
TABLE 11.1 OVERALL INPUT OPERATIONAL MODEL (OWN ILL.)	156
TABLE 11.2 OPERATIONAL MODEL EXCEL (OWN ILL.)	156
TABLE 11.3 OPERATIONAL MODEL RESULTS (OWN ILL.)	157
TABLE 11.4 GROUPS OF TYPE BUILDING ELEMENT AND MATERIAL (ROYALHASKONINGDHV, 2018)	159
TABLE 11.5 EXAMPLE DATASET STRUCTURE ALBA CONCEPTS	161
TABLE 11.6 OVERVIEW MATERIALS STANDARD ROW-HOUSE ANALYSIS (OWN TABLE)	162
TABLE 11.7 OVERVIEW MATERIALS SEMI-DETACHED HOUSE ANALYSIS (OWN TABLE)	163
TABLE 11.8 OVERVIEW MATERIALS DETACHED HOUSE ANALYSIS (OWN TABLE)	164
TABLE 11.9 DISASSEMBLY FACTOR WEIGHTS AND ATTRIBUTE WEIGHTS ADOPTED FROM DURMISEVIC (2006) (VAN VLIET, 20	18)165
TABLE 11.10 OVERVIEW COSTS VIRGIN BUILDING MATERIALS (OWN TABLE)	166
TABLE 11.11 OVERVIEW DECONSTRUCTION / TRANSPORTATION COSTS REUSABLE BUILDING MATERIALS (OWN TABLE)	167
TABLE 11.12 OVERVIEW MATERIAL VALIDATION REUSABLE BUILDING MATERIALS (OWN TABLE)	168
TABLE 11.13 INPUT PAGE PROJECT-BASED MODEL (OWN TABLE)	169
TABLE 11.14 SOLVE PAGE INPUT PROJECT-BASED MODEL (OWN TABLE)	170
TABLE 11.15 PROJECT-BASED MODEL OUTPUT (OWN TABLE)	171
TABLE 11.16 PROJECT-BASED MODEL OUTPUT SOLVE TAB (OWN TABLE)	172



2

PART 1 INTRODUCTION

The research presented in the following paper focusses on one of the most current topics within the construction industry. The term circularity and circular economy are trending and most of the starting construction projects take action to provide a connection with circular goals. This research tries to contribute further development of the construction industry, concerning the transition towards the circular economy. For this transition to happen, the entire structure of the construction industry has to be taken into account. The large amount of different processes makes it difficult to stimulate a transition for the entire industry. This research focusses on the specific part of the construction industry which has the potential to be re-organised, concerning the usage building materials for new construction sector, due to lack of overview of information and the possibilities for implanting reusable materials. There is a missing link between information and most of the stakeholders which are active in the construction industry. By comparing the costs of virgin building materials to the reusable building materials for new construction projects, this research will try to provide insight and an overview of the possibilities for the stakeholders, promoting to reduce the usage of virgin building materials in future construction projects.

The structure of the report consists of four main chapters, starting with the introduction that presents the research by displaying background information of the main subject and the problem statement, which formulates the research questions. Following is the literature research to substantiate the research topics. In the third chapter, the methodology is explained. The research design is elaborated and the different models that are developed within the research are explained and discussed. The demands and variables of the final objectives are presented, the setup of the operational model is introduced and the input for the operational model is elaborated. The fourth chapter introduces the main findings, analyses the model and reacts to the research questions of the first chapter. The report is discussed and concluded in the fifth and the sixth chapter, reflecting on the introduction and recommending on to further developments within the field.

1.1 RESEARCH BACKGROUND

This chapter presents the background of the research, following by the problem statement. It shows the main topics of the research, concerning the construction industry, the circular economy, the case of Groningen and the energy transition. The introduction sets up a framework on which the problem statement and research questions are developed.

1.1.1 THE CURRENT STATE OF THE BUILDING SECTOR

In the building sector and environment, every active stakeholder notices there is a transition starting from linear to circular. Current trends show that there are many developments concerning sustainability taking place in multiple sectors, to create a circular economy (CE) in the Netherlands by 2050 (Rijksoverheid, 2016a). Digitalization is becoming more and more important, providing innovation, smart business models and artificial intelligence (Kurzawska, 2018). The building sector needs to take action as well and is lacking to join the innovation, even though the building sector is growing in size at unprecedented rates and it will continue to do so over the next 40 years (UN Environment and International Energy Agency, 2017). With all these future projects predicted to be down cycled, the current methods of the construction industry should be taken into consideration. At the moment, more than 90% of the world's construction projects are either over time or over budget (Flyvbjerg, 2014). With innovation in the construction industry, the number of projects being over time or over budget could be downgraded. If building methods could develop into more efficient and sustainable methods, the efficiency and circularity of the sector could grow, effectively completing more projects in time and on budget.

In most of the traditional starting design and construction housing projects, most stakeholders use virgin materials for building elements and to design and construct new dwellings. Recycling and downgrading are techniques that are already incorporated at such level, but the complete reuse of building elements and building materials could be increased. The question stakeholders should ask themselves: Is using virgin materials the most feasible option? There are a lot of building materials available in different artefacts which existences is unknown for interested stakeholders. For example, large construction and demolition companies produce a lot of waste. This is being recycled on a large scale for e.g. roads. However, building materials are hardly ever reused in the construction of new buildings (RIVM, 2016). If the reuse of building materials will be increased, it decreases the production of virgin building materials, which means that less building waste and processing system are necessary. Buildings and roads do account for an estimated 60% of the total materials used



Figure 1.1 Combined overview of transition LE to CE models (own ill.)

globally (Black, 2018) so if only a part of the materials could be collected from another source instead of virgin produced materials, it would save a huge amount of materials being down cycled to a less valuable product after demolition. This would create a reduction of materials to be down cycled, incinerated or send down to landfill, resulting in fewer volumes of waste coming from the construction industry, an improvement for the CE. Eventually, the goal for the building sector in the Netherlands towards a CE is to create more circular processes, instead of linear processes (Figure I.I). This research focusses on the transition of this process, with the goal to reduce the virgin building materials in starting design and construction process and to make the Dutch building sector join the transition towards a CE step by step.

1.1.2 CIRCULAR ECONOMY

The gap between the knowledge of available materials and the direct implementation of these materials within new projects is the challenge to face. This research reviews the possibilities of closing this gap and providing a useful tool for the construction industry to join the transition towards a CE. The term circular economy (CE) is used in multiple sectors. In the building sector CE is defined as an economic system focused on maximizing the use and reuse of products and raw materials and minimizing value destruction (RIVM, 2016). The CE is a current trend in multiple sectors. The need for a CE is coming from the Sustainable Development Goals, which are highlighted in section 1.1.5. Following the Sustainable Development Goals by the United Nations, the government of the Netherlands decided on three reasons why we need to switch towards a CE: (I) the increasing demand for raw materials, (2) dependence on other countries and (3) the impact on the earth's climate (decrease of CO₂ emissions). Moving towards a CE also offers opportunities for businesses (Rijksoverheid, n.d.-b). Although, what do we exactly mean by CE? Many different researchers made an attempt to figure out the right description for the term, but because of the broad perspective and multiple fields the term is mentioned in, it is impossible to determine the one correct description for the term, that makes it such an interesting and hot topic. For example the RIVM (2016) describes the CE as an economic system that focusses on maximizing the use and reuse of product and raw materials and minimizing value destruction. While van Hemmen (2016) states that the CE is a state of the global economy that is capable of continuous recursion. This implies infinite material productivity regarding non-regenerative materials, which intends recursive recovery of wastes and an absence of resource extraction. The focus on the CE as a regenerative system is mentioned multiple times. For example Geissdoerfer, Savaget, Bocken, and Hultink (2017) define the CE a to be regenerative, a system in which resource input and waste, emission, and energy leakage are minimized by slowing, closing, and narrowing material and energy loops, while EMF (2017) based the transition towards a CE on three principles: Design out waste and pollution, keep products and materials in use and regenerate natural systems. This can be achieved through long-lasting design, maintenance, repair, reuse, remanufacturing, refurbishing, and recycling. To achieve the transition, a redesign of the current economic system is suggested, largely based on linear resource flows, towards closed-loop resource flows that can preserve the embedded environmental and economic value in resources for as long as possible (Nußholz, 2018). The closed loop system which the CE proposes is also mentioned by EMF (2016), where products and services are traded in close loops or cycles.

It is noticeable that most of the literature provide definitions are mostly similar to each other. Reducing waste and creating loops are essential in a CE, to create as less value destruction of the materials as possible. The system should be transform into a regenerative system. The entire economy should be designed in a sense it could adapt the reusability of products and elements. This is comparable to the goals of the construction industry, creating design and construction projects, keeping the factor of reusability of the building elements in mind. Only when there is an integrated system or loop, the effects of a CE can actually benefit society. Multiple literature studies also propose the way material flows should be managed to achieve the change, in order to maintain and restore the materials throughout material banks. This sheds new light on the value of the building materials and products (R. J. Geldermans, 2016). In the research question and the problem statement chapter, it is explained how the CE relates to the main research question of this thesis.

1.1.3 JOINT INTERDISCIPLINARY PROJECT

A graduation research at the Technical University of Delft, especially at the Faculty of Architecture and the Built Environment, consists of a time period of approximately one year. The first period of this graduation research started with a separate project. This Joint Interdisciplinary Project (JIP) was initiated by a collaboration between Royal HaskoningDHV and the Technical University of Delft. The project team consisted of three students from different faculties of the TU Delft, guided by a mentor from the company Royal HaskoningDHV. The end results of the JIP were presented to multiple other teams with different assignments, presenting and collaborating together. Also, the progress for the graduation thesis, developed during the JIP, was presented at the same moment in time, as the PI presentation. The structure of the JIP consisted of a joint product goal during the IO weeks of the internship and personal research for the graduation proposal that continued when the internship was finished. The data and knowledge gathered during the internship support the further graduation thesis. The internship provided documents and data about existing dwellings in Groningen to use during the research and as a test case. During the internship, the setup of the main research question and sub-questions of the graduation project were formulated.

The JIP team consisted of the following students and mentor:

- **Brian Reinders** Student Mechanical Engineering, Faculty of Mechanical, Maritime and Materials Engineering
- **Pratul Nema** Student Construction Management and Engineering, Faculty of Civil Engineering and Geosciences
- Niels Franssen Student Management in the Built Environment, Faculty of Architecture and the Built Environment
- Koen van Viegen Company coach and mentor, a structural engineer at Royal HaskoningDHV

The case of the JIP was focused on the northern province of the Netherlands, Groningen. The Northeast part has been experiencing increased seismic activity in the recent past. This change in seismic behaviour has been attributed the gas extraction in the area. Although natural gas extraction has come to an end, the risk of a possible earthquake remains. Many of the existing buildings require to be reinforced to ensure that they can continue to be used. This reinforcement needs to be able to handle an earthquake with a magnitude of 5 on the Richter scale without showing any significant damage. Since the beginning of gas extraction by drilling activities of oil companies such as Shell and Exxon, over 1500 earthquakes have taken place in the province of Groningen (Dwarshuis, 2018). The

largest earthquake since a period of over 5 years took place on the 8th of January 2018, in the village Zeerijp. It was measured at 3,4 on the Richter scale (Leijten, 2018). Approximately 104 buildings in the province of Groningen have been destroyed by earthquakes caused by the gas extraction and 607 buildings are on the shortlist of endangered buildings (Haije, Schüren, van Sluis, & Wind, 2018). Most of the buildings are masonry structures, which led to unavoidable damage like cracks. In a certain situation, it could make more sense to rebuild the house, instead of renovating the old building. The materials of the current housing used for the rehabilitation should be recycled, reclaimed or sourced locally instead of dumping the remains. This could be achieved with a large amount of data about each of the buildings and recover the materials by urban mining.



Figure 1.2 Division of elements within design process tool (Franssen et al., 2018)

The datasets which are available within Groningen, collected by Royal HaskoningDHV, made the case interesting to use as a test for this graduation research. Together with the 2 other students, a division was designed to focus on during the process of developing the tool. These three elements were (I) information, (2) logistics, and (3) design (Figure I.2). Each student was responsible for the development of an element. The data of the dwellings collected by Royal HaskoningDHV had to be managed. The building information of the projects is analysed and the different data files were handled in programs such as Python. These types of programming language could extract datasets from IFC and JSON files, to create accessibility of the information which are stored in public databases. The building information became understandable and ready to apply within the research. The logistic element focused on an efficient operation on the transportation of materials, based on theoretical background concerning several logistic models. To create the transportation system some routing theory has been used to figure out the best possible routing for minimum travel distance, that eventually created a discrete event model to simulate the situation (Franssen et al., 2018).



Figure 1.3 Urban Mining Data Management Tool (Franssen, Nema Nagendra Kumar, & Reinders, 2018)

These elements provided the Urban Mining Data Management Tool (Figure 1.3). This tool creates a potential use for the building materials into new housing projects. The model uses one single database, created by the programming methods. This database has all the information and processes that are used to complete the stored products. From the beginning to the end, it is possible to input data. As a result, a design advice has been created. This is the function of the third element. Upcoming housing projects that could be initiated by the client, in this case, the municipality of Groningen, could use this data management tool to directly use the data of the existing buildings in their area and input preferences on their quantities, locations and dwelling types (Franssen et al., 2018).

The tool is a conceptual setup of what could be a working model. It is an interdisciplinary connected tool that could and should be extended with multiple data and tools, but within the possible time frame, we made it possible to create the tool so it can be used in a conceptual manner. Figure 1.4 shows which parts the data goes through and how the different disciplines are connected. The blue round boxes are connected to the logistic scheme, the red boxes are the flow of information, and the green circle is the design area of the scheme. It shows the route of the materials within a process from the deconstruction site, towards the depot, where it is stored. Next, the flow, if necessary, goes to the processing area, eventually returning back to the depot. When the design is created with the information of accessible materials at hand, it can be transported towards the new construction area (Franssen et al., 2018).



Figure 1.4 Material flow & Logistics flow (Franssen et al., 2018)

This project was initiated with the goal to provide a research problem to be solved within the graduation thesis. The case in Groningen showed the possibilities when a certain amount of information is available concerning building elements. The housing demand, on the other hand, is growing in the Netherlands and materials are needed to provide new dwellings. However, the gap between the information on housing demand and the availability of reusable building materials is too substantial. This gap is researched and discussed. These problems and related research goals are described in section 1.2.

1.1.4 ENERGY TRANSITION

Every new housing project that is initiated in the current time must keep the Paris Agreement in mind. It has been decided in 2015 that all the greenhouse gas emissions should be reduced to 80% in 2050, according to the low-carbon economy roadmap, through domestic reductions alone (EU, 2011). One of the overall goals is holding the increase in the global average temperature below 2 °C above pre-industrial levels and pursuing efforts to limit the temperature increase to 1.5 °C above pre-industrial levels (Fabius, 2015). The Dutch government established legal requirements among starting construction project for new dwellings, according to the climate agreement. The energy performance coefficient (EPC) of the dwellings should be equal or lower to 0,4 according to the government, which is an important step towards an (almost) complete stock of energy-neutral dwellings in 2020. This applies to the energy usage of the dwellings, which involves heating, ventilation, cooling, and lighting (Haytink & Valk, 2015). All the newly designed building, dwellings and utility construction, should meet the requirements for almost complete energy-neutral buildings (BENG), to get an environmental permit. These requirements will initiate on the first of July in 2020 (Rijksoverheid, n.d.-a).

This means many of the upcoming housing projects focus on an energy-efficient home design, designing multiple techniques to reduce as much of the emission as possible. Future dwellings will not be connected to gas in 2020, eventually with the goal to have all the dwellings cut off from gas by 2050 (Rijksoverheid, 2016b). An example that is developed, to react on the demands of the government, is to keep track of the energy usage with the Home Energy Score, developed by the U.S. Department of Energy (EnergySaver, n.d.), where each dwelling receives a score, based on its structure, heating, cooling, and hot water systems. The newly to build dwellings throughout an urban mining process should be designed of such quality that it complies with the regulations stated in the Paris Agreement. I directly combines the costs for a renovation, which would happen anyhow to answer to the new regulations. This creates possibilities to combine the use of leftover building materials with the demand for new dwellings, which could lead to a useful and achievable business case. It directly stimulates designers to innovate standardized dwellings, with the future perspective of the possibility to disassemble the buildings for future purposes. It could also provide subsidies or other financial investments from municipal organisations or the government, to support or fund a project that is in line with the Paris Agreement of 2015. These subjects are put at hand within the thesis' main research goal in chapter 1.2.3.

1.1.5 SUSTAINABLE DEVELOPMENT GOALS

In the same time period as the Paris Agreement, the United Nations set up the Sustainable Development Goals (SDG's) in 2018. These 17 SDG's are the blueprint to achieve a better and more sustainable future for all. Global challenges are all connected to the 17 goals and each of them are targeted to be achieved in 2030 (UN, 2018).



Figure 1.5 Sustainable Development Goals (UN, 2018)

The goals are connected to the main societal challenges to achieve a better future, including the building sector and the CE. Social enterprise Circle Economy, organized as a cooperative, described the connection to most of the SDG's to the CE. This has great potential to help meet global sustainability targets and the Paris Agreements goals (Thelen et al., 2018). The overview of the link between 8 of the 17 goals connected to the CE is described in Figure 1.6. The connection between these descriptions and the building sector shows a clear comparison. For example, SDG #3, good health and wellbeing, connects with the CE by avoiding and removing hazardous substances from material use cycles. Decreasing building waste in construction projects and reusing building materials in new projects is an example of how this goal could be achieved in a future CE. This could be an attempt to tackle one of the goals within the building sector. Another example is SDG #II, Sustainable cities and communities, where the CE focusses on repair and extending producer-consumer contracts. This design could be adapted in new building projects, instead of buying interior and movable products, change the system to leasing products for a certain period of time. The products will be retrieved by the owner after a certain contract, decided on the lease time. SDG #12 is also a perfect example of how the building sector and the SDG's meet. Responsible consumption and production focus on responsible care for products and that virgin materials are minimized, which is exactly the goal of this research.

The SDG's, which are set by over 190 countries, are meant to tackle climate change and should inspire businesses to act on the goals. However, the question arises how to get companies inspired to invest to join the fight against poverty, climate change and other targets (Hardcastle, 2015). It shows that only 16% of companies have explicitly assessed the implication of the SDG's on material issues (Davy & Guar, 2018). This is of course among all the companies, not only the building sector, but these companies and their effort which should be put in reaching the SDG's could be the missing link in the transition to a CE. Davy and Guar (2018) also mention 5 things that companies can do to realize the business opportunity from the SDG's:

- Assess the SDG's against company policies and practices.
- Use the SDG's to inform strategy development.
- Review the SDG's as part of target setting.
- Apply the SDG's to impact monitoring and measurement.
- Consider the SDG's as part of reporting, such as an SDG index.

The SDG's could be of stimulation for this thesis, to support certain choices made in relation to the transition towards a CE. During the answering of the main and sub-research questions, the SDG's are taken into account concerning closing the gap between the companies and the goals.

SDG		LINK TO THE CIRCULAR ECONOMY
3 _⁄√∕•	Good health and wellbeing	The circular economy avoids and removes hazardous substances from material use cycles, decontaminating the economy and allowing recycling without risks to nature and human health (e.g. phasing out asbestos or the use of leaded paints).
6 V	Clean water and sanitation	Most of the circular economy principles on material reuse, recycling, resource efficiency and industrial symbiosis are equally applicable to water, thus increasing the quality and the accessibility of clean water.
7 Ø	Affordable and clean energy	Circular energy solutions result in reduced energy demand. Sharing energy, using geothermal energy, increasing buffering capacity and using of renewable energy sources changes the way we use energy and stimulates clean energy use.
8	Decent work and economic growth	New business models lead to new companies and job opportunities. Circular economy business models in which companies collect or take back, repair and refurbish products are usually more labor intensive than linear models and create additional job opportunities. ^{24,25} In many emerging economies, waste collection supports a vast informal economy that lacks safe working conditions and fair remuneration.
9	Industry, Innovation and Infrastructure	The circular economy requires large-scale innovations in the built environment. This relates not only to industrial or technical innovations but also to infrastructural innovations. Circular economy is part of the solution, making industry more sustainable and resilient at local and global scale.
	Sustainable cities and communities	The circular economy, with its focus on local production, repair and leasing and extended producer-consumer contacts, requests a new, small-scale, spatial design to which cities should be adapted.
12 CO	Responsible consumption and production	The circular economy requires that: responsible care for products extends to the use and the post-use phase; that use of virgin materials are minimized; and that programmed obsolescence is phased out.
13	Climate Action	The circular economy can contribute to GHG reduction in many ways. Using secondary materials instead of virgin materials often requires less energy when considering energy associated with extraction. For example, reusing steel instead of having to mine ore and process it into steel can dramatically reduce GHG emissions. In addition to this, circular energy production and water management are key in climate change mitigation and adaptation.

Figure 1.6 Link SDG's to Circular Economy (Thelen et al., 2018)

1.2 PROBLEM STATEMENT

In the current starting construction projects, most of the materials used for the dwellings are gathered from processing factories. The use of virgin materials is common among regular construction projects and has been used in most of the projects in the past. Considering the downgrade of the availability of some of the materials in the world other options should be researched (Kleinjan, 2017). Fossil materials are getting extinct, especially structural building elements (Rijksinstituut voor Volksgezondheid en Milieu, 2016). A new system of reusing different materials have to be considered to pursue environmental change. According to the possibilities of a CE, it should not be standard anymore to use virgin materials within such a big industry that produces one of the largest

downgrades of valuable products and waste flows of leftover materials. The amount of waste in the construction sector will keep on increasing towards 2025 double to 2.2 billion tons (Slowey, 2018). Still the building sector is one of the bigger sectors that is lacking to innovate, compared to the other sectors (Agarwal, Chandrasekaran, & Sridhar, 2016). Still, multiple projects are over budget and over time, which shows the construction industry is ready for innovation in any way. Developing a system that connects information on the availability of building elements to the contractors would be an innovation to help improve the building sector. If there would be a so-called middle man who researches the number of materials available, it creates an interesting business case and could provide the connection which is missing, shown in Figure 1.7. In the current situation, multiple starting platforms are creating the role of the middle man adding value to the construction sector by connecting supply and demand. The platforms are relatively small market parties, testing their systems with a limited amount of stakeholders involved. There is much innovation possible in data management systems, but also a collaboration between stakeholders. Contractors and designers could pursue an environmental change in their company setup. Creating an overview and insight by investigating the feasibility would reduce the use of virgin materials, the amount of building waste and provides building projects from a circular point of view. Engineering and architectural companies would focus during their designs on the availability of existing materials and implement the technique into modern-day projects. This direct adaption of UM into the design process and the use of circular materials and processes could be a solution for future construction and design processes, considering the fact virgin building materials are less and less available in the future in the Netherlands (Kleinjan, 2017).



Figure 1.7 Problem Statement (own ill.)

This chapter focusses on the setup of the research question, based upon the problem statement that has been mentioned. The societal relevance and the scientific relevance of the research are elaborated in the first and the second section. The research question, together with the sub-questions are explained in the third section. Finally, the research goal, research deliverables and the setup of the research organisation are described.

1.2.1 SOCIETAL RELEVANCE

Multiple existing artefacts consist of usable building materials. However, not all of these artefacts are liveable, due to for example earthquake damage, which is the case in the Groningen example. The materials of these dwellings are of value for new building projects. Instead of demolishing the

buildings, gathering the materials by using urban mining and using them for new projects could be a way to extend the lifespan of building elements. Reusing materials of buildings could have a positive effect on the emotional value for residents. The people of in this case Groningen could intend to move because of how they handle the experience of being struck by earthquakes (Jansen, Hoekstra, & Boumeester, 2017), but while reusing some of the materials of the old building for the new construction, certain memories and sensitive parts could be saved. It makes the process of the construction much more circular, by reusing the materials it generates a circular and sustainable design. The homeowners of Groningen experience a lot of burden because of the situation around the gas extraction and the multiple earthquakes (Wijnbergen, 2018). Creating a method for reusing the materials of dwellings in Groningen for new housing projects as circular and efficient as possible will probably gain support by the owners of the endangered buildings if the materials are being used in the right matter. Other (earthquake endangered) areas around the world with a lot of unused building materials could benefit from this methodology, keeping in mind that the context of every area is different. It depends per case what level of urban mining can be implemented and which stakeholders would take part in the organisational process. For the citizens of endangered areas, the societal relevance of these reconstruction projects is of a high level. For example, architectural firm MVDRV started a project in a village called Overschild, where Winy Maas and associates in collaboration with the residents of the village, developed a plan to strengthen the village that is damaged by gas extraction. What makes this project so unique, is that the residents of the village whose dwellings are not suitable for reconstruction are directly involved with a design of their new house in the same village, using elements of their former house within the new design (Maas, 2018).

The case of Groningen is just an example of how this system of construction can be of use in a certain regional area, where there is a lot of sensitivity and emotional value connected to certain buildings. In this case, reusing materials could solve construction problems without losing the emotional value of dwellings, which could be an argument of importance to pursue reuse of building materials for designers and/or contractors of new dwelling projects in Groningen. However, the reuse of building materials in other projects is of relevance as well, due to the transition to a CE and the in Paris set 17 SDG's (UN, 2018). SDG goal #9 and #12, 'Industry, Innovation and Infrastructure' and 'Responsible Production and Consumption' shows the need for large innovation in the building sector and the need for reducing the use of virgin materials to a minimum. Companies could use the SDG's as well to be funded by the government for projects that involve reusing building materials, because of the motivation of a company to join the road to achieve the SDG's.

1.2.2 SCIENTIFIC RELEVANCE

The knowledge on implementing used materials of existing dwellings into new building projects is not that far developed yet. In current projects, this method has not been used that much, or there is not that much information about it yet. The gap lies in the fact that this system could be improved. Whether or not this model could be financially an improvement to contractors is relevant. The possibilities of implementing these materials into the process and researching whether or not this could be a valid business case from the perspective of the middle man are also interesting and create possibilities for starting companies. The scientific factor lays within the operationalization of the system into new projects, researching the feasibility.

Another important factor that could be improved is the matter of sharing information on building materials. New projects could be started and mostly designed out of reusable materials, but if there
is no information available or being shared by stakeholders that are in possession of this information, the stakeholders will not initiate using reusable materials as building materials and there will be no projects to start designing. Because of the confidentiality of the information, it is difficult to make the connection between the contractors and the designers. Multiple companies who are in the possession of information or datasets of different projects or material sets are in competition with similar companies. The demand for this material is getting bigger, which means it makes more sense for a company not to share the information about the available materials with other parties, without receiving compensation. However, the goal of a CE is to find a solution together to reduce waste and promote reusability of products. This is not possible if one of the most important stakeholders, meaning the companies that are in possession of the materials, keep the data to themselves, not participating in the circular mind. There should be searched some kind of solution, which makes government and private parties collaborate, without the fear of losing information and data systems to other clients or competition. This is where the gap lies within the structure of reusing materials on such a large scale and where further research could be conducted on how to solve this communicative problem.

1.2.3 RESEARCH QUESTION

As mentioned in the former chapters, there are a few challenges related to implementing the process of reusing building materials into traditional construction projects. This thesis researches the possibilities of increasing the circularity of these processes, to reduce the usage of virgin building materials in design and construction projects and help the construction industry with the transition towards a CE. The building materials that can be gathered by UM must be considered while initiating new projects and the overview of supply on existing building elements must be up to date. The feasibility of the process by adopting reusable materials into the design compared to virgin building elements is also of importance. The market has to be researched, whether or not there is a demand for this type of building materials or this type of system for contractors. It should be researched whether or not it is feasibly possible to adapt reusing materials to make it more cost-efficient, or more interesting according to the goals of the United Nations to receive funding. This is just a small list of the topics within this research, which in overall tries to clarify the challenges and research how the transition towards a CE could be stimulated in the building sector.

The problem statement combined with the societal and scientific relevance provides the following main research question:

How can an operational model link the supply of existing building materials with the demand for new construction projects in order to reduce the use of virgin materials and thereby improve circularity in the construction industry?

Figure 1.8 shows a conceptual model of the research at hand. The term UM is supported by the collection of the data. The data arises from the supply that consists of case studies coming from a project-, enterprise- or region-based level context. This supply provides neglected buildings that consists of building elements. These buildings elements are the reusable materials that trough UM will be used for new projects. A comparison of feasibility shows if the circularity of the building sector could be improved (Akadiri, Chinyio, & Olomolaiye, 2012). An extended explanation of the research

model and how the research is interpreted in design and planning over time is elaborated on in section 3.1.



Figure 1.8 Conceptual model research question (own ill.)

In the following sections, the sub-questions based on the conceptual model are presented and briefly discussed.

1.2.3.1 SUB-QUESTION 1

What specifications are necessary for a case study to be implemented as input for the operational model?

The first sub-question focusses on the input for the model. The cases selected are divided into three levels of scale, which is analysed in section 2.3, before elaborating on the cases. The goal of the levels of scale is to provide results from diverse models at a various size and the possibility to analyse the models from a different perspective. The following deliverables of this sub-question are:

- An overview and analysis of all the available case studies.
- A clear demand on how the data required as input for the model should be organized.
- A substantiated choice on which case studies are and aren't used as input for the model.

1.2.3.2 SUB-QUESTION 2

What is the current demand in the construction industry for new housing projects?

The second sub-question focusses on the demand for new dwellings, based on the wishes from the housing market, but also on the possible innovations within the construction industry for new housing projects. This is elaborated in section 3.4. The following deliverables for this sub-question are:

- A literature review of the current status of the housing market.
- A clear overview of different prototype dwellings, with each prototype thoroughly substantiated.

- An overview of the materials of which the standardized dwellings consists of, showing the current potential and specifications of these materials.
- An overview of wishes from stakeholders for new dwelling projects.

1.2.3.3 SUB-QUESTION 3

How can a comparison between the overview of the cost of virgin and existing building elements for new dwellings provide a feasible business opportunity?

The third sub-question focusses on the development of the operational model. Especially the feasibility of the different assemblies of dwellings are compared. The comparison results in advice for a business case. This sub-question provides the following deliverables:

- A clear overview of construction costs for virgin building elements, based on sources.
- A clear overview of construction costs for results building elements, based on expert discussions and sources coming from practice.
- A comparison between the costs of reusable and virgin building elements.
- Substantiated overview of the determination of the costs.
- Clear explanation and guidelines on how to operate the tool and how to implement variables and constraints for the calculations.
- Setup of the interface for the tool for practice.

1.2.4 RESEARCH GOAL

The main research question of this graduation thesis is focussed on how the supply of existing building elements and demand for new construction projects can be linked in order to reduce the use of virgin materials and thereby improve circularity in the construction industry. Because of the mentioning of the word 'how' in the research question, it aims the research towards a solution space (Barendse et al., 2012). In this space, a solution for answering the research question is generated, based on the available factors that emerge during the graduation project. How the circularity exactly could be improved depends on the output of the different parts of the research design. The literature overview and the research on the demand side from stakeholders and consumers could provide an answer on a social level, while the operational model could provide a specific tool for improvement, which can be used in further projects such as Groningen and other area's within the Netherlands or the world, where housing associations or municipalities have the ambition to create housing projects related to the transition towards a CE. The objectives of this research consist of creating a realistic advisory tool, investigate enough collaborating stakeholders and discuss the possibilities of UM within dwelling projects. The goal of this research is to minimize the usage of virgin building materials in starting construction projects. How this goal can be achieved is elaborated and substantiated by literature research and experts discussion. The supply and demand of the research questions is connected to each other. Based on the input from real-life cases and suppliers, the model is substantiated. This creates a set of bottlenecks, such as gathering the correct datasets and figuring out the right numbers for feasibility, which have to be overcome. When the correct numbers can be found and a feasibility study is executed, determining whether or not reusable materials or virgin materials are less expensive, the research goal of reducing the use of virgin building materials could be achieved.

1.2.5 RESEARCH DELIVERABLES

The research goal, as mentioned in the former chapter, presents how this research is going to be of value within the transition towards a circular construction industry. It aims towards a reduction of the usage of virgin building materials. As mentioned in the problem statement, there is a connection between supply and demand missing in the construction industry. Because of the absence of this connection, stakeholders cannot engage in circular projects. This link, or at least a fraction of this link, should be fixed, which will lead to more collaboration between different entities, increasing the circularity of building materials, eventually leading to a CE.

The main deliverable of this research is to create a hypothetical advisory tool that gives an overview to the different companies on the connection between the supply of reusable building materials and demand of circularity in construction projects. This tool is developed with the help of operational research. The tool is tested on multiple scales, based on the level of scale, explained in section 2.3. The tool can be evaluated and tested, according to the following subjects:

- The tool is able to perform correct calculations and provide a realistic output.
- The tool is based on literature research and fills a missing gap in research.
- The input of the tool is determined by experts within the field of construction and building sciences.
- The tool can be tested by running calculations with hypothetical and realistic data.
- The tool is adaptable on case studies of different scale, size and context.
- The tool consists of multiple modules, focused on transportation, construction and material validation.
- The tool is able to be implemented into practice and used by companies as test cases.
- The tool has the potential to be further developed with multiple modules and real-life cases from practice.

1.2.6 RESEARCH ORGANISATION

This MSc thesis is related to the master track Management in the Built Environment, which is part of the Architecture, Urbanism and Building Sciences master of the Faculty of Architecture and the Built Environment of the Technical University of Delft. The research is carried out in one of the four domains of the master: Design & Construction Management. This is shown by the connection within the management of building sectors, designing a more circular process and improving the structure of current and traditional construction projects. The development of the decision-making model is related to the operations research methods course, which is also part of the master Management in the Built Environment.

The first mentor is Dr. Ir. L.H.M.J. (Louis) Lousberg from the Design & Construction Management department of the master Management in the Built Environment. His expertise is related to feasibility, planning and managing the design phase of construction projects. His goal is to integrate practice with education and research. Next to educating in project management and academic skills, he is also researching the design of project management.

The second mentor is Dr. Ir. A. (Alexander) Koutamanis from the Design & Construction Management department of the master Management in the Built Environment. His focus is on information management and computational design. Also, he is an expert on the topic of urban mining, publishing the paper 'Urban mining and buildings: A review of possibilities and limitations' in 2018.

The third mentor is Dr. Ir. R. (Ruud) Binnekamp from the Real Estate Management department of master Management in the Built Environment. He is an expert on designing operational models and helps to develop the decision-making model that is related to the operations research course that is one of the electives of the master Management in the Built Environment.

The research commenced at the company Royal HaskoningDHV, with the Joint Interdisciplinary Project (JIP). The mentor during this project was structural Engineer Koen van Viegen, which provided guidance during the setup of the research and was the mentor of the project team during the JIP.

The graduation company for the second part of the MSc thesis is Alba Concepts, a circular organisation focused on consultancy, management and project development, located at 's-Hertogenbosch. Alba Concepts provided information from cases that are used in the research and guidance throughout the graduation process. The main supervisor during the graduation internship was Jim Teunizen, assisted by Jip van Grinsven and several other colleagues from Alba Concepts.

1.3 CONCLUSION

The circular economy has the potential to become the standard in multiple sectors, but the construction industry still has a lot of improvements to make before it has the certificate of a circular industry. The problem statement and different research questions mentioned in the former chapter shows a framework within the construction industry which has the potential to become circular. The different topics mentioned which are of relevance for the problem statement are being researched in literature studies and practice, elaborated in chapter 2. Literature research is necessary to determine the context for the operational research, the setup of the input and the output for the model and the eventual framework in which the model will operate to deliver the overview of the possibilities among reusable building materials, in order to answer the research questions.





PART 2 LITERATURE STUDY

The following chapter introduces a literature study about the different topics which are relevant for the research. The different sections provide a combined view on the current situation of the construction sector and shows where there are still gaps within the literature, which are the basis of this research and why this research is scientific relevant. The goal of the research is to improve the construction sector and its transition towards a CE. The literature study explores the current status and the possible improvements to be developed. This overview and conclusion is used as a basis for the input of the decision-making model, which is elaborated in chapter 3. Each topic is analysed and visualized in the conclusion of every section.

2.1 URBAN MINING

As mentioned in the introduction of the research, construction projects are still very traditional. The industry has been criticized for its traditional approach, particularly the one-off approach to projects and the fragmented structure not being efficient (Vrijhoef & Koskela, 2005). There are not that many examples yet of projects that are completely designed with reusable materials, almost every project which is designed is being built with mostly virgin building materials. This is because the traditional building sector focusses the expenses of new projects on foundation costs, instead of the life cycle of the materials. They are not willing to invest in circular initiatives, because of other priorities or because it is often simply too expensive (Ten dam, 2018). This shows that the construction sector is set in his ways, very traditional and not yet adapting to the transition towards a CE that is about to come and happening in different sectors already. But if new projects should be integrated with materials that are not virgin, the materials should be gathered, coming from a specific place. They should be mined from somewhere.

2.1.1 DEFINITION

Gathering materials from existing buildings that could be reused for new housing projects is called urban mining (UM). This term is a relatively new concept in the construction industry. It has been mentioned as a fancier term to describe different types of material recycling from annually generated waste flows (Cossu, Salieri, & Bisinella, 2012). Products taken from buildings could contain different materials, such as steel, wood, bricks (Krook & Baas, 2013), but also, in the case of housing, complete parts that could directly be used instead of putting the materials to urban waste, which the construction industry and relating projects are largely responsible for in many parts of the society (Agamuthu, 2008; Li, 2015). These materials hidden in buildings are interesting alternatives for the raw materials that in general are used for construction projects (Koutamanis, van Reijn, & van Bueren, 2018). Any anthropogenic stock could be used as a source for UM, such as infrastructure, industries and out of use products and is an extension on landfill mining, which only focusses on the extracting and processing waste stored in deposits, such as municipal landfills (Cossu & Williams, 2015). So, implanting this term into constructing processes and reusing materials seems to be the next step, using it as a strategic component of sustainability and circularity to improving the processes (Arora, Paterok, Banerjee, & Saluja, 2017; Cheng, Hsu, Li, & Ma, 2018).

In the current construction projects, UM is already adapted, but in a basic manner. The gathering of old materials and using those materials in a new project is simply what we call recycling. As mentioned by Koutamanis et al. (2018), it is stated that the performance of UM in resource recovery is already as high as it can get. On this level, it is not possible to improve the impact of UM, but looking at the situation from a different perspective could help. Organizing UM in the background process of AECO projects (Architecture, engineering, construction and operation of buildings) (Fox, Leicht, & Messner, 2010) still has possibilities (Koutamanis et al., 2018). The connection between the different phases and how UM in one phase is interpreted in the other phase shows the lack of overlap between the different project phases. If UM is being implemented by actors in different project phases, but collaborate in the project, it could improve the integration

of the process. This depends on the scale of operation, project-based, enterprise-based or regionbased, the amount of stakeholders and how much material is available in a project (Koutamanis, 2018). In the Groningen case, where materials could be gathered all over the province and the municipality is the main client and stakeholder in this case, the UM would take place on a regionbased level. In this case, the houses are vacant or are becoming vacant, which is a good example of refined natural recourses in the form of building components and materials. These empty buildings can be used as reserves for housing and repositories for UM or material extraction (Huuhka, 2016).

2.1.2 CHALLENGES

It is a challenge to determine whether or not UM is possible to implement in other levels of scale. The demand for more circular projects is of importance if on a smaller scale and direct adaption is asked for. The UM of materials could directly be used within new projects on the same location. The transportation factor is low because materials could be directly reused on the exact same location. If UM would take place at a region or enterprise-based level, the transportation of the materials is of larger influence, cost-wise and logistic wise. Even with these factors included, the comparison with virgin building materials should be made to have insight in whether or not it is a feasible business case to use existing building elements. There has to be thoroughly researched and documented what kind of materials are located at certain building sites and if they could be reused in new projects, making sure there are no mistakes made by suppliers and contractors about the information of certain building elements. This information should be up to date and shared among all stakeholders, to make UM a serious option within new construction projects. If the materials or the waste of former projects are not of the needed quality, there has to be taken care of on-site processing, which is generally difficult and expensive (Koutamanis et al., 2018; Ulubeyli, Kazaz, & Arslan, 2017).

2.1.3 CONCLUSION

The different literature on UM shows that the term is still in development and could be of use within the transition towards a CE, but there is still a lot of information and knowledge to be gathered to be sure of the added value of UM in the construction sector. Every case is of a different context, which makes it difficult to develop a certain tool on the concept of UM that could be used in every project. Also, because the term is interpreted differently by stakeholders, it is unclear whether or not the term is the right way to go when talking about reusing building materials. A clear definition of the concept is necessary to have it being used on a large scale in projects. Also, the challenges mentioned in the former section concerning information about building materials and especially the transparency on data is a problem that is yet to be solved within the near future, to make UM a realistic way of future construction projects.

2.2 MATERIAL FLOWS IN THE CE

When the construction industry mentions the transition towards a CE, the suggestion is made that most of the materials within the building sector have a linear lifespan and are not being used circular yet. The material flows mostly are being used for a single purpose, before they are being demolished or disintegrated to landfill, turning into waste that cannot be reused anymore and eventually being transformed to energy or even being burned or deposited, which are the lowest classification on the Ladder van Lansink (Lansink, 1979), which is used in the European Waste Framework Directive since 2008 as general guideline for the classification of waste flows. The European Union has mentioned making it its mission to face the challenges relating resource scarcity and shift from a linear to circular systems, whilst regarding waste as a recourse (R. J. Geldermans, 2016). So the waste of materials flows and flows itself should be reinvestigated and researched, to make sure only the highest classifications of the Ladder of Lansink are at stake for these materials (Figure 2.I).



Figure 2.1 Ladder van Lansink (Lansink, 1979)

So what schematic flows on the transition to the circular use of materials are already existing? An example is powered by Cradle to Cradle[®]. Figure 2.2 shows how the material flow of techno sphere and biosphere materials would work in a CE. In this system, no waste is being produced and the materials are either being used as renewable energy or as know-how, the feedback loop to redesign & prototype the product, after collecting it when a certain material fulfilled its lifespan (Brey &



Figure 2.2 Material flows in a circular economy [Source: EPEA & Returnity Partners] (R. J. Geldermans, 2016)

Hansen, n.d.). It shows an excellent example of the different paths materials take to recreate value within their lifespan and that material, from a biological or a technical nutrient, can still be of value even if it the material is not being used in its primary function.

Figure 2.3 shows a similar scheme, where especially the minimization of the landfill is being highlighted (EMF, 2013). The arrows on the right illustrate the circulation of materials in technical cycles. These range from maintenance of existing products, to refurbishment and lastly recycling. The longer an arrow the less sustainable the option presented (Isaac, 2018).



Figure 2.3 The circular economy—an industrial system that is restorative by design (EMF, 2013)

Cheshire (2016) proposes a gradation of the steps of what is possible with materials when their life cycle is completed. The gradation consists of six levels of material utilization, visible in Figure 2.3, which are CE principles applied to the built environment. This concept of gradation of the action to be taken for materials could be implemented into the construction industry, as Cheshire (2016) proposes. The concentric circles represent the technical and biological loop (van Vliet, 2018). The 6 steps of material utilization are:

- Retain
- Refit
- Refurbish
- Reclaim/reuse
- Remanufacture
- Recycle/compost



Figure 2.4 Applying Circular Economy principles to the Built Environment (Cheshire, 2016)

2.3 LEVEL OF SCALE

Reorganizing the construction industry is considered to be a long-term objective for a country, regarding the transition to a CE and the many different processes that are integrated within this sector. To create a system to support the entire building sector, it is important to realize not all the different aspects can be handled at once. It takes time and multiple steps to transform a traditional, linear industry into a circular industry. This is also relevant for the scale of the project, on which level of scale a certain revision of a system would be implemented. The level of scale depends on what type of case studies could be used for the testing and development of new construction industry techniques, giving insight how much of the processes could be improved, in line with the transition to a CE. This could, in small steps, trigger the construction industry to innovate and follow another sector such as finance, transportation or healthcare, that are actually seeing the potential in technology and grasp opportunities to boost productivity and commercial gain (Robinson, 2018). In the next sections, three levels of scale are elaborated: The project-based level, region-based level and the enterprise-based level (Koutamanis, 2018), illustrated in Figure 2.5. For every level of scale, a short elaboration and an example project are illustrated.



2.3.1 PROJECT-BASED LEVEL



The first level of scale is the project-based level. In this level, the context is based on one or just a few projects, to deliver input of data for a particular model. This model could deliver a solution for a small setup of a project, which could find the solution within a short time span. The area that would be focused on in project-based levels is relatively small, so subjects as transportation of materials are of a smaller factor. The materials from certain cases could directly be used for a new project, which could be at the same or a close location and it of the same size or scale, compared to the existing

projects, which is being used as a case study. In this case, not that many stakeholders have to be involved within the collaboration, making the transition of the materials efficient, reducing the number of intermediary's. The number of different materials and building elements can be of a wide range, aiming to reuse as much as possible within the new projects. An important innovative design decision that should be made on this scale of projects is to implement the possibilities of materials available to reuse within the design phase of the new projects. In this case, there would already be a new destination for most of the parts of building elements, which are still located within the current buildings.

An example of a project that is designed and build conforming the circularity standards of the current age is the Green House in the station area of Utrecht. A collaboration between Albron, Strukton and Ballast Nedam, following the design of architectural firm Cepezed, created the completely circular and extremely sustainable pavilion with hospitality and meeting functions (van der Voort, n.d.). The pavilion has a lifespan of 15 years, at the end of this time period the complete building should be exploited and because of its focus on circularity, the design of the Green House is completely detachable. This makes it possible to deconstruct the pavilion and rebuild it, in a different or the same shape on a different location. The materials that are used to construct the building are reused as much as possible. The rest of the elements are acquired by lease, still in possession of the supplier, creating a complete circular business model (van der Voort, n.d.). This project is an example of a project-based scope, collaborating with the designing firm about the qualities of the future building. The parties decided on a circular building, looking for materials that could be used and adapt the available materials into the design decision. For example, the curtain wall used for the Green House is retrieved from the former Knoopkazerne, which was just next door (Duurzaam Gebouwd, 2018). A combination of the reuse in projects that are close by is an effective solution for project-based level cases.



Figure 2.6 The Green House (Duurzaam Gebouwd, 2018)

2.3.2 REGION-BASED LEVEL



The second level of scale is focused on a region. A specific area within a province or a country is the basis for the data input and the data will be of a larger proportion compared to the project-based levels. Multiple stakeholders have to be involved, because of the larger size of the projects and the multiple information points that is part of the research. A municipality or larger housing corporation should be involved to carry out a project on a regional level. The collaboration between the multiple stakeholders is of great importance. The datasets should be shared and the demands for new projects should be aligned. If so, the information coming from different existing projects could be assembled and exploited in the best division of projects.

The case of Groningen explained in chapter I.I.3 is an example of a region-based project. In this case, the information on the materials has been gathered by a certain stakeholder and the demand for new buildings is coming from the municipality. A sufficient amount of buildings was (partly) demolished by the earthquakes. These shocks by gas extraction are still going on, disrupting houses and creating an insecure living situation for the residents of Groningen (Nu.nl, 2019). Even though the buildings are damaged, most of the building materials are still intact and ready to be reused. A collaboration between multiple stakeholders could be the solution to the problems of such an area. You would need a stakeholder that could analyse the buildings which are (partly) destroyed and that could make an inventory of the usable building elements. The municipality should be responsible for the demand and financial structure for new dwellings for the residents within the area. A structural engineering firm or an architectural firm should be in charge of designing earthquake resisting dwellings, which could be circular designed by using the inventory of the existing building elements. This would create a circular business model, benefitting on fewer transportation costs, less processing costs and less use of virgin building materials, instead of reinforcing dwellings with wooden beams, which is not a circular and effective solution (Figure 2.7).



Figure 2.7 Earthquake damaged dwelling at Groningen (KNAG, 2016)

2.3.3 ENTERPRISE-BASED LEVEL



The third scope level is based on the concept of an enterprise. An 'enterprise' means a unit or business organisation or activity, especially a business organisation (Enterprise, 2003). In this case, the enterprise scope would focus on a system of a certain material or multiple materials, which could be used for different projects over a considerably large area. The business case is the implementation of the waste of a specific product, which could be of value for other sectors, the material being used in a different function. From the perspective of the company, materials could have value, while for different stakeholders the same materials belong to their local waste streams. This creates a more circular flow of materials. A similar concept is possible within other sectors as well, such as the construction sector. A company or enterprise could be leading on the availability of reusable elements from a certain material, being used for a specific part of new buildings.

An example of a company who was looking for a specific type of waste that could achieve much more than landfill is the company bio-bean. Instead of in the construction sector, bio-bean sees opportunities in the waste of coffee beans and turns this waste into value, advanced biofuels and biochemicals. The company, based within the United Kingdom, has found the link between different sectors and turns the landfill of the first sector into value for certain other sectors, by creating products from as much waste flows of coffee beans that could be achieved before putting it to landfill. Examples of products created out of coffee waste are the high-performance, sustainable heating briquettes Coffee Logstm (Figure 2.8), biomass pellets that are carbon neutral and used for industrial-scale heating, and biochemicals for a range of commercial purposes (Harrison et al., 2016). The entire country could perform as a case for the employees of bio-bean because coffee waste can be found everywhere and nobody is using it as a resource for products.



Figure 2.8 Coffee Logstm, product by bio-bean (Harrison et al., 2016)

2.4 MATERIAL MANAGEMENT SYSTEMS

The concept of having one department within a company responsible for the flow of materials, from supplier through production to consumer, is relatively new. Although many companies have adopted this type of organisation, there is still a number that has not (Arnold, Chapman, & Clive, 2008). This function within a company is called material management and provides possibilities for a company to improve the profit and the reusability of materials. In the transition to a circular construction industry, information about the building materials that are present and the management of the inand outgoing materials within a company's structure are of great importance to reach a fully circular system. These materials passports are important for clear communication and collaboration between for example contractors and design firms. The difficulty within this information is the competition that consists between the different parties. There are multiple systems currently being developed, creating the problem of a non-fully integrated market in material passports. The developers of the material management systems are seeking to invent the most optimal system or platform, which could have an entire country or even bigger integrated. The consequence of this market is a decrease in interaction and collaboration between different sectors and stakeholders.

The call for a CE and innovation in the construction industry should encourage the competing stakeholders to collaborate and share the information, generating an integrated flow of material information. To classify information on building materials would become unnecessary and every interested stakeholder could browse between all the different available materials. Unfortunately, this would imply that an entire structure of the financial market and competition that drives these markets and innovation would be obliterated. The factors that stimulate private companies to develop systems and thrive for innovation in the building sector would not remain to exist.

2.4.1 MISSING OF THE 'MAN IN THE MIDDLE'

As mentioned within the problem statement of this research, there is a missing connection between the demand for new projects and the information on materials that are available to be reused. In the current economy, the platforms that are being developed focus only on the companies which are eager to join the organisation, while the construction companies, such as contractors or developers are only interested in discovering the most profitable price for materials. The responsibility of creating the link between both sides is not assigned to anyone. This missing of the 'man in the middle' concept creates an interesting business case. A third party could provide a link between the different parties. This stakeholder could be commissioned by the collaboration to research the link between supply and demand on building materials. First, there should be transparent information coming from the owners of materials available for reuse and on the other hand, there should be a clear demand for building materials for new projects. The business case focusses on the efficiency of the process, which could be improved, saving time and costs within production and processing, even transportation. The middle man could design an advisory tool that could modulate the reuse of the materials from a specific stock promoting the efficiency and advantages of reusing materials. The strength of this concept lies within the fact that it creates a service that is not yet existing. It combines the demand within different aspects of the construction industry, connecting parties that normally do not collaborate. Multiple existing companies have experienced successes and development from an idea to create a service for a not yet existing solution to the demand. For example Airbnb, an online marketplace that creates a connection between the demand for holiday apartments and the supply on private dwellings, offered by the owners. Also Uber, a company which offers services such as peerto-peer ridesharing, but also ride service hailing, food delivery and a bicycle-sharing system (Uber, n.d.), reacted on the demand of visitors in cities and created the connection with providers. A Dutch example is 'Marktplaats', which gives people the possibility to buy and sell second-hand products. Every transaction is between customers, so the only responsibility of Marktplaats is providing a safe and transparent negotiation environment, creating a connection between supply and demand.



Figure 2.9 Airbnb / Uber / Marktplaats

So these are examples on different sectors, but there are also already many different systems and platforms available on building materials, which are partly up and running, but also some of the

systems are still yet to be developed. The following sections provide an overview of the different systems and platforms, related to the construction industry.

2.4.2 MADASTER

Waste is material without identity', quoted by Thomas M. Rau, one of the authors of the book 'Material matters; Het alternatief voor onze roofbouwmaatschappij' (Rau & Overhuber, 2016). The book describes the transition to a new economic system where consumers are no longer owner, but a temporary user of products and materials. Madaster was based on the concepts and insights described in the book (Madaster, 2019).



Figure 2.10 Madaster logo (Madaster, 2019)

Madaster is the cadaster for materials, making sure raw materials are being documented so that they can remain unlimited. A material passport for all the materials within certain buildings is created so that every material gets an identity and could never disappear into waste. The platform Madaster acts as a library where all the passports are stored and generated. The platform is independent and being developed under the supervision of the Madaster Foundation (Madaster, 2019).

To make use of the Madaster platform as a company, a subscription gives access the construction data online. A certain price related to the number of users and the total gross square meter of a specific part of real estate are generated and the minimal subscription of the platform has a minimum duration of a year. The platform is also available for private individuals, which enables you to make a dossier for your own home (Madaster, 2019).

The Madaster project is a Dutch non-profit foundation with an aim to eliminate waste. The project has received funding from the European Union's Horizon 2020 research and innovation program (Madaster, 2019).

2.4.3 CIRDAX – RE-USE MATERIALS

The second tool is designed by the company Re Use Materials, located in Heerlen in the most southern province of the Nederlands. The 'Cirdax' tool is a material management software, aspiring to be the first national system that makes it possible to manage materials during every step of the way in a real estate process. The focus within the tool is efficiency and efficient use of raw materials within demolition, construction and maintenance processes. The software of the tool includes multiple tools and methods to create insight within the materials (Re Use Materials, 2019). A difference between Cirdax and Madaster is the fact that Cirdax also provides assistance with clients in the follow up of the reuse of the building materials, by offering a (BIM) design tool and a social return tool. Madaster works as a market place for materials, but Cirdax gives insight in design possibilities, but also measurement on circularity within the realization of the new buildings.

The tools that Cirdax uses are described and visualized in Figure 2.11. The first three tools are up and running, the other tools are still in development by Re Use Materials, but still give an idea of what skills the tool will possess in the near future.

- Assessment app: Assessment of the materials, which are being added into the system. The app (iOS) makes it possible to identify the materials at hand.
- Material passport: Gives identity to the materials. This identity is determined by specifications such as quality, quantity, dimensions, color recyclability and detachability of the material. All of these passports are stored in Cirdax, which grants accessibility at any time.
- **Performance dashboard:** Visual representation of the performance of the tool, based on circularity, finance, carbon dioxide emissions and social return.



Figure 2.11 Overview Cirdax tools (Re Use Materials, 2019)

The tool is yet to be developed even further and is only accessible for clients who are in collaboration with Re Use Materials. It is not possible to log in or review the system of Cirdax when you are not involved directly with the company, so, unfortunately, it is not possible to review the actually developed software of the tool, because of its restrictions. It creates a perfect example of the current status of the CE in the construction industry. The lack of accessibility and transparency about building information or material management tools is still a difficult border to pass.

2.4.4 EXCESS MATERIAL EXCHANGE (EME)

The third example is the Excess Materials Exchange (EME) platform, which is a digital facilitated marketplace where companies should be able to exchange excess materials and products. EME tries to solve the problems mentioned of lack of transparency, lack of reliability and a lack of connection by developing the following products: a material passport for all the recourses, tracking and tracing of materials, a valuation module and al assisted matchmaking. By giving materials identity and providing a location where data of a product can be collected through every step of its lifecycle, there is actual information about the possibilities of each material and its location throughout value chains. The blockchain technology, which can be illustrated as a distributed database of records or shared public/private ledgers of all digital events that have been executed and shared among participating stakeholders (Crosby, Nachiappan, Pattanayak, Verma, & Kalyanaraman, 2016; Saberi, Kouhizadeh, Sarkis, & Shen, 2019), stores and secures the data, while the al assisted matchmaking will try to connect within different sectors. Waste in one sector could be of high value for other sectors. The matchmaking process looks for the most optimal reuse opportunities for any type of material (Damen

& van Maaren, 2018). The EME focusses on the area of the Amsterdam Metropolitan Area (AMA). Within this region, six priority types of materials are taken into the transition to a CE, which are: Textile, biomass, construction/demolition, E-Waste, plastic and diapers. These six flows of materials are being observed in the following six sectors: Services, tourism, industry, consumer goods, building and biomass (RoyalHaskoningDHV, 2018). But next to the waste flows, EME focusses on any possible reuse of material types and already pursued a lot of pilots to review the possibilities. For example, neglected carpet pieces or peels of oranges. Multiple companies have been involved within the testing and developing of the platform, such as Heembouw, Schiphol, ProRail, DSM and Rijkswaterstaat (van Doorn, 2018). Such as Cirdax, EME is still developing and hasn't optimized all the factors of the platform yet.



Figure 2.12 Excess Materials Exchange logo and modules (Damen & van Maaren, 2018)

The EME-platform provides four modules, visible in Figure 2.12, which are crucial for lifting the barriers between companies and providing them a chance to join the transition towards the CE. The first module is the material passport, which identifies materials and records the relevant specifications of the materials into a database. Next, the tracking and tracing system makes sure the information on when and where materials are being released is up to date, providing the most optimal transportation routes and up to date knowledge for possible distributors for the materials. The valuation module gives an insight into the financial, social and ecological value of the materials. Based on this information, information coming from the residual flows could be adapted into the design process, which is called reverse engineering. This will directly increase the value of the residual flows. The fourth model provides the matchmaking between supply and demand, creating a marketplace for secondary materials and providing information on whether or not the residual flow of materials could be reused. Stakeholders could place an ad requesting materials, but also offer their materials to other interested stakeholders, which are connected to the platform.

2.4.5 CONCLUSION

The former mentioned platforms or modules are just a small grasp out of the many different systems that are being developed throughout the world. It shows the demand for a CE and the motivation for multiple start-ups and companies to be the leading expert in reusing material systems. This provides an interesting market with competing companies, trying to win over clients to join their own developed system. However, as mentioned within the introduction of this chapter, competition should be exchanged for transparency and collaboration, to achieve a complete circular industry on building materials. The problem is the commercial interests for the typical marketplaces. This provides a business model that stimulates the companies to keep on developing these platforms.

When all the residual waste flows and information sets on available building materials are being made transparent, there should be a different business model for the companies to keep the innovation flow going. This is a challenge that is still yet to be solved.

2.5 DEFINITION OF BUILDING ELEMENTS

The internship company Alba Concepts provided insight in the 'Building Circularity Indicator' (BCI). The BCI is an assessment model that aims to provide guidance during the decision-making process to concretize the circular ambition of different stakeholders (Verberne, 2016). The structure of the model is developed by adopting the basis of the material circularity indicator (MCI), by the Ellen Macarthur Foundation & Granta Design (2015). The model provides the next step towards measuring how well the principles of the CE are implemented in a building project (van Vliet, 2018).

For this research, the practical version of the academic model, developed by Verberne (2016) and van Vliet (2018), is currently being used by Alba Concepts in practice. This model is elaborated, analysed and connected to this research in the following sections. During the development of the assessment model, it was of importance to discuss in different level of details with stakeholders on the subject. This ensured a set of definitions used for the different parts of the building, which are defined based on models from literature. These definitions of the building elements are elaborated and implemented in further development of this research.

2.5.1 BUILDING LEVELS

There are multiple definitions of the structure of the building levels and mentioning of different components. This creates a lot of haziness between stakeholders, discussing data in the construction industry. The research of van Vliet (2018) aims to use classification for building materials to differentiate between detail levels in an objective matter. This classification consists of multiple different from literature adopted overviews, eventually being used in the BCI in practice. The



Figure 2.13 Theory of material levels (Durmisevic, 2006)



Image source: Alba Concepts (2018)

Site: This is the geographical setting, the urban location, and the legally defined lot, whose boundaries and context outlast generations of ephemeral buildings. "Site is eternal"

Structure: The foundation and load-bearing elements are perilous and expensive to change, so people do not. These are the building. Structural life ranges from 30 to 300 years (but few buildings make it past 60, for other reasons).

Skin: Exterior surface now change every 20 years or so, to keep up with fashion or technology, or for wholesale repair. Recent focus on energy costs has led to re-engineered skins that are air-tight and better-insulated.

Services: These are the working guts of a building: Communications wiring, electrical wiring, plumbing, sprinkler system, HVAC (heating, ventilating, and air conditioning, and moving parts like elevators and escalators. They wear out or obsolesce every 7 to 15 years. Many buildings are demolished early if their outdated systems are too deeply embedded to replace easily.

Space Plan: The interior lay-out – where walls, ceilings, floors and doors go. Turbulent commercial space can change every 3 years or so; exceptionally quiet homes might wait 30 years.

Figure 2.14 Shearing Layers (Brand, 1994)

level to building level, which is also complementary to the NEN 2660:1996 top-down and bottom-up overview (NEN, 1996).

The shearing layers overview by Brand (1994) is a well-known method for describing the different building layers from inside out, combined with the expected lifespan of each layer. A brief elaboration of each layer is described by Brand (1994) in Figure 2.14. The overview is very generic and it does not require specific definitions for everybody to classify a building material (van Vliet, 2018), which makes it an applicable overview for defining materials to layers. Based on the assumption that these layers have different life cycles, design decisions can be made regarding their end of life scenarios (van Vliet, 2018). The different lifetimes result in multiple moments during the lifespan of a building when different components have to be replaced, some components more frequently than others (Brand, 1994). This creates the opportunity to look at the building in a more circular matter and see what the possibilities are concerning the circular ambition of buildings.

Where (Brand, 1994) proposes an assembly or disassembly, which is necessary within the process of calculating the BCI, on functional decomposition, Durmisevic (2006) proposes a physical decomposition. Figure 2.15 shows an overview, where the levels of a building are ordered hierarchical. In these three pyramids, a visualization is presented of three different divisions of building the structure. The first shows that the structure of materials is fixed, the second partially fixed and in the third pyramid, the structure of materials is completely open to disassemble individually (Durmisevic, 2006). These multiple building levels from literature, which are part of the development of the BCI is taken into account further within this research, in determining which materials to use.



Figure 2.15 Fixed, partially fixed and open structure on different building levels (Durmisevic, 2006)

2.5.2 BUILDING CIRCULARITY INDEX – PRACTICE MODEL

The BCI as used by Alba Concepts in practice to provide guidance during the decision-making process to concretize the circular ambition of different stakeholders is designed by Verberne (2016). This section shows the theoretical and conceptual model of the tool. The BCI as mentioned is a theoretical model developed to create a simple measure of achievement to enable the transition to a CE (Verberne, 2016). The BCI focusses on the technical cycle in the CE model and defines a sum of key performance indicators that are included in the calculation model, displayed in Figure 2.16. This model presents the theoretical requirements of the BCI, essentially broken down into four steps (Verberne, 2016);



Figure 2.16 Conceptual model of the Building Circularity Indicator assessment model (Verberne, 2016)

- Calculate the Material Circularity Indicator (MCI) with the material input, output and lifecycles of products.
- Calculate the Product Circularity Indicator (PCI) by determining the disassembly possibilities of products and multiplying this with the MCI of the products.
- Calculate the System Circularity Indicator (SCI) by categorizing products according to shearing layers of Brand and normalizing with a factor like weight, volume, price, etc.
- Calculate the BCI by multiplying the SCI with the level of importance of the shearing layers of Brand.

Product Circularity Index (PCI)		Element Circularity Index (ECI)		
Materiaal Index (MI)	Losmaakbaarheid Index (LI)	Herbruikbaarheid Index (FI)	Losmaakbaarheid Index (LI)	
• Herkomst materialen • Afvalscenario • Technische levensduur • Volume	 Type verbinding Toegankelijkheid verbinding 	"Een element is een clustering van producten die onl baar met elkaar verbonden zijn. Pas op het moment o verbinding losmaakbaar is en de schade beperkt blijf de clustering en vormt het een element."		

Figure 2.17 Product Circularity Index & Element Circularity Index (Alba Concepts, 2018)

The theoretical model of the BCI assessment model is adapted and transformed by Alba Concepts (2018) into an understandable model for practice. This model assesses the circularity ambitions of clients in a couple of steps. First, a list of key performance indicators, in Figure 2.19 described as drivers and preconditions, are presented. To eventually calculate the BCI, the PCI and ECI have to be determined. The PCI (Product Circularity Index) is determined by the Material Index (MI) and the disassembly potential. The MI describes the origin of the material, whether the materials are virgin or non-virgin, it describes the future waste scenario (landfill/incineration/recycling/reuse), the

technical life span and the functional life span. The disassembly factor describes the type of connection between materials combined with the reachability of these connections within the product. Appendix II.8 shows an overview of each type of connection with a different factor to be calculated with, based on the disassembly list of connections (Durmisevic, 2006; van Vliet, 2018; Verberne, 2016). The ECI (Element Circularity Index) describes elements as clusters of products, which are inseparably connected to each other. Once the elements are able to disconnect from each other without damaging (to a certain level) the products, the cluster ends and creates a new element. The amount of reusability combined with the disassembly potential defines the ECI.



Circularity Index (BCI)

Figure 2.18 Building Circularity Index Calculation steps (Alba Concepts, 2018)

2.5.3 CONCLUSION

The different factors described within the theoretical and conceptual model of the BCI are based on literature and attempts to create a general overview of definitions for different components of the construction industry. This substantiated universal language of construction terms is something that has the potential to be infiltrated within multiple companies who tackle the same issues. For this research, the given description used for the BCI calculations could be of use in defining the input and output for the operational model.

2.6 HOUSING DEMAND IN THE NETHERLANDS

In the transition to a CE, starting new housing projects should be integrated within the goals to reach in the upcoming deadlines, such as the Paris Agreement in 2050. Materials and information about current stocks are present, but what is the future perspective of dwellings and the housing market? Multiple articles express the need for growth in housing in the future concerning the increase of population coming years, but also a change of demand in living environments compared to the dwellings build the last 80 years ago. Next, to the change in demand, the same articles express the concerns included within this ambition of the new dwelling projects. Van der Meulen (2018) expresses the problems within the growth of the dwellings capacity of the Netherlands. In 2017 70.000 dwellings are added, which is still not a significant growth to fulfill the needs of the growing population. At the moment there is a shortage of 246.000 dwellings in the Netherlands, which is 3.2% of the current housing stock. According to Van der Meulen (2018), if every year the same amount of dwellings is added to the housing stock, it will not conquer the growth of the demand for new dwellings because of the rising population. In the current speed dwellings are being constructed, multiple construction companies such as VolkerWessels, Heijmans and TBI predict not even half of the prospected dwelling will be built in 2030 (ANP, 2019). And is it even possible to reach this amount of growth of new dwellings? Because the government puts the responsibility of most of the large housing projects at most of the municipalities, which mostly lack of expertise, resulting in either too expensive projects without any idea of costs factors or quick and cheap production of standard dwellings, without any concern for municipal plans and integration with the surrounding area (Vermeeren, 2018).

2.6.1 GROWTH OF POPULATION IN THE NETHERLANDS

Prognosis shows that between 2017 and 2029 the population of the Netherlands will grow with approximately 697.000 households, which is 8,7% of the current households (Rijksoverheid, 2018). Table 2.1 represents the growth among the different types of households. The different prognoses still has a difference of approximately 300.000 households between the highest and lowest variant. The number of single households will increase the most, such as the people with the age of 65+ (Table 2.2).



Table 2.1 Prognosis of the household size in the Netherlands in the period 2017 – 2030 (Rijksoverheid, 2018)

The predicted growth of population and households does not necessarily mean the demand for dwellings is equally divided over the country because the prognosis differs for each province. Especially close to the big cities in the western part of the Netherlands, the population in 2030 will have increased more compared to the northern and southern provinces. Figure 2.20 visualizes the percental growth of each municipality among the entire country in 2030 compared to 2015 (Rijksoverheid, 2018). This shows that the demand for new dwellings is at a higher level in certain provinces compared to others, looking at the near future of 11 years. These areas will, because of its higher demand for dwellings due to faster-increasing population, create multiple starting dwelling projects. The demand for a CE will create ambition for the new dwellings, to connect to the SDG's will have to be taken into account when developing and designing the new dwellings.

Households Configuration	2017	2030	Increase 2017 - 2030	Procentual Increase 2017 - 2030	
Туре					
Single	2.961.000	3.459.000	497.000	16,8%	
Single parent family	562.000	589.000	27.000	4,8%	
Living together with childeren	2.024.000	2.006.000	-18.000	-0,9%	
Living together without childeren	2.206.000	2.372.000	166.000	7,5%	
Other households	41.000	48.000	6.000	14,7%	
Age					
< 30 years	950.000	962.000	13.000	1,3%	
30 - 44 years	1.801.000	1.957.000	155.000	8,6%	
45 - 64 years	2.926.000	2.669.000	-258.000	-8,8%	
> 65 years	2.117.000	2.886.000	769.000	36,3%	
Total households	7.794.000	8.473.000	679.000	8,6%	

Table 2.2 Development of households in size and age, in the period 2017 – 2030 (Rijksoverheid, 2018)



Figure 2.20 Bevolkingsgroei per gemeente, 2015-2030 (Centraal Bureau voor de Statistiek, 2016)

2.6.2 GROWTH OF DWELLINGS PROGNOSIS

The growth of the population will result in a growth in dwellings (Rijksoverheid, 2018). The overview by Rijksoverheid (2018) predicts, based on actual data about the current status of the dwelling stock and the demand for new dwellings because of the predicted growth of population, what the development of the dwellings stock will be towards 2030. Table 2.4 shows for every province of the Netherlands in how many dwellings the development possibly will result, while Table 2.3 shows the prognosis of the planned capacity when the demand and current stock are being put next to each other.

Provinces	Increase 2017 - 2030			Difference vs demand	
	Prognosis demand	Prognosis supply	Netto plancapacity	Prognosis demand	Netto plancapacity
Groningen	12.000	15.500	16.500	3.500	4.500
Friesland	11.000	12.500	15.000	1.500	4.000
Drenthe	10.000	11.000	14.500	1.000	4.500
Overijssel	25.000	27.000	23.000	2.000	-2.000
Flevoland	31.000	29.500	38.000	-1.500	7.000
Goldorland	65.000	69.500	77.500	4.500	12.500
Utrecht	77.500	75.000	81.500	-2.500	4.000
Noord-Holland	166.000	175.500	213.000	9.500	47.000
Zuid Holland	169.000	167.000	158.500	2.000	10.500
Zeeland	5.000	6.500	10.500	1.500	5.500
Noord-Brabant	92.000	104.500	119.500	112.500	27.500
Limburg	2.500	7.000	26.000	4.500	23.500
Total	666.500	701.000	793.500	134.500	127.000

 Table 2.3 Prognosis on increase in dwellings needs, dwelling stock and plan capacity per province in the period 2017 – 2030 (Rijksoverheid, 2018)

So we know that many buildings have to be developed in the near future to fulfil the housing shortage prognosis, but what kind of buildings are necessary for the future? Looking at the SDG's, most of the houses should be energy neutral, cut from natural gas and from a sustainable point of view developed. Multiple cities are already in the progress of redesigned current dwellings, which are poorly insulated and still running on gas. One of the cities is the capital Amsterdam, which aims to establish effective collaboration between the city and stakeholders which results in scalable, cost-effective building design that answers to the wishes of the residents of Amsterdam (Kanyemesha, 2017). Together with the residents, the questions are being asked what their potential living situation could look like and how to transform current dwellings and design new dwellings into neutral energy homes, within a feasible budget (Kanyemesha, 2017). Eventually, 100.000 new dwellings are the amount that is currently set as the goal for the project from Amsterdam Smart City, in combination

with Alliander and Arcadis (Kanyemesha, 2017). This project shows an interesting example of how to tackle the challenges for the need of new dwellings, but still the motivation to keep the focus on developing energy-neutral homes and also the feasibility in costs and duration of the project. Eventually, this will result in residents and municipality joining the transition towards a CE.

Provinces	Supply 2017	New constructions 2017 - 2030	Deconstructions 2017 - 2030	Supply 2030	Development supply 2017 - 2030
Groningen	276.500	24.500	9.000	292.500	15.500
Friesland	296.500	19.500	7.000	309.000	12.500
Drenthe	219.000	15.000	4.500	230.000	11.000
Overijssel	497.500	40.500	13.000	524.500	27.000
Flevoland	164.000	30.500	1.500	193.500	29.500
Gelderland	888.000	85.000	15.500	957.500	69.500
Utrecht	554.500	89.500	14.500	629.500	75.000
Noord-Holland	1.307.500	208.000	32.500	1.483.000	175.500
Zuid-Holland	1.666.500	219.500	53.000	1.833.500	167.000
Zeeland	184.500	11.000	4.000	191.000	6.500
Noord-Brabant	1.104.500	122.500	18.000	1.208.500	104.500
Limburg	527.000	17.500	10.000	534.500	7.000
Total Netherlands	7.686.000	883.000	182.500	8.387.000	701.000

Table 2.4 Prognosis on development of the housing stock, in the period 2017 – 2030 (Rijksoverheid, 2018)

2.7 CONCLUSION

Looking at the current state of the construction industry, multiple factors are ready to innovate. Especially the rapid transition towards a CE, in which the construction industry should participate. There are already multiple platforms and construction companies that are making an attempt, but a fully integrated system within the construction industry is far from existing yet. The literature study shows that it is recommended for researchers of the construction industry to tackle the innovation step by step and research the factors that could be improved, concerning feasibility and efficiency, instead of trying to tackle the entire structure of the construction industry. A focus to improve a particular module of the entire system will result in follow up innovations, as a reaction to the development of smaller structures. The creation of the middle man would be a starting example, which could be used as a pilot within to construction industry and promote the possibilities of the rapid development. Systems already developed could connect within the transition and future starting design and construction projects could be realized, keeping the SDG's, sustainable demands and the CE in mind.





PART 3 METHODOLOGY

As mentioned in the first chapter, the construction industry should be aware of the innovation that is necessary to follow up the transition towards a CE. The goal of this research is to participate in the transition. The following chapter focusses on the explanation and elaboration of the methodology that is used for this research and how the research is executed related to the research questions and research goals, mentioned in section 1.2.3 and 1.2.4. This chapter also connects the conclusion and information delivered by the literature study to the research, which showed the gap in the literature and the possible business case of the missing middle man.

At first, the research setup is elaborated, explaining which methods are used. The next section focusses on the setup of the operational model, based on Barendse et al. (2012) and why the operational research method is applicable to the setup of the current research. The input of the model is provided by the current supply, on which is elaborated in chapter 3.3. The demand and output required for the operational model are explained in section 3.4, the feasibility in section 3.5. Finally, the execution of the model is elaborated and concluded, presenting the results of the model in chapter 4.

3.1 RESEARCH SETUP

The research started with the JIP, which is elaborated in section I.I.3. Next, the execution of the main research and answering the main and sub-questions of this graduation proposal are conducted at Alba Concepts. In this section, the structure of the research setup as conducted are elaborated.

3.1.1 INTRODUCTION

The topic of reusing building materials and the transition towards a CE contains a lot of different aspects that could be of importance. The subquestions alone are a combination of the conceptual model, providing multiple topics to be researched, before being able to answer the main research question. The different topics are researched by a conducted literature review, defining each subject that comes to the attention. This delivered an overview of theories developed together with the research question of this graduation project. The next step consisted of an exploratory analysis of the subject. The input for the model is delivered by the literature study and expertise of colleagues within the graduation company, during expert discussions. Eventually, the correct information is abducted in order to design the operational model.



Figure 3.1 Research setup small (own ill.)

The JIP is involved in the research design as well, during the internship at Royal HaskoningDHV. The given business case of Groningen delivered a research subject, a set of datasets with information about example cases and other input necessary for the operational model. The setup and extended version of this model are explained in Appendix II.2 and a representative dataset is visible in Appendix II.1. Both the exploratory research and the research during the JIP period are compared and used as input for the operational model. To design the operational model, the exploratory research in combination with the JIP provided the problem definition, consisting of the objectives, the user requirements, the constraints, and the functions. This is necessary for the setup of the model (Dym et al., 2004). Eventually, the results from the operational model provide an answer to the main and sub-questions of the research.

Figure 3.2 shows an overview of the setup of the research. The main research question is focussed on how the supply of existing building elements and demand for new construction projects can be linked in order to reduce the use of virgin materials and thereby improve circularity in the construction industry. This is the starting point of the research, which has been developed during the orientating period within the JIP. The setup of the research is separated into two different research paths. The left path focussed on the supply side, which consists of several case studies. This supply, which is

partly provided by graduation company Alba Concepts, provided a contextual situation on a certain level. As mentioned earlier by Koutamanis (2018) and in section 2.3, the possible levels of scale are project-based, region-based or enterprise-based. Every level of scale has a different set of case studies, which means different datasets and multiple projects. In section 3.3 the different case studies are analysed, the data is obtained and explained how they are connected to the operational model. This section also answers the first sub-question of the research.

The second route of the research setup model focussed on the demand side of the research question. For the variables of the operational model, certain program requirements are defined. This side consists of the conducted literature study of the subjects, such as urban mining, circularity, feasibility et cetera. On the other hand, expert discussions took place to provide information on the subjects coming from practice. It also provided arguments for the right demand for the program requirements. These two separate actions provided input for the operational model on the variables and helped answering the second sub-question.

Through multiple design stages, the operational model is analysed and tested. The goal of the operational model is to answer the main and third sub-question of this research proposal. The final design of the operational model is checked with the main research question through a feedback loop. The feedback provides information about whether the first or the second path needs to be adjusted, providing different input for the model, to reach a final, successful design that shows eventually if supply and demand fit and if it reduces the use of virgin materials. Through the feedback loop, the model is constantly updated and refined to create an optimal working design. This operational model is designed to apply in real-life cases for the stakeholders who are interested to reduce the use of virgin building materials in new projects. The conclusion of the qualities and possibilities of the model are discussed in chapters 5 and 6.



Figure 3.2 Research setup (own ill.)

3.1.2 EXPLORATORY RESEARCH

Even though the context of this graduation project consists of multiple popular topics in the construction industry such as UM and CE, there are still undefined subjects with multiple interpretations. These type of broad concepts are complex, open to all kinds of qualifications and whether or not the subject of UM will have any influence in the development of design and construction processes are questions yet unanswered. With the help of literature and field experts, these questions could be answered (Verschuren & Doorewaard, 2010). Therefore a qualitative approach is performed, existing of exploratory research. As well, the most prominent methods of data collection of qualitative approach are organisational research and unstructured or semi-structured interviewing (Bryman, 1989). The results from the exploratory research is used as input for the design research, which are elaborated in section 3.1.3.

The exploratory research is conducted in collaboration with colleagues of the graduation company Alba Concepts and the visitation of multiple conferences on circularity and CE. Through the exploratory research phase, information on different topics related to the graduation is researched and discussed with experts on the topic, creating a bulk of information that is used for design purposes. The multiple colleagues, which are specialists in the areas circularity, costs analysis, design, consultancy and the built environment provided knowledge as a base for further development of the research.

In relation to the discussions with experts and literature research, case studies are used as well within exploratory research. Case studies have often been viewed as a useful tool for the preliminary, exploratory stage of a research project, as a basis for the development of the 'more structured' tools that eventually is developed further on in this research (Rowley, 2002). This stage consists of the collection of data and the research for the housing demand in the Netherlands. According to Bryman (2012), the basic case study requires details and intensive analysis of the chosen cases. This means it will necessarily be a part of the supply side research, to figure out the constraints for a future decision-making model, based on the analysis and final decision on cases to use as input.

3.1.3 RESEARCH THROUGH DESIGN

The following section provides a concise definition of 'research through design' or 'design research', based on scientific papers and how design research is being operated in finding a solution for the typical research problems which is this graduation thesis. The most important aspect of research through design is that it seeks to provide an explanation or theory within a broader context (Frankel & Racine, 2010). It creates the possibility to solve complex problems, projects that could be defined as 'ill-defined' or 'wicked' (Rittel & Webber, 1973; Rowe, 1991). Such complex problems are difficult to explain and often formulated in a different sense, which makes it complicated to predict a solution in advance. With the help of designing a solution space, it creates the possibility to define the research problems in a more constrained area, narrowing down the context to integrated solutions (Nijhuis, de Vries, & Noortman, 2017). For this graduation research, creating design research provided the eventual solution space to solve the main research question.

Based on Van den Akker, Gravemeijer, McKenny, and Nieveen (2006) literature of multiple design research authors shows 5 different characteristics, related to this type of research:

- Interventionist: the research aims at designing an intervention in the real world;
- Iterative: the research incorporates a cyclic approach of design, evaluation and revision;
- Process-oriented: a black-box model of input-output measurement is avoided, the focus is on understanding and improving interventions;
- Utility oriented: the merit of a design is measured, in part, by its practicality for users in real contexts;
- Theory oriented: the design is (at least partly) based upon theoretical propositions, and field testing of the design contributes to the theory building.

These 5 characteristics show the qualities of the yet to be developed operational model, which is explained in section 3.2 and creates a setup of how to answer the problem stated in the research question. The goal of this research is to eventually propose and advice on how in the real world the use of virgin building materials could be reduced. The research questions also propose a comparison between supply and demand to make this happen. This creates a possible solution space, which could be reached by the development of the tool.



Figure 3.3 Cyclical process of the development of idea's, based on Van den Akker, Bannan, Kelly, Nieveen, and Plomp (2013)

The design process to achieve the characteristics mentioned by Van den Akker et al. (2013) can be accomplished throughout operational research. Figure 3.4 shows the 8 steps of the development of a particular research process (Kumar, 2011), which is used in operational research and shows the essence of combining the theoretical research steps with practical development of an operational model, to eventually produce a binding research report. The design process of an operational model is shown in Figure 3.5 by Dym et al. (2004). Combining these two schemes of a design and research process is essential in resolving operational related complex problems, such as the main research question proposes. During the process, design research is always being further developed, while new ideas and proposals are being further evaluated in every phase. This creates a cyclical process, where every cycle starts with an idea, which is represented, tested and evaluated (Figure 3.3). The proposed design goal is the main objective, but because of the cyclical process, the final design solution is adjusted towards a different angle, due to the process of designing and generating new knowledge and insights on the subject. Thus the final design would be of a different context, compared to the proposed results at the beginning of the design research (Van den Akker et al., 2013). The cyclical process of research through design is an important tool to develop models for complex research problems. It gives the author the possibility to reflect on earlier work and look from different perspectives over different moments in time at the progress of the research. Eventually the products of design-related research consists of thoroughly reflected characteristics, compared to common research techniques. The following chapters show the adaption of design research in the operational model development and final results, which provides the feedback loop to the main- and subresearch questions.


Figure 3.4 Steps of an empirical research process (Kumar, 2011)



Figure 3.5 Steps of a design process (Dym, Little, & Orwin, 2004)

3.1.4 STUDY LIMITATIONS

Possible study limitations within this research could be the dependence on the motivation for third parties to refuse to collaborate in sharing datasets. To make it possible to test the business case of this thesis, it is important that there is information available coming from practice, to be used as a test case. Different companies and universities are in possession of information about certain waste flows and available building materials but are contractually bound to privacy on the specifications of these datasets. Mostly, companies and platforms that gather and assess these datasets have a contractual agreement, which states the procedure on how to collaborate and communicate with third-party stakeholders. This is because there is still an open market around this upcoming subject, which creates competition and the need for protecting the information of the adjoining companies. To create the decision-making tool for this research, there should either be parties involved that are not participating within the competition and are open to sharing datasets, or there should be a clear Non-Disclosure Agreement (NDA) between the companies and the Delft University of Technology on how the thesis is being reported and structured. When there are clear agreements on how the presenting functions, the datasets could be shared and the business model could be tested within the practice.

3.2 OPERATIONAL MODEL

Besides the literature research and the expert discussions, a part of this graduation research contains designing an operational model or decision-making model. Operational research deals with operation-related problems, specific research that starts with a typical 'how to' question. It deals with design problems, research about new artefacts that do not yet exist. The goal is to improve a certain subject or topic through designing the operational model and improving the model by running it, again and again, creating a detailed final design (Barendse et al., 2012). This research provides a model, which uses linear programming to determine what the most optimal solution is, concerning the research goal. Within this research, the model calculates what combinations of dwellings maximizes the use of already available materials, directly minimizing the use of virgin building materials (Binnekamp, 2018). The difference in costs between the dwellings designed with reusable building materials is compared with dwellings built completely with virgin materials. Eventually, this generates results, creating overview and an advice for stakeholders whether or not executing reusing materials will improve the circularity of their project. The model is adaptable to different contexts, by upscaling or downscaling the setup and adjusting the variables and constraints. A schematic overview of the model is presented in Figure 3.6.

The input for the model is provided by an additional analysis of certain case studies. The case studies are used as an empirical inquiry that investigates a certain topic such as UM within its reallife context, especially when the definition of the phenomenon UM is not clearly evident. In other words, the case study covers the conditions of the context given, related to the study of UM (Yin, 2014).



The following section explains the setup of the concept of operational research and the creation of the model, which helps to answer the research main- and sub-questions of this thesis. First, the specifications and the methods that substantiate the setup of the model are elaborated, divided into three chapters: First the input for the model, describing the variables and constraints that are necessary to set the boundaries of the different models to be developed. Next, how the model runs its operations. In this chapter, the 'What'sBest!' add-in, which performs the calculation and the formulas behind the module, showing how the operational model can operate within Microsoft Excel is presented. The output is elaborated next and why this is necessary to determine before starting the calculation. Finally, the execution of the models are highlighted.

3.2.1 SETUP

For the operational model to be developed, multiple criteria should be determined in which the operational model performs. A set of constraints and variables provide the limitations of the model, which are part of the input. The model runs on the 'What'sBest!' add-in for Microsoft Excel, in combination with a linear programming (LP) model. The final output is determined by the constrictions and the demand coming from stakeholders. A schematic overview of the three phases is shown in Figure 3.7 and are elaborated in the next sections. The goal of the decision-making tool is to find the most optimal solution for a certain operation related problem set of variables, depending on the constraints given by a certain context. During the JIP at Royal HaskoningDHV, explained in section 1.1.3, the basic setup of the model is developed. An example of what the datasets look like, which are used during the JIP, is presented in Appendix 11.1. It shows a 3D image (BIM model), the different type of building elements (walls, floors, beam, etc.) consisting of a certain material (bricks,

steel, wood, etc.) and the quantity of that type of building element. A larger explanation of the model is shown in Appendix 11.2.



Figure 3.7 Schematic overview three phases of the operational model (own ill.)

When the scope of an operation related question is larger, the amount of constraints and variables is more extensive as well. It increases the number of possible solutions. This example model functions as an expansion tool of the usage of the materials and data available. It generates an output where for example an organisation such as the municipality of Groningen could have a lot of use of, discovering the possibilities with the materials of the current housing available, depending on the quality the new type of dwellings will be. Also, a collaboration between contractors (which could retract the materials from the buildings) and architectural firms (designing new dwelling types) could be improved with the help of this model. In this context, the model is used from a region-based perspective. However, small changes in the model could provide a working system for a project- or enterprise-based context. In each case, the model proves if the reduction of virgin materials and the improvement of circularity in design and construction projects are succeeded.

The following section explains the entire setup of the model and the mathematical theory behind the structure. The input, model and output specifications are highlighted and adjusted to the three different levels of scale.

3.2.2 INPUT SPECIFICATIONS

As is mentioned in the introduction of this thesis, it is necessary to rethink the structure of construction projects and use different databases. The number of materials that are available in our world is limited, to reuse materials from dwellings that are no longer in use is the way to go. These dwellings could be found in case studies related to this subject and available by external parties or the graduation company Alba Concepts for this thesis. These datasets act as input for the operational model, but existed of different sets of materials that could be used. In section 3.3 it is shown which datasets are used for the models and of what type of information the datasets exist of.

This section explains the different type of information used and how the information operates in the model. Figure 3.8 shows which type of information is used to define the variables and the constraints possibilities of a typical model. These constraints and variables are explained in section 3.2.2.2 and 3.2.2.3. The case studies are as mentioned provided by third parties. The discussion takes place with experts in the area of circularity and UM. The literature research provides information on how to determine which techniques are already used and how specifications of datasets are defined and validated, such as the disassembly module.



3.2.2.1 DESIGN SPACE

Creating a perfect connection between supply and demand requests an optimal configuration of the different elements. When a design has to be made, it is very time-consuming to research and develop every possible concept of the design, which is not efficient and could be performed in another way. Barendse et al. (2012) proposes a design methodology in operations research, where the alternatives to be evaluated are not known a priori. Barendse et al. (2012) uses the concept of 'design space', described by Dym et al. (2004):

'A design space is a mental construct of an intellectual space that envelops or incorporates all of the potential solutions to a design problem. As a broad concept, the utility of the notion of design space is limited to its availability to convey a feel for the design problem at hand. The phrase large design space conveys an image

of a design problem in which (1) the number of potential designs is very large, perhaps even infinite, or (2) the number of design variables is large, as is the number of values they can assume.'

This space is defined by a combination of design attributes (variables) and constraints, with defined goals and objectives by the interested stakeholders, to help in the optimization process. The design space creates an area where the most optimal design alternative for the specific problem could be found, by a combination of the different variable values.

3.2.2.2 EXOGENOUS AND ENDOGENOUS VARIABLES

To design a decision-making tool, variables and constraints are decided. The design variables are design attributes. When designing a building, the design variables could be the buildings lettable floor space, a number of floors or the ratio between lettable and gross floor space (Barendse et al., 2012). There are different types of variables within the design space. The input variables, for example, the demand for a certain number of dwellings to be built, may be referred to as parameters, or exogenous variables. These variables cannot be controlled and are referred to as X_i .

The decision variables, the variables that can be controlled by the developer, are called the endogenous variables (Barendse et al., 2012). These variables are the results from the calculation and are determined by the exogenous variables and the restrictions made by the constraints, goals and objectives of the stakeholders. These variables are referred to as Y_i , they are given, determined from the context and are immutable as far as the decision-maker is concerned (Barendse et al., 2012). An example is the Groningen case, where the materials available define the endogenous variables. Datasets that are available provide a number of building materials to use for the case. Depending on the case, if the input is different, the endogenous variables could be different as well. These variables are necessary to develop the formula that is the foundation of the calculation method, which are explained in section 3.2.3.1.

3.2.2.3 CONSTRAINTS, GOALS AND OBJECTIVES

The constraints ensure restrictions and limitations within the decision-making model. The design is created by the constraints. For example, a client could decide how much of a certain material should be used, or how many dwellings of a certain type should be built. Also, a maximum of certain materials that are available is a constraint for the model. Without constraints, there would be no solution. The goals and objectives within operations research have a different meaning. Zeleny (1982) describes the conceptual and technical differences between constraints, goals, and objectives:

'A constraint is a fixed requirement which cannot be violated in a given problem formulation. Constraints divide all possible solutions (combinations of variables) into two groups: feasible and infeasible.'

'A goal is a fixed requirement which is to be satisfied as closely as possible in a given problem formulation.'

'An objective is a requirement which is to be followed to the greatest extent possible (either by minimization or maximization) given the problem's constraints.'

Within decision-making models, there are two different kinds of restrictions, referred to as constraints. The first consists of a minimum or maximum (or both) limitations for the amount of a specific variable. For example, when designing a certain dwelling with concrete walls, the restriction of how many concrete walls to use can be a constraint. This constraint is determined by the case studies, which provide input for the model. Often this number is more than zero. The input is the number of concrete walls that are available, accessible and of a certain quality, well enough to be reused. These types of constraints are called non-negativity constraints. On the other side of the model, there are the functional constraints, which represent the total usage of a particular resource, such as the number of brick walls that are in total accessible within the case. (Barendse et al., 2012). The goals and objectives support the decision whether or not the model fulfils the wishes of the client and tries to follow the requirements as much as possible, defined by the constraints.

3.2.3 MODEL SPECIFICATIONS

When the input is defined by the context of the case and the constraints for the model are set, it is important to create the calculation model. What is noticeable within Figure 3.9 is the importance of the 'What'sBest!' add-in in Microsoft Excel. The following sections explain the procedure within this add-in tool and elaborate on the mathematical background that supports the 'What'sBest!' add-in.



3.2.3.1 OPERATIONS RESEARCH FORMULA

Looking back on section 3.2.2.2, the variables can be defined as the exogenous and endogenous variables. The relationship (f) between the endogenous variables (X_i) and the exogenous variables (Y_i) can be determined by U. This represents the utility of the system's performance and is presented in the following formula (Ackoff & Sasieni, 1968):

$$U = f(X_i, Y_i)$$

To explain the formula and the utility of the system, the case of Groningen is used as an example. As mentioned in section I.I.3, if new buildings in Groningen would hypothetically be developed, the goal would be to reuse the maximum amount of available building materials. As described in Appendix II.I, data of structural elements from a single dwelling consists out of floors and walls. The materials used for this dwellings differ between bricks (M_b), concrete (M_c), steel (M_s) and wood (M_w). If there are multiple of the same dwellings available to reuse, the formula to maximize the total amount of materials (M_t) out of the four different types of materials mentioned are as followed (Binnekamp, 2018):

$$M_t = M_b + M_c + M_s + M_w$$

The exogenous variables (X_i) in this example case are the dwellings eventually to be built. This amount is decided by the demand of the stakeholders. As explained in Appendix II.2, three different dwelling prototypes are developed. In section 3.4, the demand is researched and designed. For this basic example, the different dwelling types are acknowledged as A, B and C (N_A , N_B and N_C) (Binnekamp, 2018).

The endogenous variables (Y_i) cannot be controlled by the designer, but are based on the context of the case. If the dwelling type shown in Appendix II.I is existing of a certain amount of materials and if there are a certain amount of the same dwelling types available, the endogenous variables are a given amount. To create the three new type of dwellings, there should be a design that shows the required amount of building elements required to build the three different dwellings. This amount is determined by the stakeholders who are responsible for the demand. To explain the formula, Figure 3.10 shows the relation between the different dwellings, the available building elements and the number of elements necessary to create the I prototype of a particular dwelling. So prototype N_A requires b_A number of bricks, c_A number of concrete blocks et cetera. To define the total amount of bricks necessary (M_b), determined by the number of dwelling types (N_A , N_B and N_C) and the required amount of bricks of each type (b_A , b_B and b_C), provides the following equation (Binnekamp, 2018):

$$M_b = b_A * N_a + b_B * N_B + b_C * N_C$$

This goes also for the total amount of concrete, steel and bricks. Therefore the total amount of materials used is the sum of these 4 modules, as mentioned in the second formula.



3.2.3.2 LINEAR PROGRAMMING GENERAL MATHEMATICAL MODEL

The case at hand is a general linear optimization problem. There is a request to search for the most optimal solution, with given constraints decided by the context. The mathematical modeling technique of linear programming helps to find the most optimal solution, given an objective function and a set of constraints (Barendse et al., 2012). This technique is often used in design methodology and also the 'What'sBest!' add-in is designed to solve LP problems. These problems are often real-life problems formulated into a mathematical model. The best solution for a certain objective is chosen by the process of LP (Kashyap, 2017). According to Kashyap (2017), LP is a simple technique where we depict complex relationships through linear functions and then find the optimum points. Depict is an important word, because even though LP solves real-life problems, but they are very complex and in LP, these relationships are simplified to linear relationships.

In traditional LP, the constraints such as b_A are considered to be fixed. Often they represent physical constraints that indeed cannot be changed, such as a number of building elements located in artifacts such as dwellings we no longer use. In Open Design (Binnekamp, van Gunsteren, & van Loon, 2006) the constraints are considered to be negotiable. This is a fundamental difference with traditional LP, which means that in practice if the mathematical outcome is infeasible, it can be changed into feasible after all (Barendse et al., 2012).

There are multiple online solvers and techniques available to find a solution for the different mathematical problems in real-life. For this thesis, the 'What'sBest!' add-in of Microsoft Excel is applied.

3.2.3.3 'WHAT'SBEST!' ADD-IN FOR MICROSOFT EXCEL

The 'What'Best!' add-in needs certain input and a definition for the output. When all the data has been implemented in the Excel sheet, the 'What'sBest!' add-in automatically finds the optimal solution for the division of materials used at the input side, as well as the optimal division of the output. As mentioned before, the model used the mathematical modelling technique of LP. 'What'sBest!' combines the proven power of linear nonlinear quadratic quadratically constrained second-order cone semi-definite stochastic and integer optimization with Microsoft Excel (LINDO Systems inc., 2019). The tool is clearly designed, with an interface that can be used by everyone who is capable of working in Microsoft Excel. The modelling is fast and easy, it is free of charge and can adopt large scale spreadsheet within the calculation. Every spreadsheet created partially with the 'What'Best!' add-in is designed with a 'solve' button, automatically running the model and creating a feasible or non-feasible solution, according to the case at hand.



Figure 3.11 Operational model schematic representation 'What's Best!' add-in Microsoft Excel (own ill.)

The add-in creates adjustable cells in the top bar. These can be adjusted by the add-in in the quest for a solution. Looking at the literature, these cells are the endogenous variables (Y_i). There also has to be a cell defined as the 'Best!' cell. This represents the results of the maximum or minimum amount of materials used (M_i), depending on the demand of the research topic. The add-in only allows one cell to be maximized or minimized. This cell is the objective function of the model. Finally, the constraints in the model have to be defined. They enforce the restrictions of the model and are easily accessible within the program (Barendse et al., 2012). Figure 3.11 shows a schematic representation of a conceptual model, with the Groningen case used as an example to define the input and output.

3.2.4 OUTPUT SPECIFICATIONS

In operations research, specifically in the 'What'Best!' add-in of Microsoft Excel, the model creates, based on the input and the given constraints, a certain output. This output differs for each case. The output is generated by a solver of an LP related problem. It is very important the model works, the formulas are entered correctly and the different constraints are defined in a correct matter. By pressing the solve button it generates a solution of the model. Because the input has been limited so far in this example, the results are limited. Given an optimization variable in the model, the solution becomes more complex and creates different possibilities, trying to get the variable maximized or minimized. When there are more variables entered, it means the factors that affect the solution space will grow, not necessarily meaning there are more solutions, but better substantiated. Only when different variables are added, it creates more different types of solutions, connected to a variety of variables.

Looking at the example of the Groningen case and the setup for the following research of this thesis, the output of the example decision making model consists of the following subjects, visible in Figure 3.12: (I) the amount of building materials used from the input of the case studies, (2) the amount of prototype dwellings that could be built with the given datasets and (3) an overview of costs combined with a comparison between prototype dwellings constructed with reusable materials and virgin materials. To generate these numbers, the case input has to be defined by the case studies and the prototypes of the dwellings to be built have to be defined in materials, size and costs. Sections 3.4 and 3.5 elaborate on the setup of the demand and the feasibility.



Figure 3.12 Schematic overview output (own ill.)

3.2.4.1 STAKEHOLDERS

The output should be determined by collaborating stakeholders. The development of the decisionmaking model could be of help to combine the wishes of multiple stakeholders related to a certain region or enterprise, acting as the middle man and improving the process of reusing materials between different organisations. It is important for the business case of the model to have as many organisations as possible involved in the process. As mentioned in section 1.2.2, if the information of accessible building materials and demand in housing projects is not shared, it creates more difficulties to empower the transition towards a CE, specifically in dwellings projects. Thus, the importance of sharing information and creating a clear overview of demand could accelerate the process. A further elaboration of the role of stakeholders is presented in section 3.4.3.

3.2.4.2 COMPARISON OF FEASIBILITY FACTOR

The main research question of this thesis is focussed on how the supply of existing building elements and demand for new construction projects can be linked in order to reduce the use of virgin materials and thereby improve circularity in the construction industry. The comparison is of importance in answering the main research question, by creating insight in whether or not reusable building materials are cheaper compared to virgin building materials. This will help to generate the first steps to substantiated results. The feasibility of the results is of importance when considering realizing projects with reusable materials. The margin of difference between the virgin dwellings and the reused dwellings determines whether or not it is feasible for an organisation to initiate reusing materials. On the other hand, if the margin is acceptable (less then or similar to 0%), it results in multiple interesting organisations to create dwelling projects with reusable materials, automatically stimulating the market secondary raw materials, but also cheaper deconstruction techniques and innovation on tools for reusable materials. The comparison is based on factual expenses on factors such as transportations, construction, deconstruction et cetera. In section 3.5 the numbers on feasibility are presented.

3.3 SUPPLY

Operations research focusses on finding a solution for questions starting with 'how to'. Mostly, these types of questions are linear programming problems, based on a mathematical model of a real-life case. To research problems from real-life, data is necessary coming from a context. As mentioned in the research design in section 3.1 and in multiple other sections, case studies are used as input for the operational model, to determine the supply side of the model and making the connection to real-life problems by implementing data from real-life cases. According to Bryman (2012), a case study entails the detailed and intensive analysis of a case. Within case studies, it is possible to use a single- and multiple-case study designs. For this research, multiple case studies are applied. The context of these case studies are the input of the operational model. It is necessary to define the level of scale on which the possibile outcome of the model is based on. The organisational level of the case is key. The possibilities are divided into three different levels of scale, as explained in section 2.3: Project-based, enterprise-based or region-based (Koutamanis, 2018). The decision on which level what kind of model are executed is depending on the available datasets and whether this is realistic or not. However, it is also depending on the demand, which are elaborated in section 3.4.

The input for the model, existing of building information datasets extracted from case studies, are provided by multiple different companies. During this graduation, the gathering of information turned out to be an interesting struggle. Multiple companies and organisations are in the possession of usable datasets. But, as mentioned in section 1.2.2 the construction industry has trouble creating transparency between stakeholders, which are the same struggles that happened during the data gathering. Due to privacy regulations, trust problems and an incomplete overview of substantiated arguments, sharing information and sharing datasets for study-related research were not always possible. This is why the data used within the three models are sometimes incomplete and sometimes hypothetical, based on estimations or public figures. The graduation company Alba Concepts provided a sum of projects, which are evaluated and checked whether or not they can be used as example cases to test the conceptual model.



Figure 3.13 Cinderela project conference Amsterdam (Pranjić & Mladenović, 2018)

Also, during the internship, a number of conferences have been visited to connect with interesting parties on the CE. In Amsterdam at the Advanced Metropolitan Solutions Institute (AMS), a conference was held on the H2020 Cinderela project (Figure 3.13), discussing the possibilities to create new CE business models for more sustainable urban construction. This conference was a collaboration between the Delft University of Technology, AMS and KplusV. During the Circular Economy Festival in the city of Nijmegen (Figure 3.14), many companies were joining, discussing the challenges to face to transform to a CE. These conferences connected different initiators of the CE development and provided new ideas and goals that could be reached together.



Figure 3.14 Circulaire economie festival Nijmegen (Vonk, de Jager, & Dirkx, 2019)

3.3.1 DATASETS STRUCTURE

Operational research tries to solve complex problems based on real-life context. This given context should provide the correct information that could be used to test the model developed so far and create a realistic result that could provide advice for interested stakeholders. This supply could exist in many different structures and quantities. This section provides information on the different possible datasets that could be used for models in the different scale levels, described in section 2.3. As explained by Bryman (2012), a basic case study entails the detailed and intensive analysis of a single case. To use the correct data from a context a basic case studies analysis is necessary to create a



Figure 3.15 Shearing Layers (Brand, 1994)

database, in order to eventually use the datasets as input. Many information sets could consist of an overkill of data, which makes it necessary to limit the analysis to the primary data, to use as input for the model. The shearing layers dwelling structure designed by Brand (1994) (Figure 3.15) is an excellent reference to limit down the datasets to the necessary building materials for the operational model. The six different layers (Stuff / Space plan / Service / Skin / Structure / Site) with each different lifespans are well-known criteria in the circular construction industry world. For this research, the materials used for the model and collected from the datasets consists of the 'Skin' and 'Structure' layers, mostly structural and external materials. Due to the shorter lifespan of the other layers and the less information available in the chosen datasets, the other layers are left out within this calculation and preserved for future research possibilities, which are discussed in section 5.2.

Next to the chosen layers of Brand (1994), the datasets should consist of the following criteria, in order to provide sufficient relevant data in a clear dataset overview, as visible in Figure 3.16.



Figure 3.16 Database structure (own ill.)

Type of building materials (horizontal)

Depending on the context, it should be clear what type of materials are included within the datasets. This could vary from basic material elements, such as bricks, to more complete elements, such as a prefab wall. A distinction should be made between the different types of data and type of elements.

Amount

A clear amount of every type of building elements that are being provided by companies or cases and are available for reuse. Also, information about material flows and predicted incoming materials.

Locations / distance to construction site

Due to privacy regulations, sharing information on locations within datasets is difficult. For this research, it is not necessary to be of knowledge which company is in the possession of the material or to which company it is transferred to. Preferably, the end location of the material after transportation is shared. Just the city or the village is adequate, which eventually provides a distance in kilometres.

Reusability potential of material

The reusability of scale is a valuation method that is used for deciding for each building element whether or not the element is fit to be reused. This method is provided by the analysis of the materials and mostly the qualities of the chosen elements should be observed visually to determine whether or not it fits the reusable scale. It depends also on the type of interface and connection with surrounding materials if the material is fit for reuse. These different type of factors are important to determine the costs for reused building materials and the influence of the quality of building elements in relation to the costs for virgin building materials, which is elaborated in section 3.5.2. The levels of reusability are determined by graduation company Alba Concepts and are visible in an example project in Appendix II.4.

Dimensions

The dimension of the materials available for reuse. If the dimensions of a material are possible to analyse, it could be of big assistance to discover whether or not the use of these elements is realistic. In the type of materials the thickness of structural elements is given, so dimensions length and width could be extra information that could improve the realistic output of the eventually developed model. However, dimensions also create a less wide solution space, if taken into account that every single building element is limited because of its given dimensions, which would make it more difficult to supply hypothetical advice.

Description

Not every building element is the same. Each set of materials should have an extra column to mention comments or a broad description, in order to add any additional information to create more insight into the datasets and improve the input for the model.

Building elements (vertical)

The vertical column of the datasets consists of the different building elements that are being located at the case studies. These columns are divided into building levels from the bottom up. So starting at the foundation, following by floors, walls, roof and other types of building elements that are located within the dwellings or other projects related to the context.

3.3.2 PROJECT-BASED CASES



The first of the three levels of scale is the smallest, the project-based cases. As described in section 2.3.I, the model is based on a small set of projects, which are relatively close by. There are just a few stakeholders involved and in this case, there is one location where the new project should be built, which would provide the different distances for the various locations. To operate a model on a project-based level, a few cases with detailed information of all the different materials available within those cases are necessary. As an example of different cases that could provide the correct datasets, Alba Concepts provided four different projects that are described in the next four sections, divided over the Netherlands (Figure 3.17). An example of a dataset developed by Alba Concepts and used as hypothetical real-life case input is shown in Appendix II.4.



Figure 3.17 Overview of project-based cases in the Netherlands (own ill.)

3.3.2.1 BO-EX APARTMENTS UTRECHT

Bo-Ex is a housing association located in Utrecht, which offers living space to around 9.000 households (Bo-Ex, n.d.). They have the ambition to participate in a more sustainable world and are influenced by topics such as energy, sustainability and circular economy. In collaboration with Alba Concepts, Bo-Ex designed a circularity strategy, existing of different themes, critical key performance indicators (KPI's) and multiple verification tools. Also, a number of 14 apartment blocks that are outdated are inventoried on the different type of materials, put together in a database. This created a clear insight for Bo-Ex on which materials could potentially be reused or recycled, which makes it a small but extensive dataset.



Figure 3.18 Bo-Ex (Bo-Ex, n.d.)

3.3.2.2 PROJECT DOEN! STATIONAREA ARNHEM

The second project-based case provided by Alba Concepts is 'Project Doen!', which includes the station area of the city of Arnhem. In collaboration with the government, Alba Concepts was involved in a project to refurbish the entire western part of the station area, which consists of 10.000 m² BVO. Alba Concepts has been asked by the government to join and advise in the quest to reach the ambitions, related to sustainability and the energy transition. The government separates the ambition circularity into two focus point for the case (Boer, 2018):



Figure 3.19 Rijksoverheid (Boer, 2018)

Circular materials (usage of materials):

- New or reused materials at the beginning of their life cycle (in percentages);
- Reusability of material at the end of the life cycle of the building;
- The technical life cycle of materials (in years).

Circular adaption, detachability (losmaakbaarheid):

- Technical detachability, based on accessibility of connection;
- Physical detachability, based on the type of connection.

To research the different types of circular ambitions, the buildings that were about to be renovated had to be analysed, thus creating a database of the materials present in the buildings, with the given potential of reusability. This database eventually consisted of multiple different types of materials compared to dwelling projects, but still most of the information available could be of use for the model in a different scale or if there is a different demand, for example office buildings or renovation projects. For dwelling projects, the given dataset is not sufficient enough, because it's lack of structural elements and big diversity of different materials.

The example database of 'Project DOEN!' is located in Appendix 11.4.

3.3.2.3 STADSWINKEL TILBURG

The third project-based case is executed by a collaboration of the municipality of Tilburg, Alba Concepts and 'W/E Adviseurs', a sustainable consultant in circularity, project management and project development (W/E Adviseurs, n.d.). The project contained a circular inventory of the building the 'Stadswinkel' of Tilburg, part of the municipality. This resulted in an intensive dataset, divided into multiple sheets, consisting of a BVO of over 21.000 m². The municipality of Tilburg showed their ambition in circularity by pointing out of three main objectives:



Figure 3.20 W/E Adviseurs (W/E Adviseurs, n.d.)

- Reusability of materials within the 'Stadswinkel' itself. The research should indicate advice regarding a select group of building materials and elements that show potential to be directly reused in renovation projects.
- The second possibility for reusability of building elements is within the portfolio of the municipality of Tilburg. For example setting up of materials bank that could provide different projects in the surrounding area of materials, based on similar examples in surrounding municipalities.
- The third option could be in a setup of a possible market place for building materials, such as New Horizon or Cirdax, but this idea is being parked for possible follow up research and action points.

Eventually, the goal for this particular project is to design a material passport of the different materials within the building that have the potential to be reused in future renovation or construction project. If this happens, it would create an innovative database with constantly updated information on the possibilities of reusability of the different materials. However, the current datasets, even

though they are very extensive, are also limited to a small stock of building elements, which makes them not relevant for new dwelling projects.

3.3.2.4 DE ALLIANTIE

The fourth project is the only of the set of the project-based datasets that hasn't been executed yet. The case has the potential to create a large set of realistic databases that would be perfect to execute and use as input for the model. The main objective of de Alliantie, which is a housing corporation in the Netherlands with a portfolio of over 50.000 dwellings, divided over multiple cities in the Netherlands (de Alliantie, n.d.), is to analyze their complete portfolio. Together with Alba Concepts, the entire portfolio is divided into I2 different archetypes, based on building age, construction method, isolation



Figure 3.21 de Alliantie (de Alliantie, n.d.)

values and installations. These 12 different types could easily form an extensive database with many building elements. Unfortunately, due to financial restraints at de Alliantie, de inventory phase didn't take off during the graduation research, but the concept of using an inventory of a portfolio of a housing association offers a lot of potential for the model and further analysis of future projects.

3.3.3 REGION-BASED CASES



While the former section described the context for the model on a project-level, this section takes a look at the possibilities when using region-based cases. Regions create more possibilities for input and output, because of the larger scale of the context. Often municipalities are involved, to regulate the flow of materials and connect different stakeholders and clients as the project manager of the case. The municipality could act as the middle man, connecting demolition and deconstruction companies to research the locations and inventories the materials, therefore connecting them directly to local housing corporations with the ambition to initiate in sustainable and circular dwelling projects.

3.3.3.1 EME-PLATFORM

The goal of this graduation research is to improve the transition to a CE, by reducing virgin building materials. A single project in a small area does not fulfil the demands for the transition. A context that focusses on a regional level would make a bigger impact in the transition. If more materials could be used as input for a certain model and if the stretch of a certain area for new buildings is bigger, a larger amount of materials are reused and lesser virgin materials are used. However, it is not possible to match all the materials directly with new projects, so it is necessary to find a project or scope from a certain case where it is clear what boundaries are at stake, what kind of materials are used as input and what the demand is for new projects in a certain area. A platform of a kind could be the solution of organizing the connection. An example case is the Excess Materials Exchange Platform (EME), the digital facilitated marketplace that is highlighted in section 2.4.4 by Damen and van Maaren (2018). The scope of the EME platform is the entire Amsterdam Metropolitan Area (AMA) and its surroundings (Figure 3.22). The construction sector within the AMA is used as a scope. Within the building sector, there is a large list of materials which are available to reuse, which are visible in Appendix II.3. By urban mining these materials, a large database could be created that eventually could be used for new construction projects. The problem is the missing of the connection between

the data and the constructors. Also, there is no direct connection between the demand for new construction projects and the supply of the building waste. This connection could be made with the execution of the materials in this region by the digital marketplace of the EME platform.

The EME-platform consists of a collaboration between Alba Concepts, Copper8 and EME. In a hypothetical situation where the tool developed in this research is being connected within the EME-platform, the following subjects show the possible topics to further develop the platform, which connects to the matchmaking module of the platform in specific, also visible in Figure 3.23:



Figure 3.22 Enterprise-based region: Amsterdam Metropolitan Area (own ill.)

- I. Gathering datasets of building materials, providing input for the decision-making tool.
- 2. Exploring the possibilities of different building materials.
- 3. Exploring the demand side of the research through panel groups with stakeholders.
- 4. Testing the matchmaking module by the use of case studies.
- 5. Researching the possibilities within the technical and organisational scope of the platform.
- 6. Testing the feasibility of the matchmaking module.
- 7. Designing a tool within digital innovation.
- 8. Constant updating and improving the decision-making tool.



Figure 3.23 EME-platform possibilities in innovation, based on Damen and van Maaren (2018)

These 8 topics are not definitive, but provide subjects that could be researched during a potential collaboration with the organisation of the EME-platform in the future. The results of the tool could be used within the platform as an advisory tool for partners connected to the EME-platform and located within the AMA. Unfortunately, the collaboration between Alba Concepts and EME did not pursue the development of the operational model based on this research, due to lack of reaching a working agreement and collaboration structure. The reasons behind the failed collaboration are explained in the discussion, section 5.I.

3.3.4 ENTERPRISE-BASED CASES

The third level of scale is one of the largest. The enterprise-based cases, as described in section 2.3.3, focus on a system or a typical material flow in a certain region, instead of one or a few small projects within an area. The scope size of this type of system is much larger, which means stakeholders that are in the possession of nation or even European wide datasets are necessary to create a working model. In the next two sections, these types are highlighted.

3.3.4.1 CINDERELA

One of the most interesting projects that comes to mind is the project called Cinderela, a collaboration project between 13 partners from 7 countries in Europe. Their goal is to design a new CE business model for more sustainable urban construction. The partners are a combination of Universities and private companies. The project has received funding from the European Union's Horizon 2020 research and innovation program under grant agreement No. 776751(ZAG, 2016).



Figure 3.24 Cindereld (ZAG, 2016)

The Cinderela H2020 tool consists of a database of all the materials that are transported from a first to a second company, due to construction, demolition or other activities. The companies that processed the waste had to record all the transition between their selves and partner, by law. The European Union gathered all the information and created an impressive dataset. Every transition of material is registered and described with code of 6 numbers, which present the type of material, the condition and an extra section with additional information about the quality of the material. Figure 3.25 shows an overview of a visual representation of a flow of materials in a specific area, based on all the different transitions in one year, in this case the year 2016.





Figure 3.25 Cinderela H2020 Complete CDW flow originating from the AMA in 2016 (B. Geldermans et al., 2019)

The sets consist of every type of material that is involved in construction activities. The specifications of the materials differ in extensiveness depending on information such as dimensions, quality and validity. These datasets, due to the completeness and accurateness of the Geodesign Decision Support Environment (GDSE) system, provides a tool which could locate and filter the necessary materials, which are available within the datasets. This tool is developed under the H2020 project REPAiR (REsource Management in Peri-urban AReas: Going Beyond Urban Metabolism) (Pranjić & Mladenović, 2018).

The Cinderela organisation focused on the following three pilots, whilst creating the datasets and organizing experts' conferences such as described in section **3.3**:

- Extraction of valuable materials
- Production of secondary raw materials (SRM) based on construction products
- Large scale demonstrations of construction with SRM based construction products

The information about the materials is confidential, which means the use of the datasets should be protected under an NDA agreement, to protect the graduation research and the information about the materials.

3.3.4.2 CIRDAX / CIRKELSTAD / NEW HORIZON

The following companies show small potential to be a possible partner in future collaboration for enterprise-related projects, but did not participate or did not provide enough information to be relevant for the research.

- **Cirdax:** Material management system (elaborated in section 2.4.3). Did not provide access to the system.
- **Cirkelstad:** Mostly focusses on developing cards with possibilities of materials, not an assessment of buildings particularly.
- New Horizon: Demolition and investing company, consists of datasets, which could be possible for use as input in the model, but the datasets lack detailed information.



Figure 3.28 Cirdax (Re Use Materials, 2019)

3.3.5 CONCLUSION



(Büch et al., n.d.)



Figure 3.28 New Horizon (Verkuijlen, 2015)

The different cases that are researched as input to test hypothetical operational models turned out to be a bit of a struggle. Most of the potential partners lacked corporation, due to privacy regulations. It resulted in a much smaller amount of datasets that were possible to use for the operational model than expected. The consequences were that this research changed the focus towards a realistic output instead of the input. The operational model should be developed in a sense that it is ready to operate, which means that if there are datasets becoming available to use, for example access to the Cinderela project, the model should be designed in a way that data could be directly implemented.

3.4 DEMAND

Based on the research design presented in Figure 3.2, one of the two paths in the figure is the demand side. With the literature study conducted on material definitions and management systems, the focus within this section is based on the demand for new dwellings and reusable materials within the current dwellings. It has been known for a while that circular materials in dwellings are necessary for the future. As mentioned by the Rijksinstituut voor Volksgezondheid en Milieu (2016), the challenge is to design buildings in such way that all of the materials in them are suitable for high-quality reuse. However, the long life of building structures – 50 to 100 years – makes it difficult to determine how the materials will be dealt with in several decades of time. So it makes sense to analyse the current standard dwellings and determine which materials have the potential to be reused. Also, as mentioned by one of the founding partners of Alba Concepts Jim Teunizen, circular construction projects with reusable materials integrated into new dwellings will be normal, way sooner than expected (Gemeente Amsterdam et al., 2017). This means innovation in circular dwellings is necessary within a short time frame.

"Circulaire bouw gaat, veel sneller dan wij denken, de standaard zijn."

Jim Teunizen, Alba Concepts

So different, both private and public, parties agree on the fact that reusability of materials in dwellings should be standardized and has the future in the construction industry. However, whether or not this transition is feasible for municipalities or housing corporations is an important question, to make this transition of the usage of materials happen. This is where the operational model enters. As explained in section 3.2, the model needs input and generates an output. In the following section, a part of the input is elaborated and how this particular part is related to the final result of the operational model.

3.4.1 DWELLING PROTOTYPES REFERENCES RVO

There are many different dwellings in the Netherlands, which makes it impossible to choose a certain standard dwelling to analyse. Also, a lot of new dwellings are necessary to be built in the future in the Netherlands (ANP, 2019) and it is not even certain if it is possible to complete the demand for dwellings in time. This raises a bunch of questions. What type of dwellings will be built in the future, which materials will be used and why will certain materials be left out of future construction projects? Not all of these questions could be answered yet, so it is important to scale down the analyses of existing dwellings and building materials to a level that is manageable and where already information is available. Fortunately, the RVO (Rijksdienst voor Ondernemend Nederland) provided, in collaboration with Royal HaskoningDHV, six different types of standardized dwellings that nationwide can be used as reference dwellings:

- Apartment block (Krijnen, 2015a)
- Gallery dwelling (Krijnen, 2015b)
- Corner dwelling (Krijnen, 2015c)
- Standard row-house (Krijnen, 2015d)
- Semi-detached dwelling (Krijnen, 2015e)
- Detached dwelling (Krijnen, 2015f)



Figure 3.29 Schematic dwelling analysis overview of Excel model (own ill.)

These documents contain all the practical information about the dwellings. Within these six types of standardized dwellings, three have been chosen to be analysed, which are the standard row-house, the semi-detached dwelling and the completely detached dwelling. This choice is based on the fact that the apartment block focusses on multiple dwellings instead of one, which makes it more difficult to determine the number of materials and costs for a single dwelling. This goes as well for the gallery dwelling, which is similar to the apartment block, information on the complete building is lacking. The corner dwelling is almost similar to a combination of the standard row-house and the semi-detached dwelling, so it will not provide any additional information. This leaves the three dwellings to be analysed in the next three sections. The dwellings are analysed on a set of topics related to the materials used within the building, which are of use to determine the right quantities of materials which have the possibility to be reused within the dwellings. This schematic overview of topics is visible in Figure 3.29.



Figure 3.30 Standard-house dwelling, semi-detached house, detached house (Krijnen, 2015d, 2015e, 2015f)

The three different types of dwellings are visible in Figure 3.30. As noticeable, they are similar in materials and height but differ in materials used in the façades, square meters (NVO and BVO) and volume. The overview of specifications and other information regarding the three dwellings are visible in Appendix II.5, II.6 and II.7. These three types of dwellings are used to perform within the project-level model, to create a well-substantiated tool that has the possibility to be up scaled. If so, a significantly larger amount of dwelling types are necessary to use as output for a model, to create diversity and different solutions. The RVO provided another document, which consists of an overview of existing standardized dwellings, based on energy performance and construction year. Over 30 different dwellings are presented, with detailed information on every aspect related to the dwelling (W/E Adviseurs, 20II). If the scale of the project would expand, these types of documents become necessary to define the correct output for the operational model.

3.4.2 REUSABLE BUILDING MATERIALS

Dwellings contain multiple building materials and elements, but not every element is fit for reuse. There are certain factors that determine whether or not materials could have a second life cycle. This is different for every element inside a dwelling. It depends on the different type of materials the element consists of, but also on the interface between the different materials and what kind of connection between the two different materials has been used to connect the interface. This directly reacts to the detachability of building materials, which influences the quality of the material when it is detached or wrecked. The factors that decide how much materials are being influenced by accessibility of connection, detachability of connection and other specifications for materials is visible in Appendix II.8, which is an overview designed by Durmisevic (2006). When looking at a certain standard dwelling, which has been done in section 3.4.I and the related Appendices, combining the list of materials to the Shearing Layers concept shown in Figure 3.3I of Brand (1994), the layers 'structure' and 'skin' are, as mentioned earlier within this research, the main layers to focus on for the reusability of the materials.



Figure 3.31 Shearing Layers (Brand, 1994)

Looking back at the section where the three reference dwellings have been analysed, a set of six different types of materials show the highest possibility of reusability. The choice on these six types of materials is based on an analysis of the present building materials in dwellings in collaboration with colleague Sven Bögels of Alba Concepts, who is an expert on construction costs calculations and the entire work area of feasibility studies and construction management on dwelling- and utility projects (Bögels, 2019). Figure 3.32 shows six different elements. The type of materials that are used for the elements within the dwellings are based on existing details, which are based on the 'SBR-Referentiedetails' for dwellings (Nieman, 2015). These details consist of exact information on what type of material is used for every particular part of every element, which directly connects to the following six different materials:

- Floors: Channel plate floor on the ground level (150mm thickness) and the first or second level (200mm thickness);
- Exterior wall: Prefab concrete walls (Residential partitioning wall) and Clickbricks (Façade);

- Interior wall: Prefab panels (Ytong or Faay panels);
- **Roof:** Ceramic roof tiles;
- Doors: Corridor doors, Façade doors (front) and French door (garden);
- Windows: Meranti frame.

The prices and costs of these virgin or reused materials are compared in section 3.5 and are visible in Appendix 11.9.



Figure 3.32 Six different building elements with high potential for reusability (own ill.)

3.4.3 PARTICIPATING STAKEHOLDERS

Different questions come to mind when discussing the possibilities of reusing materials for new dwelling project. What is the current housing, renovation or other projects demand in the construction industry? Which parties are willing to collaborate and have the vision to reuse materials instead of virgin materials? What is the current housing demand of the bigger regions in the Netherlands, such as the AMA? The tool which is created has to react on important variables, such as the influence of the stakeholders involved.

To combine the wishes of multiple stakeholders involved related to big housing projects, deleting the middle man and improving the process of reusing materials, as part of the matchmaking tool, it is important to analyse which stakeholders could be involved and in which phase of the process. Also, what the eventual goal of the tool could be for the different stakeholders. For example, it could provide advice for a municipality, to show the possibilities of reusing materials within their region and promote the possibilities to housing associations. Costs and locations factors implemented could provide the setup for an entire structure of housing projects, almost completely designed with reusable building elements and materials. In this case, the municipality would act as project manager, connecting the different housing associations that have sustainable goals to each other and creating projects in which big housing companies could collaborate. The tool would be used in the initiative phase, which creates first insights into the possibilities for new dwellings and helps the municipality in convincing housing associations to cooperate.

The different level of scale also has an influence on the participation of different stakeholders. On a project-level, the only stakeholder involved could be the owner of the dwellings to be demolished, which has the objective to create new dwellings close by. However, when the scale of the project gets larger, multiple locations will provide input in materials. These different locations are frequently in the possession of several companies, so they will have a say within the transition of ownership of materials. Also, multiple deconstruction companies will have to be involved to provide the materials and multiple companies are involved in transportation or storage of these materials. Next, when construction is starting, the tool calculates the possible dwellings to build, but on a larger scale there are multiple locations instead of one. So multiple housing associations or municipalities are involved to create enough construction sites on different locations, which generates even more stakeholders, such as local communities, different construction companies and other housing corporations. If the size of the scope grows from a project-based to a region-based level, the growth of the number of stakeholders should be taken into consideration as well, especially focusing on adjoining costs on ownership, man-hours and legal contractual agreements. The government can also play a role whether or not the tool will provide feasible advice. Due to legal regulations, it could withhold the business case of reaching its full potential. The government could prevent deals between stakeholders to be made because of ownership of ground and land-use plans. The model, when designed on a region- or enterprise-based level, should have incorporated these regulations to see how the outcome reacts to these influences and whether or not the business cases are still feasible.

3.5 FEASIBILITY

The following section elaborates on the feasibility of the study and the operational model. As mentioned multiple times within this research, the goal is to reduce the usage of virgin materials in new dwelling projects. A method to reach that goal is to gain confidence by private stakeholders to invest in reusable material projects. The companies that are interested in reusing building elements should have a clear view on the cost overview of virgin and reusable materials. How the numbers are being decided is elaborated within the following sections. First, the virgin building material costs are determined, based on practice and prices used by contractors in current traditional projects. Next, based on input generated from the current innovation on circular economy of the construction industry, a substantiated calculation on the costs of reused building materials is presented. This chapter also shows the sources of the calculation that are used as input. These sources consist of hypothetically existing dwellings, with a set of building materials at the location and a certain distance connecting the old and new location.

Costs in the construction industry is an extensive topic. The multiple traditional building phases of dwelling projects, starting from initiative to design, followed by construction and finally the maintenance of the project, have a lot of expenses related to the project. Also, the number of stakeholders in large projects, as mentioned in section 3.4.3, brings along many factors to take into account. So for this research, it is necessary to define a scope on which the calculations are limited. Figure 3.33 shows the different costs that are components of a building project (Gemeente Amsterdam et al., 2017).



Investment costs

Figure 3.33 Investment costs of a construction project (Gemeente Amsterdam et al., 2017)

Direct construction costs (MAMO: Materiaal, Arbeid, Materieel en Onder aanneming) (Figure 3.34) Indirect construction costs (AUK; Algemene uitvoeringskosten, AK; Algemene kosten, W&R; Winst & Risico, CAR en prijsstijgingen) This is just a small conceptual overview of the most general costs. For this research, the direct construction costs are taken into account into the calculation of the costs for virgin and reused materials. Due to the fact that indirect costs probably are similar whether or not the building is constructed with either virgin or reused materials on a project-based scope, the indirect and additional costs are not taken into account. Also, labour costs such as man-hours will probably be similar for the different materials as well, so these are not adopted into the formula as well. The projects which are initiated with reusable materials will eventually show what kind of additional and indirect construction costs are at hand. But because the still yet to be executed projects, there is not much information on budget overviews of these types of projects yet. There simply haven't been that much projects completed yet, in which the usage of reusable materials had been taken into account from the beginning to the end. This makes it not possible to use the number of costs from traditional projects and implement them directly into the costs of projects initiated with the goal to only focus on reusable building materials. So for this research, the direct construction costs are only taken into account.

Direct construction costs (MAMO)



Figure 3.34 Direct construction costs (MAMO) (Gemeente Amsterdam, Concepts, & Copper8, 2017)

This goes the same for the different aspects of the direct construction costs. As shown in Figure 3.34, the construction costs consists of the building materials, the labour performed by workers, the equipment used and the usage for subcontractors. The last three topics will probably have the same kind of price budget compared to traditional construction projects, but due to the fact no projects of worth mentioning scale of context being executed yet, these costs will not be taken into account.

It is recommended to adopt the indirect costs into the model as soon as possible in cases with different size of scale and amount of stakeholders, to make sure the financial output of the model is as realistic as possible and directly adaptable into cases. Multiple stakeholders will bring extra costs into the calculation as well, such as transition of ownership, contractual agreements and other legal-related subjects.

3.5.1 COSTS VIRGIN BUILDING MATERIALS

The traditional construction methods use overall virgin building materials for new projects. The costs for traditional building methods are located on online platforms with an overall view of the different types of costs. These overviews are based on real-life projects, but the numbers differ in context, related to the



Figure 3.35 Cobouw (Cobouw, n.d.)

different elements regarding the source. For this research, most of the traditional virgin materials the source of the costs is located at 'Cobouw Bouwkosten' (Cobouw, n.d.). This website shows an overview of almost all the different elements and materials of different sizes and types. If the specific material is not located at Cobouw, other websites are referred to for the costs estimation, mostly contractors or companies that deal in the same materials as well.

Based on the 'SBR-referentiedetails', the correct type of materials could be determined that are located in the dwelling prototypes, as elaborated in section 3.4.I. Afterward, the costs per square meter [€/m²] of each material is cross-referenced with the different suppliers and the correct type of material. The materials listed at Cobouw do differ in specifications regarding the total amount of costs per square meter. For example, the specifications of the costs for prefab concrete exterior walls consist of every small detail in constructing the facade. While in this research, only the direct material costs are taken into account. So the recipe of each material on Cobouw has been checked and labour costs or other additional aspects are left out. Appendix II.9 shows an overview of all the different types of material for each element and the related costs for every type of material per square meter, visually represented in Figure 3.36. The strokes in the overview in the Appendix that are marked in green have the closest connection to the design of the details, based on the dwelling prototypes, so these are chosen as reference for the calculation of the costs.

In this overview, the lifespan of the chosen elements has been attached as well. It is important to realize the degradation of the quality of materials due to aging. The lifespans of the materials are based on the website 'NIBE, Experts in sustainability' (NIBE, 2011), which has detailed information of almost every structural building product in every layer of buildings. Next to lifespans, NIBE has information on subjects as environmental class, general transport distance, weight per functional unit. The reason why lifespans are attached within the information overview of the virgin costs is related to the calculation of the material validation of the reused materials, which are elaborated in section 3.5.2.3.



Figure 3.36 Virgin building materials costs schematic overview of Excel table located in Appendix 11.9 (own ill.)

3.5.2 COSTS REUSABLE BUILDING MATERIALS

The second section of the feasibility study of this research concerns a topic that is still yet to be developed, also in practice. The calculation on how to determine costs for reusable structural building elements is a discussion that is still going strong. Multiple companies and research organisations are cracking their heads on the different factors that should be taken into account when calculating the costs. A few of these factors are discussed and presented within the following section. The difficulty of calculating costs for reusable materials depends also on information coming from practice, but the problem is there are not that many large scale projects executed yet, which involve

the reuse of building materials on such a large or complex scale, meaning creating almost an entire dwelling out of structural materials. This means the formulas to determine the costs are based on assumptions, substantiated from literature and reports created by experts in the field. Also, as shown in Figure 3.37, at the 26th of August 2019, a discussion group took place with the colleagues from Alba Concepts, which are experts of consultancy in circularity, sustainability and project management. During the discussion, the different factors that influence the costs of secondary building materials for new dwelling projects are being discussed, from expertise of personal knowledge and reference projects. The comments and insights are being implemented in the following sections as well.



Figure 3.37 Discussion session experts Alba Concepts 26-08-2019 (own image)

3.5.2.1 DECONSTRUCTION

In the process of urban mining building materials from existing dwellings, the first obstacle to overcome is to deconstruct the dwellings to obtain the different materials. Deconstruction costs are tricky to determine, due to the many different techniques of demolition, the many different materials and the different interface between the materials which makes it more difficult to obtain the value of the material. Also, multiple companies that are active in the field of the demolition of buildings, are not used to the process of deconstructing buildings to save the materials but mostly just demolished the dwellings to a state where it could be only recycled or down cycled.

The market on reusable materials is not at the level yet that every demolition company focusses on maintaining the value of building elements. This means there are not many correct feasibility studies or calculations performed yet on this concept. Though there is an ongoing project, which is a research of a collaboration between Alba Concepts and 'De Nederlandse Organisatie voor toegepast-natuurwetenschappelijk onderzoek' (TNO), commissioned by Madaster. This research focusses on calculating and determining the valuation of commodities or recourses (Madaster et al., 2018). The formulas used to determine the deconstruction costs can almost be directly applied within this research, regarding some minor changes or a different interpretation of the factors. Figure 3.38 shows the formulas and the different elements that influence the final financial outcome, which is created within the research of Madaster et al. (2018).



Figure 3.38 Deconstruction costs formula, total deconstruction costs formula (Madaster et al., 2018)

The first formula calculates the entire demolition costs for a building. As a reference, the dwellings presented in section 3.4.1 are used. Depending on each building material, the costs compared to the complete dwellings will differ and calculated back to costs per square meter. The LI (Losmaakbaarheidsindex) stays in this calculation at factor I, so the table presented in Appendix II.8 which shows the disassembly factors does not apply to the calculation, due to the fact that the reference projects at hand are traditionally constructed, without considering future disassembly methods. The advice concerning this calculation is to take disassembly factors in new dwellings into account, which makes the future perspective for developers more feasibility attractive because the demolition costs will decrease if the dwellings are designed to be deconstructed and not lose any value on the materials. Thus generating fewer costs for reusable materials. The BDB-index (bouwkostenindex), together with inflation and the discount rate of materials hasn't been taken into account within the calculation.

When the total deconstruction costs are determined, the deconstruction costs for a single material can be determined, based on the square meter BVO of the reference dwelling and the average deconstruction costs based on the Layers of Brand (1994). The results of the calculation of the costs are visible in Appendix II.IO for the six different materials, mentioned in section 3.4.2.

3.5.2.2 TRANSPORTATION

The second calculation is based on the transportation of the materials. Figure 3.39 shows the costs formula for taking materials to a processor, but the concept for the calculation can be adapted for transportation of building materials as well. For the calculation to be executed, a distance from the building spot where existing dwellings are located to the new building location should be determined, creating a certain distance (*A*). These exact distances are shown in section 3.6. Based on the price for transportation units per kilometre, the cost for a single ride of materials can be calculated (de Transporters, n.d.). Eventually the formula calculates a certain amount, which shows the price

per square meters $[\pounds/m^2]$ or per kilograms $[\pounds/kg]$. Appendix 11.10 shows the calculation method and the final prices for the six building materials per square meter.



3.5.2.3 MATERIAL VALIDATION

The materials in dwellings that are reused are probably old, damaged, or affected in some kind. This is where the material validation calculation comes into play to determine the influence these factors have and what it means for the costs to reuse these materials. The validation factor consists of three subjects, which are elaborated in this section and are also visible in Figure 3.40. The concept of the material calculation is established in collaboration with expert colleagues of Alba Concepts (Stijn van Enckevort, Sven Bögels, Jip van Grinsven) and after the discussion group on the topic of material validation costs, elaborated in section 3.5.2.

Material validation factor



Figure 3.40 Material validation factor reusable building materials (own ill.)

External influences and life span

The building elements that are part of the dwellings are affected by external factors over time. The first factor in this case is whether or not the material is located on the inside or the outside of the façade. Materials located on the outside could be more damaged in relation to materials on the inside due to weather factors or other external influences. The same goes for the inside. The exact factors of how much this influences the different materials is related to the location within the building. Next, most of the building materials could be influenced by pressure of other elements within the building, due to a vertical or horizontal connection or interface between the materials or elements. Both of

these external factors are influenced by time, so the age of the building should be considered as well. This is where the life span calculation of each material comes in. Every type of building materials has a different life span, depending on several factors, as already elaborated on in section 3.5.1 (NIBE, 2011). It is also necessary to have information on the construction year of the dwelling. If this is a known fact, the life span of materials could be used as reference to determine how much the material is influenced over time. Figure 3.41 shows a drawing of how the concept of time would influence the value due to the different factors (Van Enckevort, 2019). Starting at a certain value for a material, over time the different factors will influence the material value and so raising the costs for the materials if still to be used for new dwellings. If for example the material is outside, the material is influenced over time more significantly compared to if it should be inside. This creates a quicker drop of value. The same goes for the pressure factor.



Figure 3.41 Valuation materials over time (Van Enckevort, 2019)

LI (Losmaakbaarheidsindex)

The second factor is based on the Building Circularity Index (BCI), designed by van Vliet (2018). As mentioned in the literature study in section 2.5.2, the BCI consists of two components: the MI (Material index) and the LI (Losmaakbaarheidsindex). The LI is the index that is used for the calculation within this research, due to the two types of factors that are involved. The type of connection of the interface of the material and the accessibility to this connection. Based on the disassembly factors by Durmisevic (2006), the amount of influence of the factors are connected to the LI, which is visible in the two first table blocks of the table in Appendix II.8. The LI is not taken into account within the life span of the material. Looking at Figure 3.4I, at the moment of construction (t = 0), the material will drop directly to a certain point on the value scale. This downgrade happens directly because of the fact that once a building is constructed, the type of connection and accessibility of connection is determined and will not change, increase or decrease, over time, not taken any reconstructions or renovations of dwellings into account.

Appendix II.II shows an overview of the calculations performed on the six different building elements. This overview also gives insight in the factors used for the calculation, the type of materials,

the type of connection, the type of accessibility and the different influences by internal and external factors.

3.5.2.4 CONCLUSION

The different factors combined provide a formula to calculate the costs for the reusable building materials per square meters, which is visually represented in Figure 3.42. It consists of the sum of the deconstruction factor, the transportation factor and the material validation factors. An important notice is that this formula is based on the assumption and interpretation of different factors, which are coming from experts that are active in the field of circular construction industry, but the formulas haven't been applied to cases yet. The overview of the chosen factors are visible in Appendix II.II. The numbers that are used do not relate to a source yet, because it is difficult to substantiate these factors. Also, the creation of this formula puts different pieces from the world of defining the value of structural reusable materials together. Every single part of the calculation should be tested and substantiated before the formula can be applied in practice. When it is eventually being used, the test cases show which factors have the most influence in the outcome of the costs, also directly putting the reliability of this factor in perspective. So the results coming from these calculations should be analysed critically when developing similar formulas in the near future, for further innovation of the building sector. For now, the formula provides calculations for the operational tool. The execution of the model and the results are presented in the following sections. Section 5.3 looks back on these factors, discussing on the critical factors and which part has the most influence within the final outcome of the calculation.





Figure 3.42 Reusable materials total $[\pounds/m^2]$ (own ill.)

3.6 EXECUTION

The goal of this research is to reduce virgin building material usage within new dwelling projects. To reach this goal, the tool developed together with the input, coming from literature and real-life cases should be executed. To execute a case, the input should be complete and sufficient enough, based on the different levels of scale. This chapter describes briefly the demand necessary for the different levels of scale to operate and how the data is being used to test the qualities of the model. Eventually, the tool should provide significant results to substantiate the value of this research, which is elaborated in chapter 4 and discussed in chapter 5.

As mentioned in section 2.3, the tools input and eventual output could be divided into three levels of scale. Next sections show how this research interpreted the levels for the execution of the tool and what datasets are used for each level.

3.6.1 PROJECT-BASED MODEL

The smallest level of scale is the project-based model. This level of scale is used as the development level of the correct output numbers, such as the dwelling demand described in section 3.4 and the feasibility in section 3.5. The model's input consists of two hypothetical sources, due to the fact that the real-life project cases presented in section 3.3.2 did not provide significant data to test the qualities of the model. These two sources are located at a certain distance from the hypothetical construction site. The materials available at the location are coming from a series of reference dwellings. The construction year of the reference dwellings is also hypothetical. These hypothetical input factors are necessary to make the model run and create a realistic output. This output consists of a series of dwellings, based on the dwelling prototypes, but also a comparison of costs between if the dwellings would be built with only virgin building materials or if the dwellings would be built by mostly reused building materials. Other factors necessary for the input should be decided by potential clients, such as the budget willing to invest and the number of dwellings to be built of each type, minimum and maximum.

In Appendix II.II it shows the input page of the Excel model, with the hypothetical input. This is an example of a potential interface overview that could be used in practice. Section 6.I elaborates on the recommendations of the interface of the tool for practice.

3.6.2 REGION-BASED MODEL

The second model is based on the AMA-input, presented in section **3.3.3**. This region-based model is larger scaled in comparison to the project-based level, because of the bigger input. Instead of a few projects as data input, the region provides materials for the cases on multiple locations, with a larger budget. The development of the region-based model focused on experimenting with the possibilities of the 'What'sBest!' add-in for Microsoft Excel, to research the possibilities in scale, width and amount of variables and constraints. The data for this model is hypothetical as well, due to the fact the collaboration with EME, as earlier mentioned in section **3.3.3.1**, did not continue and why no actual datasets were provided by the EME-platform. This level of scale has the potential to be realistic and to provide insight and advice for larger projects by housing corporations or municipalities. The tool just needs real-life cases and datasets implemented to generate actual results. For now, the project-based tool provides the results necessary to substantiate the conclusion of this research, while the region-based tool provides recommendations and further innovation possibilities for the future.

3.6.3 ENTERPRISE-BASED MODEL

The model that is executed on an enterprise-based level is in need of a tool or program that has input on material flows of a scale much larger compared to the other levels of scale. The collaboration with the Cinderela project, elaborated in section **3.3.4.1**, promised to be an interesting input. However, the process of collaborating between an intern at a private company and the project organisation funded by the European Union turned out to be a more difficult process than expected. The organisation is willing to collaborate in the near future, meaning sharing access to the databases and in return providing the input of this research as additional material for the Cinderela project, if a clear Non-Disclosure Agreement (NDA) is signed. This document consists of the datasets, without sharing any
private information on correct locations of companies or other stakeholders involved. The sharing of this explicit data could be in contrary to the privacy regulations. If the Cinderela project is willing to accept the contractual agreements, the collaboration could take place and the datasets could be implemented in a future enterprise-based model. However, this model will be interesting to develop in further research, thus it is discussed in chapter 5.





PART 4 FINDINGS

In the former chapter, the entire setup of the main deliverable for this research has been presented, elaborated, substantiated and developed to a model that is being prepared to be used in practice. The model generated an outcome, based on different levels of scale, substantiated input and output. This chapter focusses on the main findings within these results, the financial numbers presented in the feasibility section, but also the scenario's to take into account when running this model and how they influence the results and possible future use of the model. Finally, the results, as presented in the research setup in section 3.I, give feedback on the main- and sub research questions in the conclusion of this research. This creates aspects on which the model could be improved, expanded or taken different scenarios into account in which the model could work differently. This is the connection to the recommendations overview in section 5.4. Other main findings are presented in the following sections. The results mostly focus on the main model developed on the project-based scale and partly results are elaborated on the hypothetical, yet still to be developed in future projects or research, region-based and enterprise-based models.

4.1 MAIN FINDINGS

Operational research is used to solve complex projects related to real-life cases. Through the usage of the operational model, a calculation took place that eventually shows the comparison in costs between the usage of reusable building materials and virgin building materials. During the collection of the input and output of the data to use for the operational model, collecting information showed the difficulty of collaborating with external stakeholders. The lessons learned and recommendations for future projects concerning such a large scale of sharing information is discussed in chapter 5. Actual real-life cases with reusable material are exchanged for hypothetical datasets, which generates the final output of the project-based model. These outcomes can be divided into the following subjects, based on section 3.2.4 and Figure 4.1:

- Reused building materials.
- Reference dwellings to be built.
- Financial comparison between reused and virgin building materials.

The complete overview of the output of the project-based model is visible in Appendix II.13.



Figure 4.1 Output specifications operational model (own ill.)

4.1.1 MATERIALS

The first output of the operational model is the actual overview of materials used for the built dwellings. How many materials are taken from which source and how many leftovers are there in different locations? For every material, a part of each source is used for the different dwellings to be built. How much has been used from a certain location depends also on the distance of the location. The bigger the distance, the higher the costs are from a certain point out. In the example presented in a project-based model, visible in the input page in Appendix 11.12, the distances are pretty close in range from each other. If the influence of transportation costs would be a bigger factor within the calculation, the division of materials over the eventual construction locations would probably be different. If the distance would be much bigger, the model would give an infeasible solution.

The materials used for this calculation are based on discussions with experts from Alba Concepts. This means that the materials in this overview show potential to be reused. However, to construct a complete dwelling, much more different materials are necessary. For now, only the possible reusable materials are taken into account in the calculation and discussed in section 3.4.2. To finalize the building, virgin materials are necessary as well, such as isolation, installations et cetera. This is further

discussed in section 5.4. Table 4.1 shows an overview of how many square meters of each material are used for each dwelling reference type. Also how many materials are used from each source is shown.

Output

Cutput							
ling type configurat	tion						
Dwallings	Dwelling type	A: Row-house	B: Semi-detached house	C: Detached hou			
Dweinings	Optimal	10	50	52			
nts of material used per type and dwelling type							
	Dwelling type	A: Row-house	B: Semi-detached house	C: Detached hou			
Floors	Channel plate floor BG	455	3.588	3.170			
	Channel plate floor 1e / 2e	676	3.729	5.574			
Exterior wall	Prefab concrete	1.110	7.399	4.043			
	Clickbrick	347	4.327	5.432			
Interior wall	Panels (Ytong/Faay/other)	528	3.030	4.445			
Roof	Roof tiles	645	3.795	4.561			
Windows	Glass + frame	245	850	2.143			
	Front door	10	50	52			
Doors [#]	Corridor door	70	350	464			
••	French door	0	100	0			

Amounts of Reused building material used per type and source							
	Used	Source O	Source P				
Floors	Channel plate floor BG	3.213	4.000				
FIDUIS	Channel plate floor 1e / 2e	3.980	6.000				
Exterior wall	Prefab concrete	9.500	3.052				
Exterior wait	Clickbrick	10.000	106				
Interior wall	Panels (Ytong/Faay/other)	8.000	3				
Roof	Roof tiles	5.000	4.000				
Windows	Glass + frame	3.238	0				
	Front door	112	0				
Doors [#]	Corridor door	884	0				
	French door	100	0				

Table 4.1 Output materials project-based model (own table)

4.1.2 DWELLINGS

Table 4.I also shows how many dwellings of each type could be built with the division of available reusable materials at the two sources. This number is calculated within the solution space and is the most optimal division. This is based on the minimum and maximum demand for each type of dwelling, stated by a hypothetical client. The dwellings are based on the reference prototypes, presented in section 3.4.I. The results show that it is more cost-efficient to build the second and the third prototype, based on the calculations of the costs for the materials, which is discussed in the next section. It also shows the potential of what amount of dwellings are taken into account, or located at different construction sites, the model has to be extended, creating a set of multiple variables that has to be implemented. An important notion is similar to the previous section: The dwellings cannot be built with just the materials at the different sources. Multiple virgin materials should be added to completely construct the dwellings. These are not taken into account in the feasibility overview, due to the fact that the goal of the model is to give insight in material costs for the potentially reusable building materials.

4.1.3 FEASIBILITY

The third main findings section relates to the financial side of the model. As mentioned in the output overview of the operational model, a comparison between the costs of reusable and virgin building materials are calculated. In chapter 5 the discussion among the results is elaborated, based on the choices made within the research and the results stated within this section.



Figure 4.2 Output amount of dwellings (own ill.)

Of the three theoretical levels of scale, the project-based model has been executed within this research. The input of this model has been described in earlier sections. The different materials that are taken into the comparison are calculated, based on formulas and assumptions made in the feasibility section 3.5. For every material, the model calculates the total budget of the type of material for all the reference dwellings to be built. This generates an overall costs price for every type of material, potentially reused from a source compared to the costs if the materials where virgin. The comparison between each of these materials shows a difference in costs, presented as a percentage of how much more or less it would cost to use reusable materials. These numbers are summed up, to finally result in a total overview of the costs for the project.

Table 4.2 shows an overview of the financial comparison of the project-based model. The first overview shows the difference in costs for the dwellings to be built in total with reusable building materials (light blue) compared to dwellings built with virgin building materials (grey). The second overview shows the difference in percentages of the costs for the dwellings and each specific building element. The size of the circle shows the size of the difference, green when it is cheaper to use the reusable building materials and in red shown when virgin materials are cheaper. Appendix 11.13 shows the entire output of the operational model.



Table 4.2 Project-based model output (own table)

According to the input visible in Appendix 11.12, it shows that building dwellings with reusable materials has margin of 14% in total. This would mean, based on the number of reusable materials available, the three dwelling types as reference, the distance of the sources of the materials, the budget and the demand of dwellings by the client, it would be 14% cheaper to use reusable building materials compared to virgin building materials on a project-based scope. It is noticeable that three of the materials, which are the channel plate floors and the prefab concrete walls are more expensive when reused, while the rest of the materials are cheaper when reused in this comparison. Especially

the floors, which are made out of heavier material, are much more expensive compared to virgin floors.

The results of the comparison depends on a lot of factors. Different assumptions are made to achieve these results and the framework used for the operational creates a large number of interesting topics for discussion. However, due to the framework which is used, the results provide a slightly misdirected vision on the possibilities on reusable building materials in new construction projects. The goal of this research is to reduce the usage of virgin materials and the findings of this model give insight in what numbers could be calculated in the given case. But the input of the case which is calculated and which delivered the numbers and costs that are presented in this chapter, are still hypothetical. The model could actually provide an influence in the construction industry, when the important aspects of this model are clearly evaluated and adopted into different scales. The model will definitely provide help in the transition to a more circular construction industry, when the framework is being increased in scale and context. The boundaries of the framework brought to this research and how the different aspects of the model responded to the boundaries are elaborated in chapter 5. Also, many follow up research ideas, such as recommendations for the different elements of the tool to be developed, other market parties or interesting strategies are discussed in chapter 7.



PART 5 DISCUSSION

As mentioned at the beginning of the research, operational research tries to solve complex real-life cases. In this research, the main focus is to discover how the usage of virgin building materials in new construction projects could be decreased. The results presented in the former chapter showed a difference in costs between reusable and virgin materials, the reusable materials in overall being cheaper. But, the choices made which resulted in this number is based on the chosen framework this research operates in. The following section discusses these factors, what influenced the decisions made and why some of the assumptions where made. Also, this chapter focusses on further development possibilities for the operational model, based on factors outside of the scope of the research. Finally, the process of the research is evaluated, eventually leading towards a final conclusion of this graduation thesis.



Transparancy of datasets



Reusability of building materials



Validation of feasibility



Involvement of stakeholders

Figure 5.1 Discussion topics (own ill.)

The results of the project-based level calculation show that it is, in fact, cheaper to use reusable materials for new dwellings to develop. But, there are many factors that should be taken into consideration when stating the fact that reusable building materials are cheaper than virgin materials. If you would discuss this result with experts or companies in practice, most of the people would disagree or do not believe the validation of the calculation. This is partly because the construction industry is far more complex than the costs for just the building materials. To transform the entire structure of the construction industry, far more extensive research should take place and even when it is researched, due to the large scale and the many stakeholders within the construction industry, it would take years, even decades to transform the system. However, for this research, certain assumptions made within the scope of the reuse of building materials are discussed. The following sections provide four main topics of discussion, each related to the scope of the research

and substantiated with arguments to take into account when stating the results mentioned in the former chapter.

5.1 TRANSPARENCY OF DATASETS

The first topic for discussion is a problem which makes the entire transition from the traditional concept of dwelling projects to reuse of building materials in a new projects much more difficult. There is a lack of transparency in the availability of reusable building materials. Even though there is so much material out there, ready to be reused, there is no clear overview of what the exact numbers, dimensions, quality and other specifications are. Another problem is the lacking of connection between demand and supply. There is a lot of material available, but also a lot of (re)-construction projects initiating. The connection should be made between the companies starting new projects and cases where materials are ready to be reused. This matchmaking should be taken place a lot earlier in new projects.

However, the problem is the competition between the different suppliers, which creates a barrier between demand and supply. The irony is that multiple organisations who are leaders in this transition are all about collaborating, sharing, transparency and other motivational speeches, but in reality, the construction industry is far from an integrated collaborating structure. During this research, the search for datasets from real-life cases turned out to be a real struggle. As mentioned in the description of the different possible input cases, section 3.3, there were many interesting partners to collaborate with, such as the EME-Platform, the Cinderela project or the Amsterdam Metropolitan Area. However, due to the fact that this research is part of the Technical University of Delft and consultant company Alba Concepts, partners hesitated and finally declined to share information as input. Most of this information contained privacy-related data, such as addresses and personalized information of participating companies.

For integrated models such as the operational model developed in this research to create actual insight into the feasibility of the project, there should be a change in sharing information and transparency of datasets. If so, there is much more information available that could be implemented in the model. This would create possibilities to develop models on the different levels of scale, which then could be compared to each other, to have a much larger overview of results, based on a much larger set of variables. The more variables, the more substantiated the results will be. When this model is used as an advisory tool to create insight for an organisation such as a municipality, the possibilities become much larger, meaning different types of dwellings could be built and much more materials will be reused. For now, the model is limited to hypothetical datasets, but if competition would be replaced by collaboration, it would be possible in the nearby future to produce real-life results.

5.2 REUSABILITY OF BUILDING MATERIALS

The second topic of discussion is related to the actual materials that will be reused. The six types of building elements, described in section 3.4.2, are divided into a set of materials which show potential to be reused or related to reference projects that showed whether or not the material could be reused. This list of materials is determined in collaboration with an expert colleague of Alba Concepts. However, the number of materials are chosen within a certain scope, while in fact much more of the

building elements of dwellings should be reused. In this research, the final calculation shows the comparison between the costs of the reusable and virgin building materials, which can be reused. However, for an entire dwelling to be realized, much more materials are necessary. Looking at the layers of Brand (1994), the other layers, such as 'services', 'space plan' and 'stuff' are not taken into account in this calculation. If more information and techniques are used to deconstruct and preserve the other building elements, the list of to be used materials will be larger, thus creating more dwellings with almost completely reused materials. Looking at the materials of the dwellings that showed the potential for reuse for now, it covers a large area of the total square meters of material within the dwellings. The question could be asked how much of a difference it would make if the other materials, which are left out in the calculation due to the fact they do not have a high potential for reusability, are investigated to be reused as well. Would it be feasible?

The design process for the future of dwellings should be taken into account as well. The dwellings that provide the building materials coming from real-life cases are constructed in earlier periods. Every different construction period used different techniques, different materials within the buildings and especially different interfaces and connections. This makes it difficult to determine how much of the materials will be preserved when deconstructed. When reusability of materials will be the standard for new dwellings, the costs for deconstructing will decrease, making it much more interesting to reuse the materials of future dwellings. This instantly creates a larger list of materials to compare with the virgin costs. However, the techniques of detachable construction and circular dwellings is a must to improve the reusability of building materials and are further elaborated in the recommendations chapter.

5.3 VALIDATION OF FEASIBILITY

The main question of this research focusses on reducing the usage of virgin building materials in the future, which would results into a more circular construction industry. The feasibility of this transition plays an important role. The validation of the feasibility is thus an important topic for discussion. The results show a comparison between the costs for reused and virgin building materials, showing the difference in the costs for each type of material and also the costs per dwelling type. This quantity is based on the material costs, while in construction costs much more domains are a factor in the final budget estimation. Figure 5.2 shows which other costs could be taken into account as well. Indirect construction costs and additional costs are not directly influenced by the use of reused or virgin materials. However, processes that are traditionally being used for many years could change when reused materials are joining the process. Labour hours could increase because constructors are not used to the technique of construction these type of materials. Designers should adapt used materials within their dwellings, which would increase the duration of the design process, compares to standardized dwellings. The entire system of collaborating with materials will raise additional costs, such as operating costs, start-up costs, administrative costs et cetera. To calculate an entire project of creating dwellings with reused building materials, much more aspects of costs should be taken into account to create a complete, well-substantiated comparison of costs and the entire budget.



Figure 5.2 Investment costs of a construction project (Gemeente Amsterdam et al., 2017)

The costs for each material type are also up for discussion. The virgin material costs overview, which is visible in Appendix II.9, is based on sources online. For example Cobouw Bouwkosten (Cobouw, n.d.) has been referred to many times. However, the source of costs for materials is different for each material. Most of the materials showed a certain amount of euro's per square meters, which description consists of a so-called recipe, showing an overview of all the factors and additional materials and processes which finally lead up to the costs per square meter. However, not all the materials are this well-defined, making it difficult to determine the actual material costs per square meter. Some recipes of materials describe besides the material costs also assembly and labour costs. To create a well-substantiated overview, a collaborating company, experts in construction costs, should provide an overview based on a singular database, where all the information of the different material are similar and based on the same sources.

The formula designed for reusable materials is probably the most interesting point of discussion. This calculation is not yet being used in practice and is an interpretation of different investigations of commodity valuations and factors of detachability and deconstruction. During the process of generating the formulas, through expert meetings and discussion with colleagues from Alba Concepts, the numbers where created. To actually validate the formulas, real-life projects with correct input are necessary to see if it would generate a realistic outcome. When tested and combined with a calculation of the different additional costs presented in Figure 5.2 it would show if it actually is interesting to use reusable materials in new dwellings. The formulas have been developed by focusing on a specific framework of factors, presented in section 3.5.2, and should be tested by example cases. These factors each have a different influence on the final calculation. Where the transportation factor is of a much lower influence compared to the external factors. The impact is much bigger, when looking at the increase in costs if materials has been damaged due to whether or pressure influences, or when the lifespan is much shorter. Also, when a material is hard to reach, the accessibility factor of the material is much lower, meaning it will be compensated in costs. The different influences these factors have are important for the realization of projects with reusable materials. It gives an overview for stakeholders which materials are interesting to reuse and which are not. When these formulas are being used in practice, it could affect the way we construct new

dwellings and deconstruct old dwellings. The formula focusses in the moment on creating insight and making a comparison, showing the potential of the model when eventually being used in reallife. When applied in practice, the factors will develop and it will show how much influence the different aspects of the formula actually do have.

Not only the financial formulas of the costs for the building materials should be validated, the operational model should be validated as well. When design a model focussing on a certain context, it should be constantly tested and updated based on results from each test. As mentioned in section 3.1.3, in research through design it is important to always analyse the results and give feedback on the design steps taken in the development. But it is also of importance to analyse and develop the variables and constraints of the model. These factors react different when the model is setup in different ways, or when the input of the model is changing. It could much clearly show which variables and which parameters are of most influence in the results, which eventually results in a model which is tested in multiple setups. This validates the different aspects which are taken into account in the model. In the follow up of the process, the same numbers, both financial formulas and specifications of the chosen materials to reuse, should be tested in similar setups of the model, but based on the different levels of scale. When this has been checked, the results of the model are much more trustworthy. This could definitely increase the potential of implementing the tool into the practice of the construction industry. But if the tool is being further developed, validating the model with multiple steps is a necessary given.

5.4 INVOLVEMENT STAKEHOLDERS

Projects such as dwellings constructed with reusable structural materials to support the transition towards a circular building economy will never take off unless there is a significant amount of stakeholders involved within the process. However, it is yet to define in what phase of the project which stakeholders will be involved and how they are involved within which part of the calculation of the model. If the initiative comes from a municipal department, partners in the possession of datasets of building materials are necessary, but also companies who are willing to take on construction projects with reusable materials. To obtain the materials, there is a discussion of ownership of the materials. Also, the materials to be reused are probably still within their current dwelling, meaning a stakeholder should be involved to deconstruct the dwellings and transport the materials. Overall, these type of projects require multiple stakeholders, especially in the initiative phase, to get the project initiated. The model will be used as an advice tool, but when the possibilities of the model are projected on the demands of the client, the insight the model gives could motivate the client to get related stakeholders with sustainable ambitions involved. On the other hand, there should be a responsibility coming from the owners of buildings to be aware of the materials that are in their possession. Most of the larger companies who possess real estate or dwellings are not aware of their own setup of building elements within their buildings. There are external stakeholders who could make inventories of complete building stocks, but to really improve the circular economy, every owner of buildings should be up to date and aware of the available building materials. When this happens, transaction of buildings and collaborated projects will initiate much easier, because the datasets are up to date and easily shared between stakeholders. There will be a bigger responsibility for the owners, but it will result in much more collaboration and thus circularity of building materials.



PART 6 CONCLUSION

This chapter focusses on the conclusion of the research, by answering to the main and sub research questions, which are stated in section I.2.3 and are described within the conceptual model. This conclusion focusses on the entire research, supported by the discussion and recommendations for further research, stated in the former chapter. The conclusion is described by answering each sub research question separately, concluding whether or not the research goal is achieved by answering the main research question.

6.1 SUB-QUESTION 1 FEEDBACK

The first sub-question of this research consists of the following:

What specifications are necessary for a case study to be implemented as input for the operational model?

During the research the three levels of scale were presented that divided the possible real-life cases into three different sizes of context. For every level of scale, a model setup was defined, which provided a dataset with a certain amount of specifications on the different materials implemented. However, to obtain the datasets of the case studies from the different partners, a large process was necessary to go through, which was based on a future collaboration. For the project-based model, the different case studies provided by Alba Concepts lacked the correct information and the collaboration of 'de Alliantie' never initiated due to financial problems. The possible collaboration on a region-based level with the 'EME-platform' had the most potential, but due to fear of sharing data because of the competition between similar material exchange platforms the data was never shared, which led to a dead-end, eventually using the hypothetical flow of information of the 'EMEplatform' as a setup for the region-based model. The third input level, on an enterprise-based level, provided collaboration with the most potential, which was the 'Cinderela' project. This overview of material flows in the entire country could be the best input this operational model could be developed to, but due to the long-overdue process of contractual agreements between both parties the collaboration didn't get initiated. Even though, an enterprise-based model with input from the 'Cinderela' project has the most potential for future research to be developed, due to the completeness and accurateness of the system. So the third datasets could be the most optimal, but due to the difficult and underestimated process of obtaining the data, hypothetical datasets are used for the input of the operational model. When in the future datasets and case studies will be used as implementation for these typical kind of calculation models, it is important there is a clear definition of collaboration and data sharing between the stakeholders, internal and external.

6.2 SUB-QUESTION 2 FEEDBACK

The second sub-question of this research consists of the following:

What is the current demand in the construction industry for new housing projects?

There are many different dwellings that are standardized by the government. For this research and the operational model, three prototypes are analysed and presented in section 3.4.1. However, as mentioned, there are many more standardized dwellings that could be used as a demand for these type of projects (W/E Adviseurs, 2011). These could be implemented in the model as well. The demand depends on each different project on the context and the budget of the client. In general, as presented in section 2.6, multiple regions, especially in the Amsterdam Metropolitan Area, are in need of new dwellings. However, the main challenges lay within the design of the future standardized dwellings. As mentioned by Jim Teunizen, founding partner of Alba Concepts, circular construction projects with reusable materials integrated into new dwellings will be normal, way sooner than expected. So the demand for dwellings in the future is to adapt to the idea of reusable dwellings, which could be detached easily to lower the costs for reusable materials in the future, improving the life span of these materials and the reduction of waste and virgin building materials. The amount of materials which

are analysed and taken into account in the operational model is relatively small, compared to the many materials which are available in a standardized dwelling. The different materials which haven't been taken into account in the calculation will be part of the future demand for new dwelling projects, as well as the techniques of how to implement the reusability of these materials.

6.3 SUB-QUESTION 3 FEEDBACK

The third sub-question of this research consists of the following:

How can a comparison between the overview of the cost of virgin and existing building elements for new dwellings provide a feasible business opportunity?

This question focusses on the feasibility of the project. The value of the comparison between virgin and existing building materials lays within insight which it gives to the client. The overview consists of the comparison of each different building materials, whether it is cheaper or more expensive. Clients can adapt their wishes to the results of the calculation. However, it also gives an insight on how to calculate the costs of reusable materials. The formula designed is innovative, based on recently developed calculations and validation factors. This formula should be tested in context, to get it validated. However, for now it shows the costs of the materials. As mentioned in the discussion and recommendations chapter, there are many more aspects of feasibility to take into account when calculating the costs of an entire dwellings project, which consists of reusable materials.

The second factor to take into account is the adaption of real-life context as input. When a client enters input in the interface of the operational model, certain variables are necessary, such as location, budget and demand for dwellings. However, also the supply of materials is an input of the client. The feasibility of the business case is being calculated and if the result is negative, different strategies are possible to build towards a feasibly business case, by for example only reuse building materials that are cheaper if reused or to use a different source. The quality of the comparison lays within the fact that it gives insight on which materials the project could benefit and how the model should be adjusted to realize the business case. The creation of the interface of the tool plays an important role when this model adapted into practice.

6.4 MAIN RESEARCH QUESTION FEEDBACK

The main research question of this research is:

How can an operational model link the supply of existing building materials with the demand for new construction projects in order to reduce the use of virgin materials and thereby improve circularity in the construction industry?

The main conclusion:

By developing an operational model which calculates the **difference in costs** between the reused and virgin building materials, showing that the reusable materials are **cheaper** compared to the virgin materials, based on the given **framework**.

This question generated the research goal of reducing the usage of virgin building materials in starting construction projects. Looking at the results of the project-based level operational model, it shows in the comparison in feasibility that the costs for reusable materials are cheaper compared to the virgin building materials. This resulted in an overview and insight for the hypothetical client of the project, which shows it is more interesting to invest in reusable materials compared to virgin materials. This could lead to a reduction in the usage of virgin building materials, which was the goal of the main research. Also, the final deliverable, which is the operational model, proves its functionality and showed insight within the solution space. It is developed with the possibility to implement different datasets, multiple variables and other factors. The qualities that the operational model presents show the value of the model for the construction industry.

With presenting this conclusion, a set of comments should be made to substantiate the statement listed above and the quality's that are related to the operational model. In the discussion chapter, the framework in which the results were executed is based on a set of assumptions that lead to a positive outcome of the comparison. This means that every context leads to a different result, so the numbers presented are purely to provide insight for companies or municipalities who would consider reusing building materials. To execute the project, much more research and additional variables are necessary to create a closed framework with a realistic outcome. These recommendations were presented in section 6.I. The supply should be realistic and based on real-life cases. The feasibility study should consider the costs that are not taken into account within this research, based on the investments costs overview of Gemeente Amsterdam et al. (2017).



PART 7 RECOMMENDATIONS

The construction industry is one of the most traditional sectors in the Netherlands. There is still a lot of space for development, innovation and transition towards a circular industry. The following section describes the recommendations for further development and research, based on this graduation thesis.

Even though the project-based operational model is being developed as a conceptual model for potential business cases, it has the potential to grow significantly. The number of factors that could be added within the formula for the linear programmed calculation is unlimited, so is the amount of solutions of dwelling types to create. Also, the input for different business cases could differ from small to extremely large. If the companies start using business models for similar projects, it could initiate a quick development of more extensive tools, interfaces, platforms and such. If there is a clear collaboration between IT companies, data suppliers and partners from the construction industry, these type of models could improve the construction industry concerning circularity at a rapid pace. An important notion to make is the validation of the operational model. It has to be validated as much as possible to test whether or not the model will work in reality. By downgrading the model to simple variables and focussing on checking each variable on its own, these type of operational models will become much more trustworthy and sooner applicable in real life projects and cases. When the system is checked on all the different constraints and variables in different contexts. It could have much more influence in the construction industry, compared to the development of the model in this research.

On the other hand, the system used for the operational model to run has its limitations. The 'What'Bestl' add-in of Microsoft Excel is limited to its amount of variables to be used. To create the possibility to add more data in one immense model, the tool has to be updated or other systems have to be developed, which are easily accessible for construction companies and municipalities. Otherwise, traditional companies are easily returning to traditional building methods and virgin building materials. Also, the information of secondary raw materials could easily never be shared, due to fear of losing advantage in competition, but also due to privacy regulations. If the transparency of knowledge on available materials is not achieved, the possibilities for a typical kind of system to grow and connect the company's nationwide is probably a myth. So the technique behind the model should be updated.

The input for the model has a lot of potential as well. The collaboration with Cinderela could create an enterprise-based model with datasets from the entire country, which has a much larger scale, directly having a much bigger impact within the transition towards the CE. This directly results in advice for the government, reacting on the challenge of I million homes in the Netherlands and directly showing the potential of reusing a large number of materials within the country. The involvement and transparency of stakeholders are necessary to achieve such a large project.

On a smaller scale, there are also many possibilities to develop. As mentioned in the discussion, the dwellings should be designed to deconstruct in the future, creating much more supply. However, other markets to use the materials for are interesting as well, for example renovation. On a small scale, structural materials of historical quality could be preserved by reusing them in nearby, similar projects. This is a different market compared to new dwellings, but if there is enough demand this factor could be taken into account in a similar model as well.

The calculation in the project-based level example results in a positive outcome. However, if the numbers would be negative, different strategies could be taken for a municipality to decrease the costs of the reused materials. Every material shows if it is cheaper or more expensive. Clients can anticipate on the distinction of costs, by redesigning the dwellings. Development of the model is to create an insight that directly gives feedback on costs estimation of the different materials to be used. In this case, when a project is initiated, the client can adjust the different materials used and the different dwellings to build to a feasible solution. This interface could be designed or further developed in collaboration with a website or app designer, but based on the system created within this research. This tool could provide insight for clients such as a municipality or a housing client. The solve page presented in Appendix 11.12 would be the setup. The input would exist of the number of dwellings the clients want, the budget which is available and the preference for type of dwellings on locations.

The main message which is delivered by this research is the great potential of improvement of the construction industry. The potential is clearly visible in the different aspects of the construction processes, but most of the stakeholders involved are holding on to traditional processes. Due to fear of risks, losing money or uncertainty, most of the stakeholders aren't willing to implement circularity within the processes. This research shows there is definitely a lot possible to work with. The 14% shows the potential, reusable materials purely are cheaper when being used for new dwellings. But there are so many other aspects involved when developing building, which means much more research is necessary. This research could act as a pioneer, focused on the first step of the MAMO costs and encouraging other students or employees in the current area of construction to start researching the following aspects and implementing them as soon as possible into the traditional processes. The sooner the transition of rearranging these processes initiates, the sooner the construction industry will benefit from all the innovation which are about to be developed. Hopefully, the results of this research will trigger the construction industry to take research of reusable building materials and implanting them in the standardized construction processes to the next level.





PART 8 EVALUATION

The graduation process started a bit different compared to the regular first weeks of a graduation project. Instead of looking for a subject between the different departments, I've looked for a project that I could start with before the semester started. This provided some difficulties but also gave me a head start. The Joint Interdisciplinary Project at Royal HaskoningDHV directly gave me a subject to focus on. The project was going on for 10 weeks, full-time. Normally it is not possible because other courses should be taken during this period. However, it was possible to combine the courses with the internship. For future students who would pursue to do something similar, it is wise to research possibilities with the credits and courses still to take. Otherwise, it is not efficient to start your research graduation with such a large project. In my case, I discussed it with Professor and chairman of the Management in the Built Environment department Vincent Gruis and Master Coordinator of Management in the Built Environment Fred Hobma, which both agreed to the situation.

The second part of the graduation course was more difficult because the project for the first quarter ended. Instead of a clear objective, which Royal HaskoningDHV provided, there was no goal anymore and it felt as if the research stagnated. It was difficult to continue on the subject because there was no case it was connected to anymore. So in this period, it was key to find an internship where the rest of the graduation could be executed. From that point, it would generate more motivation to continue.

After the P2 presentation, I started the graduation internship at Alba Concepts at 's Hertogenbosch. Associate professor Hilde Remoy put me in contact with one of the partners and after two meetings we decided I could join the company to execute my master thesis. From the start of the internship, the idea was developed that my thesis would collaborate with an organisation in Amsterdam, the Excess Material Exchange Platform. During the first few months, the process of setting up the collaboration between the EME turned out to be a struggle. Both parties couldn't agree on a collaboration, not willing to share data with a graduate student which was connected to a University, fearing the data would leak. It showed the sensitivity of the practice on innovation, but also the underestimated vision I had on obtaining these datasets. There was too much risk apparently to share data, which eventually led to a shutdown of the collaboration. This turned out to be a setback in the progress of the research and the development of the model. After a final determination of the collaboration, other projects were becoming an option for the use of the model. One of those options

was a possible collaboration with de Alliantie. This housing corporation would start a project where archetypes within their portfolio would be analysed, which would create perfect datasets to use as input for my model. Unfortunately, the project never started due to financial problems at de Alliantie. Finally, in consultation with my graduation councilors, we decided to focus on the output and the calculation of the model and use hypothetical datasets, but with the notion that datasets are necessary for the validation of future projects.

The main lesson learned from this experience is the fact that innovative projects within large sectors such as the construction industry take much more effort to accomplish the correct components. Large projects never go exactly as planned from the beginning, which means that during the graduation process events will happen which push you of the track and you need to adapt to the changes. Working with multiple partners makes you dependent on third parties, which means it will not always go as you expect. The setback I've had due to the fact datasets were not shared with me meant I had to change my research setup and perspective towards a final achievement. It is important to realize during the process what value can be taken out of the data you actually can attain and how you use this data.

The graduation company provided me with a lot of experience on this subject from practice and provided me with multiple datasets, such as example projects, expert panels and multiple discussion with colleagues, internal and external. The value of a company that supervises you during a graduation process is a must-have for a project that directly connects with the demand in the practice. The choice to be part of Alba Concepts took this research to the next level.



PART 9 REFLECTION

Within the field of operational research, the main objective is to solve real-life problems. Within the transition towards a circular construction industry, the main objective in this research is to reduce the usage of virgin building materials. However, the challenge to complete the transition and make the entire industry of construction circular is much and much bigger. Though there is many literature already produced on these topics. The final chapter of this graduation thesis reflects the current literature and how this research adapts to the existing publications.

The literature that describes the operational research methods, published by Barendse et al. (2012) which has been used for most of the references and the development of the operational model, is a substantiated report of the technique of operational research, how it can be applied to design solutions and to solve problems. The course, which is lectured at the second year of the Management in the Built Environment master at the Faculty of Architecture and the Built Environment, has the mission to teach methodological concepts, research methods and problem solving methodologies that can be applied by students for their master thesis (Barendse et al., 2012). The reader, which the students use during the course, lacks of example graduation projects to explain schematically how the model is interpreted. This graduation thesis could supply an example project on how the model is developed, which variables and constraints have been used and how the formulas and systems are being linked to the 'What'sBest!' add-in for Microsoft Excel. This technique also showed its limits, due to the maximal amount of variables that could be used when the trial version of the add-in is used. This thesis provides an example of the possibilities within operational research and could give students more insights in how to execute an operational related graduation thesis.

In the development of the calculation of the costs for the reusable building elements, literature has substantiated the factors that influenced the final outcome of the total costs per material. The LI (losmaakbaarheid) factors are based upon the development of the BCI (Verberne, 2016), which is developed by Alba Concepts. Earlier within this research, it has been mentioned that an important innovation within the building sector is creating a substantiated universal language of the terms of the different layers of materials, elements and products. Within practice, as mentioned by multiple colleagues of Alba Concepts, the discussion on the different levels of definition is still ongoing. There is still no universal language that states when a material is a material, an element is an elements and a product is a product. Within this research, the building materials and elements are referred to a lot

and used consistently within naming of definitions. A material consists of a certain recourse, while building elements consists of set of materials. This covers almost all the different building elements, meaning that for some building elements the discussion continues whether or not it is an element or a material. This research reflects on the use of these terms and provides a structure that could be applied, but it depends on future research if it eventually will.

The third topic reflected on the literature is the matter of factors that influence the formula of reusable building materials. The literature consist of mostly existing formulas developed by Madaster et al. (2018), but they haven't been tested in real-life projects yet. They are still under construction. The transportation and deconstruction calculation has been implemented in this research, with some minor adjustments. This research gives an overview of the outcome of these calculations and can provide feedback on the validation of the formulas. Of course the data input within this research is hypothetical as well, but the formula has been added to the complete overview of costs for reusable materials. When the other pieces of the puzzle are reflected as well, Madaster et al. (2018) could use the results of the operational model developed for future development of determining the value of reusable materials. The external influences calculation could be added within the development of Madaster et al. (2018) as well, but first the model has to be tested with real-life cases to validate the formulas developed. When connected, this research could bring a lot of pieces of the puzzle in adding value to reusable materials in new construction projects together, thus supporting the transition towards a circular construction industry.



PART 10 BIBLIOGRAPHY

- Ackoff, R. L., & Sasieni, M. W. (1968). *Fundamentals of Operations Research*. London: John Wiley & Sons,.
- Agamuthu, P. (2008). Challenges in sustainable management of construction and demolition waste. *Waste Management & Research*, 26(6), 491-492. doi:10.1177/0734242X08100096
- Agarwal, R., Chandrasekaran, S., & Sridhar, M. (2016, 06 2016). Imagining construction's digital future. Retrieved from <u>https://www.mckinsey.com/industries/capital-</u>projects-and-infrastructure/our-insights/imagining-constructions-digital-future
- Akadiri, P. O., Chinyio, E. A., & Olomolaiye, P. O. (2012). Design of A Sustainable Building: A Conceptual Framework for Implementing Sustainability in the Building Sector. *Buildings*, 2.
- Alba Concepts. (2018). Building Circularity Index.
- ANP. (2019, 05-01-2019). Woningnood gaat nog tientallen jaren duren, helft van 1 miljoen benodigde huizen op tijd af in 2030. Retrieved from <u>https://tpo.nl/2019/01/05/woningnood-gaat-nog-tientallen-jaren-duren-helft-van-1-</u> miljoen-benodigde-huizen-op-tijd-af-in-2030/
- Arnold, T. J. R., Chapman, S. N., & Clive, L. M. (2008). *Introduction to Materials Management* (Sixth edition ed.). Upper Saddle River, New Jersey, Columbus, Ohio: Pearson Prentice Hall.
- Arora, R., Paterok, K., Banerjee, A., & Saluja, M. S. (2017). Potential and relevance of urban mining in the context of sustainable cities. *IIMB Management Review*, 29(3), 210-224. doi:10.1016/j.iimb.2017.06.001
- Barendse, P., Binnekamp, R., de Graaf, R. P., van Gunsteren, L. A., & van Loon, P. P. (2012). Operations Research Methods for managerial multi-actor design and decision analysis. Amsterdam: IOS Press.
- Binnekamp, R. (2018). Structuring the inter-actor design problem: the virgin materials reduction case. Management in the Built Environment. Delft University of Technology. Delft.
- Binnekamp, R., van Gunsteren, L. A., & van Loon, P.-P. (2006). *Open Design, a Stakeholderoriented Approach in Architecture, Urban Planning, and Project Management* (Vol. 1). Delft: Eburon Press.
- Black, K. (2018, 29-03-2018). Urban Mining and Circular Construction what, why and how it works. Retrieved from <u>https://www.metabolic.nl/news/urban-mining-circular-construction/</u>

- Bo-Ex. (n.d.). Bo-Ex Innovatieve Netwerkorganisatie. Retrieved from <u>https://www.boex.nl/</u>
- Boer, E. (2018, 20 december 2018). Memo Selectie- en gunningsfase project DOEN! Rosmalen.

Bögels, S. (2019). Sven Bögels. Retrieved from https://albaconcepts.nl/sven-bogels/

- Brand, S. (1994). *How buildings learn : what happens after they're built.* New York, NY: Viking.
- Brey, M., & Hansen, K. (n.d.). *The Circular Economy Powered by Cradle to Cradle*[®]. Hamburg. Retrieved from <u>https://epea-hamburg.com/wp-content/uploads/2017/10/The-</u> <u>Circular-Economy-powered-by-Cradle-to-Cradle.pdf</u>
- Bryman, A. (1989). *Research Methods and Organization Studies*. New York: Routledge, Taylor & Francis Group.
- Bryman, A. (2012). *Social Research Methods* (Fourth ed.). New York: Oxford University Press Inc.
- Büch, R., Meiberg, E., Wuestman, E., de Werk, G., Heideveld, A., & van Leeuwen, K. (n.d.). Cirkelstad, geen afval geen uitval. Retrieved from <u>https://www.cirkelstad.nl/</u>
- Centraal Bureau voor de Statistiek. (2016). *Bevolkingsgroei per gemeente*, 2015-2030. Retrieved from: <u>https://www.cbs.nl/nl-nl/nieuws/2016/37/pbl-cbs-prognose-groei-steden-zet-door</u>
- Cheng, K. L., Hsu, S. C., Li, W. M., & Ma, H. W. (2018). Quantifying potential anthropogenic resources of buildings through hot spot analysis. *Resources, Conservation and Recycling*, 133, 10-20. doi:10.1016/j.resconrec.2018.02.003
- Cheshire, D. (2016). *Building Revolutions: Applying the Circular Economy to the Built Environment.* . Newcastle: Riba Publishing.
- Cobouw. (n.d.). Cobouw Bouwkosten. Retrieved from http://www.bouwkosten.nl/
- Cossu, R., Salieri, V., & Bisinella, V. (2012). Urban mining: A global cycle approach to resource recovery from solid waste. *CISA*.
- Cossu, R., & Williams, I. D. (2015). Urban mining: Concepts, terminology, challenges. *Waste Management*, *45*, 1-3. doi:<u>https://doi.org/10.1016/j.wasman.2015.09.040</u>
- Crosby, M., Nachiappan, Pattanayak, P., Verma, S., & Kalyanaraman, V. (2016). BlockChain Technology: Beyond Bitcoin *AIR: Applied Innovation Review*(2).
- Damen, M. A., & van Maaren, C. (2018). Excess Materials Exchange *Turns your waste into wealth*. Amsterdam.
- Davy, N., & Guar, N. (2018). The SDGs: Materially impacting the bottom line, 2018 Annual Review of Progress on the Global Goals. Retrieved from London: <u>https://corporatecitizenship.com/wp-content/uploads/The-SDGs-Materially-Impacting-the-Bottom-Line-FINAL.pdf</u>
- de Alliantie. (n.d.). Hart voor wonen, altijd dichtbij. Retrieved from <u>https://www.de-alliantie.nl/</u>
- de Transporters. (n.d.). Onze tarieven, categorie 7 vrachtwagen. Retrieved from https://detransporters.nl/tarieven/
- Durmisevic, E. (2006). *Transformable building structures : design for dissassembly as a way to introduce sustainable engineering to building design & construction*. Delft University of Technology, Delft. Retrieved from

https://www.researchgate.net/publication/27348053_Transformable_building_struc tures_design_for_dissassembly_as_a_way_to_introduce_sustainable_engineering_to _building_design_construction

- Duurzaam Gebouwd. (2018, 12-04-2019). Circulaire horecapaviljoen Green House in Utrecht geopend. Retrieved from <u>https://www.duurzaamgebouwd.nl/artikel/20180412-circulaire-horecapaviljoen-green-house-in-utrecht-geopend</u>
- Dwarshuis, K. (2018). Earthquakes in the Groningen Gas Field, the Netherlands. Retrieved from <u>https://dwarshuis.com/earthquakes-groningen-gas-field/</u>
- Dym, C. L., Little, P., & Orwin, E. J. (2004). *Engineering Design: A Project-Based Introduction,* 4th Edition.
- Ellen Macarthur Foundation & Granta Design. (2015). CIRCULARITY INDICATORS An Approach to Measuring Circularity.
- EMF. (2013). Towards the circular economy, Economic and business rationale for an accelerated transition. Ellen Macarthur Foundation Rethink the Future. Retrieved from <u>https://www.ellenmacarthurfoundation.org/assets/downloads/publications/Ellen-MacArthur-Foundation-Towards-the-Circular-Economy-vol.1.pdf</u>
- EMF. (2016). *Intelligent assets: Unlocking the circular economy potential:* Ellen MacArthur Foundation.
- EMF. (2017). What is a circular economy? A framework for an economy that is restorative and regenerative by design. Retrieved from

https://www.ellenmacarthurfoundation.org/circular-economy/concept

EnergySaver. (n.d.). Energy-Efficient Home Design. Retrieved from <u>https://www.energy.gov/energysaver/design/energy-efficient-home-design</u>

- Enterprise. (Ed.) (2003) (11th ed.). Springfield: Merriam-Webster.
- EU. (2011). 2050 low-carbon economy. *Climate strategies & targets*. Retrieved from <u>https://ec.europa.eu/clima/policies/strategies/2050_en#tab-0-0</u>
- Fabius, L. (2015). *Adoption of the Paris Agreement*. Retrieved from https://unfccc.int/resource/docs/2015/cop21/eng/l09r01.pdf.
- Flyvbjerg, B. (2014). What You Should Know About Megaprojects and Why: An Overview (Vol. 45).
- Fox, S., Leicht, R. M., & Messner, J. I. (2010). Assessing the Relevance of Media Synchronicity Theory to the Use of Communication Media in the AECO Industry. *Journal of Architectural Engineering*, 16(2).
- Frankel, L., & Racine, M. (2010). *The Complex Field of Research: for Design, through Design, and about Design*. Paper presented at the Proceedings of the Design Research Society (DRS) International Conference, Montréal.

http://www.designresearchsociety.org/joomla/proceedings/drs-2010.html35

- Franssen, N., Nema Nagendra Kumar, P., & Reinders, B. (2018). Urban mining of earthquake affected structures for construction of new building projects for "The Groningen Project" at Royal HaskoningDHV. Joint Interdisciplinary Project: Delft University of Technology. Rotterdam.
- Geissdoerfer, M., Savaget, P., Bocken, N. M. P., & Hultink, E. J. (2017). The Circular Economy – A new sustainability paradigm? *Journal of Cleaner Production*, 143, 757-768. doi:<u>https://doi.org/10.1016/j.jclepro.2016.12.048</u>
- Geldermans, B., Wandl, A., Meister, K., Munoz Unceta, P., Kamps, M., & Streefland, T. (2019). Resource Management in Peri-urban Areas: Going Beyond Urban Metabolism;
 ADDENDUM D3.3 (Process model for the two pilot cases) CDW in the Amsterdam Metropolitan Area. Retrieved from Delft:
- Geldermans, R. J. (2016). Design for Change and Circularity Accommodating Circular Material & Product Flows in Construction. *Energy Procedia*, 96, 301-311. doi:https://doi.org/10.1016/j.egypro.2016.09.153
- Gemeente Amsterdam, Concepts, A., & Copper8. (2017). *De impact van circulair bouwen op de bouw-en investeringskosten*. Retrieved from Amsterdam: <u>https://www.amsterdam.nl/wonen-leefomgeving/duurzaam-amsterdam/publicaties/impact-circulair/</u>
- Haije, M., Schüren, L., van Sluis, B., & Wind, M. (2018). Het Verdwenen Groningen. Retrieved from <u>http://database.hetverdwenengroningen.nl/</u>

- Hardcastle, J. L. (2015, 23-12-2015). Sustainable Development Goals: Why Business Should Care. Retrieved from <u>https://www.environmentalleader.com/2015/12/sustainable-</u> <u>development-goals-why-business-should-care/</u>
- Harrison, C., May, G., Dr. Mills-Lamptey, B., Griffiths, P. W., Barbara, Mitchell, D., & Keniston, M. (2016, 2016). bio-bean, about us. Retrieved from <u>https://www.bio-bean.com/about-us/</u>

HaskoningDHV, R. (2018). VIIA Projecten - 495M [JSON / IFC].

- Haytink, T. G., & Valk, H. J. J. (2015). *Gerealiseerde Nieuwbouw Woningen met EPC < 0,4, In opdracht van het ministerie van Binnenlandse Zaken en Koninkrijksrelaties.* Utrecht: Rijksoverheid.
- Huuhka, S. (2016). Vacant residential buildings as potential reserves: a geographical and statistical study. *Building Research & Information, 44*(8), 816-839. doi:10.1080/09613218.2016.1107316
- Isaac, R. (2018). Restitching the Common Thread The Potential of Closed Loop Recycling in the Textile and Clothing Industry for Regional and Entrepreneurial Resilience in Northern Portugal.
- Jansen, S. J. T., Hoekstra, J. S. C. M., & Boumeester, H. J. F. M. (2017). The impact of earthquakes on the intention to move: Fight or flight? *Journal of Environmental Psychology*.
- Kanyemesha, M. (2017). 100.000 Energy Neutral Homes. Retrieved from https://amsterdamsmartcity.com/projects/100000-energy-neutral-homes#about
- Kashyap, S. (2017, 28-02-2017). Introductory guide on Linear Programming for (aspiring) data scientists. *Datahack Summit 2019*. Retrieved from <u>https://www.analyticsvidhya.com/blog/2017/02/lintroductory-guide-on-linear-</u> programming-explained-in-simple-english/
- Kleinjan, B. (2017). DREIGEND TEKORT AAN BOUWMATERIALEN IN NEDERLAND. Retrieved from <u>https://www.bnr.nl/nieuws/economie/10322255/dreigend-tekort-aan-bouwmaterialen-in-nederland</u>
- KNAG. (2016, 24 september 2016). KNAG-excursie Groningen. Retrieved from <u>https://geografie.nl/agenda/knag-excursie-groningen</u>
- Koutamanis, A. (2018, 25-10-2018) Graduation proposal meeting Niels Franssen/Interviewer: N. Franssen.
- Koutamanis, A., van Reijn, B., & van Bueren, E. (2018). Urban mining and buildings: A review of possibilities and limitations. *Resources, Conservation and Recycling, 138, 32-*39. doi:<u>https://doi.org/10.1016/j.resconrec.2018.06.024</u>
- Krijnen, M. (2015a). *Referentiewoning EPC 0,4 Appartementencomplex*. Eindhoven: Royal HaskoningDHV, in opdracht van RVO.
- Krijnen, M. (2015b). *Referentiewoning EPC 0,4 Galerijwoning*. Eindhoven: Royal HaskoningDHV, in opdracht van RVO.
- Krijnen, M. (2015c). *Referentiewoning EPC 0,4 Hoekwoning*. Eindhoven: Royal HaskoningDHV, in opdracht van RVO.
- Krijnen, M. (2015d). *Referentiewoning EPC 0,4 Tussenwoning*. Eindhoven: Royal HaskoningDHV, in opdracht van RVO.
- Krijnen, M. (2015e). *Referentiewoning EPC 0,4 Twee-onder-eenkapwoning*. Eindhoven: Royal HaskoningDHV, in opdracht van RVO.
- Krijnen, M. (2015f). *Referentiewoning EPC 0,4 Vrijstaande woning*. Eindhoven: Royal HaskoningDHV, in opdracht van RVO.
- Krook, J., & Baas, L. (2013). Getting serious about mining the technosphere: a review of recent landfill mining and urban mining research. *Journal of Cleaner Production*, 55, 1-9. doi:<u>https://doi.org/10.1016/j.jclepro.2013.04.043</u>

- Kumar, R. (2011). *Research Methodology, a step-by-step guide for beginners* (3th edition ed.). Los Angeles, London, Thousand Oaks, New Delhi, Singapore: SAGE.
- Kurzawska, K. (2018, 09-02-2019). 5 Ways in Which Digitization Is Changing Business World. *Business*. Retrieved from <u>https://www.timecamp.com/blog/2018/02/5-ways-digitization-is-changing-business-world/</u>
- Lansink, A. (1979, 2017). Ladder van Lansink. Retrieved from <u>http://www.adlansink.nl/voorbeeld-pagina/</u>
- Leijten, J. (2018). Zwaarste aardbeving in jaren in Groningen. Retrieved from <u>https://www.nrc.nl/nieuws/2018/01/08/zware-aardbeving-in-groningen-3-4-op-schaal-van-richter-a1587514</u>
- Li, J. (2015). Wastes could be resources and cities could be mines. *Waste Management & Research, 33*(4), 301-302. doi:10.1177/0734242X15581268
- LINDO Systems inc. (2019). What'sBest! 16.0 Excel Add-In for Linear, Nonlinear, and Integer Modeling and Optimization. Retrieved from
- https://www.lindo.com/index.php/products/what-sbest-and-excel-optimization Maas, W. (2018). Residents and MVRDV develop a vision for a village battered by gas extraction. Retrieved from https://www.mvrdv.nl/en/news/residents-and-mvrdv-

develop-a-vision-for-a-village-battered-by-gas-extraction

- Madaster. (2019). About us. Retrieved from https://www.madaster.com/en/about-us
- Madaster, TNO, & Alba Concepts. (2018). *Madaster Functionele Omschrijving V2.0*. Retrieved from
- NEN. (1996). NEN 2660:1996.
- NIBE. (2011). NIBE Experts in Sustainability. Retrieved from https://www.nibe.org/nl
- Nieman, H. M. (2015). SBR-referentiedetails woningbouw 2015 Studenteneditie (Vol. 8): Stichting Isso.
- Nijhuis, S., de Vries, J., & Noortman, A. (2017). Ontwerpend onderzoek (pp. 257-263).
- Nu.nl. (2019, 22-05-2019). Ruim tweehonderd schademeldingen na forse aardbeving in Groningen. Retrieved from <u>https://www.nu.nl/gaswinning-</u> <u>groningen/5903499/ruim-tweehonderd-schademeldingen-na-forse-aardbeving-in-</u> <u>groningen.html?redirect=I</u>
- Nußholz, J. L. K. (2018). A circular business model mapping tool for creating value from prolonged product lifetime and closed material loops. *Journal of Cleaner Production,* 197, 185-194. doi:<u>https://doi.org/10.1016/j.jclepro.2018.06.112</u>
- Pranjić, A. M., & Mladenović, A. (2018). *Cinderela Newsletter #1*. European Union's Horizon 2020 Research and Innovation Programme under Grant Agreement no 776751. Ljubljana.
- Rau, T. M., & Overhuber, S. (2016). *Material matters; Het alternatief voor onse foorbouwmaatschappij:* BL+.
- Re Use Materials. (2019). Cirdax Materialen Management Systeem. Retrieved from <u>https://www.cirdax.com/</u>
- Rijksinstituut voor Volksgezondheid en Milieu. (2016). Need for circular use of materials in construction. Retrieved from <u>https://www.rivm.nl/en/news/need-for-circular-use-of-materials-in-construction</u>
- Rijksoverheid. (2016a). A Circular Economy in the Netherlands by 2050, Government-wide Programme for a Circular Economy. Den Haag: Rijksoverheid.
- Rijksoverheid. (2016b). *Energieagenda, naar een CO*₂-*arme energievoorziening*. Den Haag: Rijksoverheid.
- Rijksoverheid. (2018). Staat van de Woningmarkt Jaarrapportage 2018. Den Haag.

- Rijksoverheid. (n.d.-a). *BENG-indicatoren*. Den Haag: Rijksoverheid Retrieved from <u>https://www.rvo.nl/onderwerpen/duurzaam-ondernemen/gebouwen/wetten-en-regels-gebouwen/nieuwbouw/beng-indicatoren</u>.
- Rijksoverheid. (n.d.-b). *The need for a circular economy*. Den Haag Retrieved from <u>https://www.government.nl/topics/circular-economy/the-need-for-a-circular-economy</u>.
- Rittel, H. W. J., & Webber, M. M. (1973). Dilemmas in a general theory of planning. *Policy Sciences*, *4*(2), 155-169. doi:10.1007/BF01405730
- RIVM. (2016). *Need for circular use of materials in construction*. RIVM Retrieved from https://www.rivm.nl/en/news/need-for-circular-use-of-materials-in-construction.
- Robinson, A. (2018, 17-04-2018). Construction is falling behind in the innovation race here's how we'll change that. Retrieved from <u>https://www.virgin.com/entrepreneur/construction-falling-behind-innovation-</u> race-heres-how-well-change
- Rowe, P. G. (1991). Design Thinking (3rd edition ed.). Cambridge, London: The MIT Press.
- Rowley, J. (2002). Using case studies in research. *Management Research News*, 25(1), 16-27. doi:10.1108/01409170210782990
- RoyalHaskoningDHV. (2018). Grondstoffenatlas MetropoolRegio Amsterdam. In FARICations & C. Economy (Eds.). Amsterdam.
- Saberi, S., Kouhizadeh, M., Sarkis, J., & Shen, L. (2019). Blockchain technology and its relationships to sustainable supply chain management. *International Journal of Production Research*, *57*(7), 2117-2135. doi:10.1080/00207543.2018.1533261
- Slowey, K. (2018, 13 03 2018). Report: Global construction waste will almost double by 2025. Retrieved from <u>https://www.constructiondive.com/news/report-global-</u> <u>construction-waste-will-almost-double-by-2025/518874/</u>
- Ten dam, I. (2018). *Circulair bouwen in de praktijk, ervaringen, inzichten en aanbevelingen*. Retrieved from Utrecht:
- Thelen, D., van Acoleyen, M., Huurman, W., Thomaes, T., van Brunschot, C., Edgerton, B., & Kubbinga, B. (2018). *Scaling the circular environment pathways for business and government*. Retrieved from Amsterdam: <u>https://docs.wbcsd.org/2018/12/Scaling_the_Circular_Built_Environment-</u> <u>pathways_for_business_and_government.pdf</u>
- Uber. (n.d.). Uber. Retrieved from https://en.wikipedia.org/wiki/Uber
- Ulubeyli, S., Kazaz, A., & Arslan, V. (2017). Construction and Demolition Waste Recycling Plants Revisited: Management Issues. *Procedia Engineering*, *1*72, 1190-1197. doi:<u>https://doi.org/10.1016/j.proeng.2017.02.139</u>
- UN. (2018). About the Sustainable Development Goals. Retrieved from <u>https://www.un.org/sustainabledevelopment/sustainable-development-goals/</u>
- UN Environment and International Energy Agency, U. E. a. I. E. A. (2017). *Towards a zeroemission, efficient, and resilient buildings and construction sector. Global Status Report* 2017.
- Van den Akker, J., Bannan, B., Kelly, A. E., Nieveen, N., & Plomp, T. (2013). *Educational Design Research Part A: An introduction*. Enschede: SLO: Netherlands institute for curriculum development.
- Van den Akker, J., Gravemeijer, K., McKenny, S., & Nieveen, N. (2006). *Educational Design Research*. London and New York: Routledge, Taylor & Francis Group.
- Van der Meulen, L. (2018). De Staat van de Woningmarkt 2018: aanpak woningnood. Retrieved from <u>https://www.omgevingsweb.nl/nieuws/de-staat-van-de-woningmarkt-2018-aanpak-woningnood</u>
- van der Voort, E. (n.d., n.d.). The Green House. Retrieved from <u>https://www.albron.nl/cases/the-green-house</u>

van Doorn, A. (2018, 27-03-2018). Heembouw deelnemer proef Excess Materials Exchange. Retrieved from <u>https://www.heembouw.nl/nieuws-</u>

<u>blog/nieuws/2018/maart/heembouw-deelnemer-proef-excess-materials-exchange/</u> Van Enckevort, S. (2019). Valuation materials over time: Alba Concepts.

- van Hemmen, B. (2016). Economic circularity in the built environment, An assessment and decision-making supporting model for the real estate sector & construction industry. MSc Thesis. Management in the Built Environment. Delft University of Technology. Delft.
- van Vliet, M. M. (2018). Disassembling the steps towards Building Circularity; Redeveloping the Building Disassembly assessment method in the Building Circularity Indicator. (MSc), Eindhoven University of Technology, Eindhoven.
- Verberne, J. (2016). *Building Circularity Indicators, An approach for measuring circularity of a building.* (Msc), Eindhoven University of Technology, Eindhoven.
- Verkuijlen, A. (2015). New Horizon Urban Mining. Retrieved from https://newhorizon.nl/urban-mining/over-ons-urban-mining/
- Vermeeren, E. (2018, 28-08-2018). Miljoen woningen erbij voor 2030: Gaat dat wel goed? . *Achtergrond*. Retrieved from <u>https://www.nu.nl/wonen/5427335/miljoen-woningen-</u> <u>erbij-2030-gaat-wel-goed.html</u>
- Verschuren, P., & Doorewaard, H. (2010). *Designing a Research Project* (Second ed.). The Hague: Eleven International Publishing.
- Vonk, N., de Jager, M., & Dirkx, I. (2019, 16-05-2019). Circulaire economie festival 2019. Retrieved from <u>https://www.smart-circle.org/circulareconomy/</u>
- Vrijhoef, R., & Koskela, L. (2005). A Critical Review of Construction as a Project-based Industry: Identifying Paths Towards a Project independent Approach to Construction *Proceedings CIB Combining Forces*. Helsinki.
- W/E Adviseurs. (2011). Voorbeeldwoningen 2011 Bestaande bouw. Utrecht: Agentschap NL, Energie en Klimaat.
- W/E Adviseurs. (n.d.). W/E Adviseurs; Maakt moeilijke materie makkelijk. Retrieved from https://www.w-e.nl/
- Wijnbergen, J. (2018). Inwoners uit aardbevingsgebied zijn er klaar mee: 'Het haalt niets uit'. Retrieved from <u>https://www.rtvnoord.nl/nieuws/198538/Inwoners-uit-</u> <u>aardbevingsgebied-zijn-er-klaar-mee-Het-haalt-niets-uit</u>
- Yin, R. K. (2014). *Case Study Research Design and Methods* (Vol. 5). Thousand Oaks, London, New Delhi: SAGE Publications
- ZAG, S. N. B. a. C. E. I. (2016). Cinderela, New Circular Economy Business Model for More Sustainable Urban Construction. Retrieved from <u>https://www.cinderela.eu/</u>
- Zeleny, M. (1982). Multiple Criteria Decision Making. New York: McGraw Hill.



PART 11 APPENDICES

пт	Appendix A · F)ata single	dwelling 'V	IA Projecten'
11.1	Appendix A. L	Jata single	uwening vi	IA I IOJECIEII

11.2	Appendix B: Operational model example
II.2.I	Main clients and goals
11.2.2	Mathematical elements
11.2.3	Model input
11.2.4	Model description
11.2.5	Model output
11.3	Appendix C: Hypothetical constraints input regional-based model
11.4	Appendix D: Example dataset structure Alba Concepts
11.5	Appendix E: Overview materials standard row-house analysis
11.6	Appendix F: Overview materials semi-detached house analysis
11.7	Appendix G: Overview materials detached house analysis
11.8	Appendix H: Disassembly factor weights and attribute weights
11.9	Appendix I: Overview costs virgin building materials

- II.IO Appendix J: Overview deconstruction / transportation costs reusable building materials
- II.II Appendix K: Overview of material validation reusable building materials
- II.12 Appendix L: Project-based model input
- II.13 Appendix M: Project-based model output

	Building element	#
0		8
1	Basic Wall:13_RHD_brick_220	16
2	Basic Wall:13_RHD_brick_50	8
3	Basic Wall:13_RHD_brick_55	14
4	Basic Wall:13_RHD_brick_65	4
5	Basic Wall:13_RHD_brick_70	21
6	Basic Wall:16_RHD_brick_220	16
7	Basic Wall:21_RHD_brick_110	10
8	Basic Wall:21_RHD_brick_220	17
9	Basic Wall:21_RHD_wood_plain_16-50-16	4
10	Basic Wall:22_RHD_HSB_50x55 hoh 400_2x gipsum_12.5	6
11	Basic Wall:22_RHD_HSB_50x75 hoh 400_2x gipsum_12.5	8
12	Basic Wall:22_RHD_HSB_50x85 hoh 400_2x gipsum_12.5	3
13	Basic Wall:22_RHD_HSB_Pui_110	2
14	Basic Wall:22_RHD_brick_110	5
15	Basic Wall:22_RHD_brick_200	1
16	Basic Wall:22_RHD_brick_220	1
17	Basic Wall:22_RHD_concrete_foamed_100	9
18	Basic Wall:22_RHD_concrete_insitu_100	4
19	Floor:13_RHD_brick_foundation_slab_100	4
20	Floor:13_RHD_brick_foundation_slab_315	1
21	Floor:13_RHD_concrete_insitu_foundation slab_100	1
22	Floor:23_RHD_beam system wood_50x120 hoh 600 + vloerdelen 25mm	1
23	Floor:23_RHD_beam system wood_60x175 hoh 920 + vloerdelen 25mm	1
24	Floor:23_RHD_beam system wood_70x230 hoh 660 + vloerdelen 25mm	2
25	Floor:23_RHD_beam_system_wood_55x175 hoh 1000 +wood_22	1
26	Floor:23_RHD_concrete_insitu_120	2
27	Floor:23_RHD_concrete_insitu_130	1
28	Floor:23_RHD_concrete_insitu_150	1
29	Floor:23_RHD_concrete_insitu_65	2
30	RHD_concrete_insitu_reinforced	1
31	RHD_stone_brick	1
32	RHD_wood_45	12
		188

11.1 APPENDIX A: DATA SINGLE DWELLING 'VIIA PROJECTEN'

Figure 11.1 Data single dwelling (HaskoningDHV, 2018)



11.2 APPENDIX B: OPERATIONAL MODEL EXAMPLE

This Appendix explains the operational model created for the JIP and the graduation research. A notion which should be made about this explanation is the fact that the model and the results are not final yet. This is just a setup of the model and an elaboration of how the model works and finally should work. The model is also used for the course Operations Research Methods (AR3Ro58).

11.2.1 MAIN CLIENTS AND GOALS

The goal is to optimize the use of materials gathered by urban mining into new housing projects. The number of materials gathered by the data supplying company, Royal HaskoningDHV, are available for this model. It contains information about many existing buildings in the province of Groningen, which mostly are endangered by the earthquakes because of the gas extractions. The municipality of Groningen, which is the direct client and so the main stakeholder of this project, is overall working together with companies such as Royal HaskoningDHV to tackle the endangered living situations (Figure II.2). Most of the projects concern renovation and reconstructing, but the process of new earthquake resisting buildings is a subject where urban mining comes into action.



Figure II.2 Main stakeholders, Royal HaskoningDHV (contractor) & Municipality of Groningen (client)

For the upcoming housing projects within the Groningen area, the usage of current building materials of the endangered buildings could be an efficient and circular option. The goal of this model is to research how to optimize the use of the materials gathered from the current buildings into the new projects (Figure 11.3). Eventually, this would help to achieve the goal of reducing the usage of virgin materials in housing projects.

What is the best division of materials used, which materials are possible for reuse and how do you define the different types of buildings which could be built mostly out of old materials? During the setup of this model, there is an ongoing decision-making process about which of the chosen elements should be maximized or minimalized.



Figure 11.3 Research goal (own ill.)

11.2.2 MATHEMATICAL ELEMENTS

When setting up the design model, certain variables and constraints must be determined to define the limitations of the model. The data, provided by Royal HaskoningDHV consists of an overview of buildings and the materials of which the dwellings consist of. The database is designed as an output of .ifc and .json files, which are the type of files provided by Royal HaskoningDHV. In the database overview, all the materials in the buildings have been inventoried. The amount of building materials what is delivered by Royal HaskoningDHV depends on the amount of building files which are available. In the setup of this model, a total of 7 buildings are used for the input. The overall amount of these materials will be used as the constraints for the model. In Appendix II.I, an overview a single house and the total overview of all the materials used for this dwelling are shown. To make the model more realistic, the number of houses used for the model has been multiplied by 100 times, which gives more materials to use for the possible optimization.

The constraints for the model exist of different building elements, such as walls, floors, and beams. The type of material which these elements consists off differ, but for this model, the four most common materials have been chosen to use as input for the linear programming. The chosen materials are wood, concrete, steel, and bricks (Figure II.4). In the model, these constraints are mentioned as recovered materials.

The variables are the different building elements which are necessary to build a new house. The list of elements needed is much longer for a house to have it completely functioning, but for this model, the three main building elements have been chosen as variables. As mentioned earlier, the chosen building elements are walls, floors, and beams (Figure 11.5).



11.2.3 MODEL INPUT

The variables and constraints for the model have been mentioned in the previous chapter. These will assist to formulate the input for the model. The goal of the model is to find the most optimal division of material to build new housing projects. So, for the input of the model, first, the different possibilities of houses should be defined. Within a city, there are a lot of different types of building, from residential to business to public buildings. To design the buildings in the model in a real sense and to listen to the client's wishes, this model focusses on three different types of residential buildings, which are apartments, terraced housing and detached housing. Figure 11.6 shows an example of existing houses within Groningen, along with the number of square meters per house, the number of



2 rooms € 175.000 k.k. 4 rooms € 369.000 k.k.

Figure 11.6 Dwelling types and information (Funda)

€ 559.000 k.k.

rooms and the costs of the dwelling, when to be bought. The information of the example dwellings has been derived from different ads at https://www.funda.nl/.



Figure 11.7 Conceptual definition of dwelling design (own ill.)

These conceptual new housing projects, which are another variable to control, demand a certain amount of building elements (variables) of different materials (constraints). One of the tasks to regulate the input of the model is defined the amount of each element required, to design three different dwellings from scratch. In Figure 11.7, it shows the division of materials per building element made which is used for the model. For every different dwelling type, a different amount of building elements is defined, which provides the input for the model.

Each dwelling consists of a different amount of building elements and materials. To operationalize the research question and simplify the model, each dwelling types has a certain amount defined, based upon example projects provided by Royal HaskoningDHV. Table I shows three example buildings, which division of elements provides the amounts used for the definition of the dwellings. These three dwellings are existing buildings, located somewhere in Groningen. Because of privacy legislation, the actual location of these dwelling remains unknown. The input which is shown within Table II.I is not the final numbers yet. They are still in the process of being defined and will be changed during the ongoing graduation project. For the explanation of the model, the following numbers will be used which are previewed in table I. Every value presents a possible number of elements of a specific material necessary to build the dwelling. The numbers are interchangeable, depending on the qualities of dwelling the client wants.



Table II.I Overall input operational model (own ill.)

11.2.4 MODEL DESCRIPTION

The input of the model is defined in section II.2.3. An example of the data of building elements and materials of a single dwelling used for the model is visible in Appendix II.I. The "What'sBest!" method within Microsoft Excel, will solve the model which is visible in Table II.2. This screenshot is a direct outcome of the input which is shown in Table II.2above.

Anound of colspan="2">Anound of colspan="2" Anound of	RHS 0 0 0
Description Upper: Description Interaction Interaction <t< th=""><th>RHS 0 0 178700</th></t<>	RHS 0 0 178700
Uniforme 301 641 180 43001 644.02 COUND Objective function 1 <th>RHS 0 0 178700</th>	RHS 0 0 178700
Objective function 1 <th1< th=""> 1 1</th1<>	RHS 0 0 178700
Unite Control (Control (Contro) (Control (Contro) (Control (Contro) (Control (Co	RHS 0 0 178700
Waits Constant form Sign Book for densing type AB.0 -40 -121 -137 1 0 = Concrete for densing type AB.0 -200 -30 -30 1 0 = Wood for densing type AB.0 -200 -30 -30 1 0 = Wood for densing type AB.0 -10 -35 -46 1 0 = Monant of densing concrete -10 -35 -46 1 0 = Amount of densing concrete -1 -10 -10 54/372 = -10 54/372 =	RHS 0 0 178700
Inscit of weeking type A.B.C 4-00 -1.21 -1.37 1 0 = Concrete for develop type A.B.C -0.00 -3.00 -1.0 -0.0 = 0 0 = 0 0 = 4 0 0 = 4 0 = 4 0 1 0 = 4 0 = 4 4 1 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 </th <th>0 0 178700</th>	0 0 178700
Concrete for dealing type A.B.C -200 -30 -30 1 0 = Wood for dealing type A.B.C -10 -35 -46 1 0 = Promotif or dealing type A.B.C -10 -35 -46 1 0 = Amount of existing type A.B.C -10 -35 -46 1 0 = Amount of existing type A.B.C -10 -35 -46 1 0 = Amount of existing type A.B.C -10 1 0 1 0 1 1 13333 = 1 13333 = 1 13333 = 1 13333 = 1 13333 = 1 13333 = 1 13333 = 1 13333 = 1 1 13333 = 1 1 13333 = 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	0 0 178700
Wood to desting type A B,0 -10 -35 -46 1 0 = Annual of cesting concele 1 64512 64512 64512 64512 64512 64512 54512 <	178700
Amount of existing brick 1 445976 cit Amount of existing concrete 1 64429 cit Amount of existing word 1 113333 cit	178700
Immunit of existing concrete 1 45076 cv Amount of existing concrete 1 84432 cr Amount of existing concrete 1 15333 cr	178700
Amount of existing concrete	
Amount of existing wood 1 13633 <=	121500
	25000
Floors	
Concrete for dwelling type A.B.C -100 -25 -20 1 0 =	0
Wood for dwelling type A,B,C -5 -20 -12 1 0 =	0
	10000
Amount of existing concrete	43800
Amount of existing wood	5400
Beams	
Steel for dwelling type A,B,C -80 -40 -60 = 1 0 =	0
Wood for dwelling type A.B.C -50 -100 -150 = 1 0 =	0
Amount of existing steel 1 43238 <=	50000
Amount of existing wood 1 50000 ==<	50000

Table II.2 Operational model Excel (own ill.)

The model calculates the optimal amount of dwelling types (A, B and C). In the formula, the number of dwellings will be put down as NA, NB and NC. It also calculates the number of materials used for each building element. If we, for example, take the brick walls, we will define the total amount of brick walls used Mbw. The mentioned amount of materials needed for every building element of every dwelling type, visible in table 8.I, will be mentioned as b, in the case of bricks. The number of brick walls needed for example for dwelling A will be mentioned as bIW.

The model calculates the total amount of the different materials used and generates the amount of the three dwellings types to build. The total amount of for example bricks walls is determined by the

number of dwelling types (A, B and C) and the required amount of bricks walls for each dwelling type. This gives the following equation:

$$M_{bw} = b_{I}w * N_{A} + b_{2}w * N_{B} + b_{3}w * N_{c}$$

There is a maximum of material to consider while calculating this model, which is the building elements available for every material. In the case of brick walls, the restriction regards to the maximum amount should be mentioned as:

 $M_{bw} \leq b_{ow}$

The restriction is mentioned in the rightest column of the model for each material.

The same formula could be repeated for the following elements with the adjoining variables:

•	Brick walls:	$M_b w$
•	Concrete walls:	$M_c w$
•	Wooden walls:	$M_w w$
•	Concrete floors:	$M_c f$
•	Wooden floors:	$M_w f$
•	Steel beams:	$M_s b$
•	Wooden beams:	$M_w b$

11.2.5 MODEL OUTPUT

In the outline of Table II.3, the specific results of the first example are shown. For this example, the input of Table II.1 is used. It shows in blue the number of houses which are possible to build with the number of materials of different elements used. The sum-product of the results shows the maximum output of this equation. It calculates the maximum output of both the materials and the number of houses able to build. In this case, mostly apartments will be built and all the concrete and wooden floors are used for the construction of the houses.



The results in Table II.3 are just one of the multiple outcomes trying to make the model work. Also, the input is constantly changing, depending on what amount of materials are available. It is important to notice what the different inputs deliver as results and to keep track of the different possibilities within the project.

11.3 APPENDIX C: HYPOTHETICAL CONSTRAINTS INPUT REGIONAL-BASED MODEL

Constraints:

- Incoming materials in building sector, waste from other projects. List of materials in the building sector which could possibly be used within the model:

	<u>Material</u>	Group
0	Isolation	Internal Wall
0	Steel	Structural
0	Concrete facades	External wall / façade
0	EPDM Roofing	Roofing
0	Sand-lime bricks	Internal Wall / Structural
0	Aluminum	External wall / façade
0	Concrete	Structural
0	Cement	Structural
0	Glass	External wall / façade
0	Plasterboard	Internal Wall
0	Stucco	Internal Wall
0	Bricks	External wall / façade / structural
0	Isolation material	Internal Wall
0	Laminate flooring	Flooring
0	Carpet	Flooring
0	Floor tiles	Flooring
0	Wood	Structural
0	Styrofoam	Internal Wall
0	Plastic	Structural
0	Roof tiles	Roofing

<u>Costs</u>

- Processing
- Transportation
- Demolition
- o Implementation

Revenues

- From adjoining companies
- For each matchmaking

Groups of types of building materials

Internal Wall	Isolation
	Sand-lime bricks
	Plasterboard
	Stucco
	Isolation material
	Styrofoam
External wall / façade	Concrete facades
	Aluminum
	Glass
	Bricks

Structural	Steel
	Concrete
	Cement
	Bricks
	Wood
	Plastic
Roofing	EPDM Roofing
	Roof tiles
Flooring	Laminate flooring
	Carpet
	Floor tiles

 Table 11.4 Groups of type building element and material (RoyalHaskoningDHV, 2018)

Project Titel Datum Auteur Versie	Project Doen! Renovatie Westelijk st. Inventarisatie hergebruik SPW 30 Arn 3-de-18 BIS 2.0	tionsgebied Adviewskep (Exter vas kannik) hom 8 Arryden 5 Arryden 5 Akurten 1 Goed 2 Gendatu 3 kang		A Prevent B Hungde C filegele D Energie E Vielene F Storten							al	ba
nr.	Onderdeel	Opmarking: De maitvoering betreft een ind	Foto	Materiaal	Afvalstrategi	e Conditie	Demontage	Afzetmogelijkheid	i Aantal (st) L	engte (mm)	Breedte (mm)	Hoogte (mm
Fundering			1	P	1	1			T			
Skelet	1	- I	-1	1	- 1	- I			. J			
27	Daken (skelet) Dakraam / lichtkoepel		86		Hergebruik	Gemiddeld			1			
	Glazenwas installatie		87		Hergebruik	Onbekend			3			
Dakafbouv	v/dakafwerking					1 1			1 1			
37	Dakopeningen Zonwering lichtkoepel		882	Metaal	Hergebruik	Goed			1			
Gevelafbo 31	uw/gevelafwerking Buitenwandopeningen		1	1	1				1	1	1	
	Kozijn vluchtdeur	Brandwerend	14	Hout	Hergebruik	Goed	0	0	1		1.263	2.500
	videntaeor	branowerend		nout	Heigebruik	Gued			1		1.155	2.35
41	Buitenwandafwerkingen Gevelpui	Ter plaatse van entree	104	Aluminium	Hergebruik	Goed	+	0	1		5.398	2.47
	Gevelpui	Ter plaatse van entree	105	Aluminium	Hergebruik	Goed	+	0	1		2.780	2.19
4.947 (F. 1. 1. 1. 1. 1. 1.	Gevelkozijnen				Recyclen	Gemiddeld	0		n.t.b.	** > ** *	divers	diver
Binnenwa	ndafbouw/ binnenwandafwerking		1	1	1	1 1			r r			
	Kozijn pul		50	Staal en glas	Hergebruik	Goed	+	+	1		6.580	2.034
	Kozijn pul Kozijn pul		58	Hout en glas	Hergebruik	Goed	+	*	4		640 1.955	2.533
	Metal stud scheidingswand	1e t/m 6e verdieping			Recyclen	Gemiddeld	-		n.t.b.		1.333	2.331
32	Binnenwandopeningen											
	Binnenkozijn wandraam		1	Hout en glas	Hergebruik	Gemiddeld	+	+	1		1.453	2.190
	Duppele binnenkozijn deur Deur		3	Hout	Hergebruik	Gemiddeld	÷	÷	1		1.770	2.162
	Deudranger	Merk Dorma	16	Aluminium	Hergebruik	Goed	+	<u>+</u>	115			
	Deunkiem Kozijn deur	Brandwerend	28	Hout	Hergebruik Hergebruik	Goed	+	0	2		1.015	2.055
	Deur	Brandwerend	25	Hout	Hergebruik	Goed	+	+	1		870	1.970
	Kozijn deur Deur	Toiletdeur Toiletdeur	115	Hout	Hergebruik	Gemiddeld	0	0	57		920 825	2.165
	Binnenkozijn wandraam	Veiligheidsglas	33	Hout en glas	Hergebruik	Gemiddeld	+	+	4		4.286	2.27
	Binnenközijn wandraam Deurközijn met zijlicht	Veiligheidsglas Veiligheidsglas (14 per verdieping)	34	Hout en glas	Hergebruik	Gemiddeld	+	+	40		2.630	2.413
	Deur	······································	36	Hout	Hergebruik	Goed	+	•	40		825	2.110
	Kozijn deur Deur vluchtroute	Maatvoering raam: b430 en h305	41	Hout en glas Hout	Hergebruik	Gemiddeld	+	:	6		1.000	2.160
	Deurkozijn met zijlichten		45	Hout	Hergebruik	Gemiddeld	+	+	20		1.773	2.266
	Deur Deurklem	Positie vloer	41	Hout	Hergebruik	Goed	+	+	1		920	2.216
	Kozijn deur		51	Hout	Hergebruik	Gemiddeld	+	*	16		1.773	2.160
	Kozijn deur	Beschadigd door constructieve balk	51	Hout	Hergebruik	Gemiddeld	+	+	4		2.359	2.110
	Schuifdeur kozijn Sebuidfeur	t.h.v. wenteltrap, veiligheidsglas (b=820, h=590)	54	Hout	Hergebruik	Gemiddeld	0	-	1		820	590
	Deurdranger schulfdeur	Elektronisch	56	Hout	Hergebruik	Goed	+		2		1.015	2.085
	Kozijn deur		65	Hout	Hergebruik	Gemiddeld	0	0	2		3.001	3.287
	Kozijn deur	Giaz 0=600 6n n=985	67	Hout en gias	Hergebruik	Gemiddeld	0	+ 0	7		2.050	3.287
	Kozijn deur	Provide analysis 2020	68	Hout	Hergebruik	Gemiddeld	0	0	7		2.500	3.287
	Gevangenisdeur	Verzwaarde deur	70	Hout	Hergebruik	Goed	+		7		1.010	2.019
	Kozijn gevangenisdeur Kozijn deur	Verzwaarde deur	71	Hout	Hergebruik	Goed	+	+	7		1.050	2.075
	Deur	b=600, h=985	73	Hout en glas	Hergebruik	Goed	+	+	3		930	2.400
	Kozijn deur		77	Hout on clas	Hergebruik	Gemiddeld	0	0	2		3.752	2.47
	Kozijn deur	too - Toolog Anton alcount allog battle transmission and the	79	Hout	Hergebruik	Gemiddeld	0	0	2	1	4.050	2.475
	Hang en sluitwerk Hang en sluitwerk		80	RVS	Hergebruik	Goed	+	+	2			
	Schuifdeur (automatisch)	hoogte deur h=2206	83	Aluminium	Hergebruik	Goed	÷	0	1		2.695	2.52
	Kozijn deur Deur		88	Staal	Hergebruik	Gemiddeld	0 +	0	55		1.375	2.65
	Kozijn deur		90	Aluminium	Hergebruik	Gemiddeld	0	0	6		1.744	2.645
	Deur Systeemwand		91	Hout Kunststof	Hergebruik	Goed	+	+	6		1.675	2.130
	Kozijn deur		96	Staal	Hergebruik	Gemiddeld	0	0	6		1.504	2.38
	Deur Systeemwand		97	Hout	Hergebruik	Goed		+	6		1.375	2.65
	Binnenwand entrée		100	Hout en glas	Hergebruik	Goed			2		4.804	2.595
	Deur binnenwand entrée Deur meterkast		101	Hout en glas	Hergebruik	Goed	++	+	1		1.867	2.533
	Schuifdeur entrée	Breedte schuifdeur b = 1520	103	Glas	Hergebruik	Goed		1	2		4.930	2.544
	beur Deur		100	Hout en glas	Hergebruik	Goed	*	0	1		2.715	2.166
lloarst	uu/ doarafuarkiar								1			
37	Vloerafwerkingen		1	1					T T	1		
	Marmoleum	begane grond, 1e verd., 2e verd., 3e verd. (deels).			Recyclen	Laag			n.t.b.			
	Tegelvioer	tpv trappenhuizen, toiletten	-		Hergebruik	Gemiddeld	-		n.t.b.			
Transen a	n hellinghanen	1	1		1				1	1		
14	Trappen en hellingconstructies		1						L L	1	1	
	Trapleuning Trapleuning	2 per verdieping	21		Hergebruik	Goed	+	•	7			
			04		nie Scoronk							
84	Balustrades en leuningen Balustrade vide	Hooste = 1000		Staal	Hergebrude	Goed			1			
	Balustrade dakterras	Hoogte = 720	94	Staal	Hergebruik	Goed			+/- 10 m1			
14	Trao- en hellingafwerkingen											
	Wenteltrap	Staal	53	Staal	Hergebruik	Goed	+	-	1			
lafonds	I			I					.II		l	
15	Plafondafwerkingen		1									
	Platond raster Systeemplafond	Rasterprofielen (per verdieping) Stalen roosters	13	Staal Staal	Hergebruik	Gemiddeld	+ +	0	1800 m2 30 m2	600	600 600	
Ntb.: vloe	s istof- en gasinstallaties											
13	Water											
	Signalering brandslanghaspel		10		Hergebruik	Goed	++	:	28			
	the second s					a second to be set as a						

			1	1		1				1	1	
Wtb.: kl	imaatinstallaties											
55	Klimaatinstallatie: koeling								1			
	Airco unit		4	4	Hergebruik	Gemiddeld	+	0	+/-5			
56	Klimaatinstallatie: verwarming											
	CV ketel	Bouwjaar omstreeks 2010	4	3	Hergebruik	Gemiddeld	+	+	1			
	CV ketel	Ketel 1 buiten gebruik, bouwjaar 1990	10	8	Recyclen	Laag			2			
	CV ketel	Bouwjaar 2005	10	9	Recyclen	Gemiddeld	+	-	2			
	CV ketel	tbv beg. Grond KLPD, bouwjaar 2005	11	0	Recyclen	Gemiddeld			1			
	CV ketel	tby beg. Grond + 1e verd. bouwiaar 1990	11	1	Recyclen	Laag			1			
	Circulatiepomp	Recent vervangen	11	2	Hergebruik	Goed	+	+	1			
	Radiator	Ter plaatse van de kantoren	11	3	Recyclen	Gemiddeld	+	0	n.t.b.			
	Plaat radiator	Ter plaatse van de trappenhuizen	11	4	Hergebruik	Gemiddeld	+	0	n.t.b.			
57	Klimaatinstallatie: luchtbehandeling							1 (1) (1) (1) (1) (1) (1) (1) (1) (1) (1				
2000/1000010100	Ventilatierooster			7	Hergebruik	Goed	+	+	6	600	600	
	Ventilatierooster	2 roosters per kantoorruimte, circa 150 kantoren	3	2	Hergebruik	Gemiddeld	+	+	150	600	600	
	Ventilatiekoker parkeerkelder		6	1	Hergebruik	Gemiddeld	+	+	2			
	Ventilatierooster	2 roosters per kantoorruimte, circa 150 kantoren	7	6	Hergebruik	Gemiddeld	+		150	295	295	
	Luchtbehandelingskast	Bouwlaar 1991			Recyclen	Gemiddeld	0					
	Luchtbehandelingskast	Bouwiaar onbekend (schatting 1995)			Recyclen	Gemiddeld	0					
	0	and the second second second					-					
Wth : er	pergievoorziening verlichting	al a second a second		J					- L			
61	Centrale elektronische voorzieningen		1	1		1			T T	1	1	
	Reweringssensor		4	8	Herzebruik	Goed	+	+	35			
	Booster		4	0	Hergebruik	Gemiddold	1	0	ath	600	600	
	nouster				Heißenigik	Gerniudeid			11.1.10.	000	000	
63	Vallahtas											
03	venichcing		-		Harrison Barrielle	Card						
	Noodverlichting				Hergebruik	Goed	*	+				
	Noodverlichting			4	Hergebruik	Gemiddeid	+	· · · · · · · · · · · · · · · · · · ·	50			
	Verlichting	spaarlamp	3	-	Hergebruik	Gemiddeld	+	0	14	1 700		
	TL armatuur	4 tot 8 per kantoor, circa 150 kantoren	- 3	8	Hergebruik	Goed	*		900	1.500	300	
	IL armatuur	4 tot 8 per kantoor, circa 150 kantoren	3	9	Hergebruik	Goed	+		900	600	600	
	TL armatuur	4 tot 8 per kantoor, circa 150 kantoren	4	0	Hergebruik	Goed	+	· · · · · · · · · · · · · · · · · · ·	900	1.200	300	
	Wandlamp	Spaarlamp	6	3					14			
				1								
Elektra:	communicatie, beveiliging											
65	Beveiliging											
	Rooksensor			6	Hergebruik	Goed	+	+	56			
	Brandalarm hoorn		1	В	Hergebruik	Goed	+	+	56			
	Breekglas brandmelder		1	1	Hergebruik	Goed	+	+	56			
	Brandblusser		1	8	Hergebruik	Goed	+	+	42			
	Brandslangkast	inbouw	7	4	Hergebruik	Goed	0	+	n.t.b.	790	790	
	Breekglas brandmelder		7	5	Hergebruik	Goed	+	+	n.t.b.			
							11-11-11-11-11-11					
Transpo	rtinstallaties											
66	Liften, roltrappen, transportbanen, tran	nsportbuizen										
	Lift	Bouwjaar 1990	2	2	Recyclen	Laag			1			
10.5	Lift	Grote lift, bouwjaar 2011	8	4	Hergebruik	Goed			1			
	Lift	Lift, bouwjaar 2010	8	5	Hergebruik	Goed			1			
Vaste in	richtingen en voorzieningen											
Vaste in	richtingen en voorzieningen											
71	Vaste verkeersvoorziening		1	1					1 1	1	1	
	Stalen bekwerk parkeerkelder		6	Staal	Hereebruik	Goed	+	+	+/- 20 m1			
	our of the second second second			0.000		0000						
72	Vaste gebruikersvoorziening		10100	11-11-11-00-01		1001000000	101000000000000000000000000000000000000				0100100000	
	Zonwaring			Intestant	Hornsbruik	nomiddold			2		1 225	1 700
	Zonwering		0	alinetaal	Hergebruik	Good			40		1 209	2 200
	zonwennig				mergenium	GOEG					1.2.90	3.2.33
72	Vaste kaukon nostianingan		-									
/3	Vaste Keukenvoorzieningen				Manage barries	and dela						
	Neuken			4	nergebruik	gemiddeid	0	0				
74	Vaste sanitair voorzieningen											
	loinetpot	Diverse typen	2	5	Hergebruik	Gemiddeld	+	+	32			
	Toiltetbril	Diverse typen	2	4	Hergebruik	Gemiddeld	+	+	32			
	Spoelbak	Diverse typen	2	5	Hergebruik	Gemiddeld	+	+	6			
	Closetpot	Diverse typen	2	6	Hergebruik	Gemiddeld	+	+	2			
	Wasbak	Diverse typen	2	7	Hergebruik	Gemiddeld	+	+	10			
76	Vaste opslagvoorzieningen		-									
	Defibiriator kluis		5	7	Hergebruik	Goed	+	+	+/- 2			
	Whiteboard (kast)		8	2	Hergebruik	Goed	+	+	+/- 15		1.510	1.200
	Evac chair		9	9	Hergebruik	Goed	+	+	n.t.b.			
			-									
				-					-			

Table 11.5 Example dataset structure Alba Concepts

11.5 APPENDIX E: OVERVIEW MATERIALS STANDARD ROW-HOUSE ANALYSIS

alba			Dwelling A: Based on: Krijnen, M.	Standard (2015): Rej	i row-house Netherla ferentiewoning EPC 0,4 Tu:	nds isenwoning			
				Dwelling	information				
				G	ieneral		_		
Dimensions (NVO) Width	(m) NVO (m) 5,10	5,40	Volume BG		[m3]	NVO [m3] BVO 118,28 149	81		
Depth (BG / 1e)	8,92	9,70	1e 2e			118,28 149 84,86 110	81 89		
Depth (2e)	6,40	7,18	Total			321,42 410	50		_
Hight BG Hight 1e	2,60 2,60	2,86 2,86						Reusability scale reusability	
Hight 2e Hight total	4,64	4,90						possible reusability	
Building el	ements		Building ma	aterial / p	roduct			Reusability	
Element	Dimensions [m2]		Type	Dir	nensions (m2) Prod	uet	Gradatio	n Comment	
Site	[m2]			Fou	Indation				
Foundation	52,38		Concrete piles Wooden piles		?			Dmv trekken hergebruiken?	
					Floors				
Surface BG	[m2] NVO [m2 45,49	52,38	Channel plate floor		45,49			Dekvloer verwijderen	
le	45,49	52,38	Channel plate floor Concreet (landing)		35,00 10,50			Dekvloer verwijderen in werk gestort met prefab scheidingswand	
2e Total	32,64	38,77 143,53	Channel plate floor		32,64				
BG north	[m2]#			Exte	rior walls				
Window Window (door)	2,60 0.40	1	Glass + frame Glass + frame		2,60 0.40	1		Product moet zelfde maat zijn Product moet zelfde maat zijn	
Door Cavity wall facade	2,40	1	Front door Clickbrick		2,40	1		Product moet zelfde moat zijn	
Total	15.44		Sand-lime brick		8,50				
BG south	10,444								
Window Gavity wall facada	9,70	1	Glass + frame		9,70	1		Product moet zelfde maat zijn	
Tetel	5,/4		Sand-lime brick		5,74				
Total	15,44								
BG east/west Party wall (scheidingswand)	[m2] # 55,48		Prefab Concrete		55,48			Prefab elements, otherwise downcycle	
1a north	[m2] #								
Window Garitana I famila	5,2	2	Glass + frame		5,20	2		Product moet zelfde maat zijn	
Cavity wall racade	10,24		Sand-lime brick		10,24				
Total	15,44								
Le south Window	[m2] # 5,2	2	Glass + frame		5,20	2		Product moet zelfde maat zijn	
Cavity wall facade	10,24		Clickbrick Sand-lime brick		10,24 10,24				
Total	15,44								
1e exst/west Party wall (scheidingswand)	[m2] # 55,48		Prefab Concrete		55,48			Prefab elements, otherwise downcycle	
					Roof				
Ze north Window	[m2] # 1,4	1	Glass + frame		1,4	1		Product moet zelfde maat zijn	
Cavity wall facade Total	31,54 32,94		Roof tiles		31,54				
2e south	[m2] #								
Cavity wall facade	32,94		Roof tiles		32,94				
				Inte	rior walls				
BG Wall (nanels)	[m2] #		Prefab nanels Ytong (cellenheton) / Faav		19.64			Gedeeltelijk hernebruiken (uitzanen	
Door Codby wall panels	7,20	3	Corridor door		7,20	3		Product moet zelfde maat zijn	
Total	26,84								
1e	[m2] #		Professional Manual Antional Astronomy (22.15			Code In 1996 have show they fully	
Wall (panels) Door	42,75	4	Corridor door		33,15 9,60	4		Gedeelterijk hergebruiken luitzagen, Product moet zelfde maat zijn	
Cavity wall panels Total	33,15 42,75								
			Bui	lding mat	erial information				
Building	material (/ produc	t)			Constru	uction costs (virgin)		Soortelijk gewicht Ikg/m3]
Floors	Dimensions to	tal [m2]		(¢/	'm2] Over	view Reference			
Channel plate floor BG	entrensions to	45,49		6823,8 €	36,80 €	1.674,11 Channel plate floor 150mm 1.331.64 Channel plate floor 200mm		Beton	2100 kg
2e		32,64		£	38,05 €	1.241,95 Channel plate floor 200mm			
Exterior walls	Dimensions to	tal [m2]			100.00	13 194 31 Drafab P		Raton	2400
Clickbrick		34,73		e e	109,80 € 110,42 €	3.834,99 Clickbricks		Keramiek	∠100 kg 1800 kg
Interior walls	Dimensions to	tal [m2]						_	
Prefab panels (Ytong/Faay/other)		52,80		€	30,50 €	1.610,30 Faay VP 54		Faay	970 kg
Roof	Dimensions to	tal [m2]							
Roof tiles		64,48		¢	48,07 €	3.099,73 Ceramic tiles		Keramiek	3500 kg 38,6 kg
Windows	Dimensions to	tal (m2)	roducts [#]						
Glass + frame		24,50	8	¢	220,70 €	5.407,09 Meranti (wood) window		Glas Hout	2500 kg
Doors	Dimensionete	tal (m2)	roducts [#]						500 Kg
Front door	ownensions to	2,40	1 7	¢	595,00 €	595,00 Bossingpaneel 700.00 Opdekdour		Hout	500 kg
French door		0	0	Ę	595,00 €	- Stapeldorpel 1/2		Hout	F.0.5
Total		400,61 m2						nult	SUO kg

Table 11.6 Overview materials standard row-house analysis (own table)

11.6 APPENDIX F: OVERVIEW MATERIALS SEMI-DETACHED HOUSE ANALYSIS

alba

Dwelling B: Semidetached house Netherlands

Dimensions (NVO) Midth (dwelling) Vidth (HG garage ballway) Vidth (BG garage)							
Midth (BG garage hallway) Vidth (BG garage)	m] NVO [m] BVO 5,80 6	,36	Genera Volume BG	l (m3	NVO [m3] BVO 186,60 247,93		
	1,99 2, 2,87 3,	,57 ,18	te Ze Total		136,02 178,13 172,24 223,76 494.86 649.87		
epth (BG dwelling) epth (BG garage hallway)	9,02 9, 2,13 2,	,79 ,57					
epth (BG garage) epth (1e) enth (2e)	5,30 5, 9,02 9, 6.40 7	.61 .79 18					
ight DG	2,60 2	.86				Reusability scale	
ight 1e ight 2e	2,60 2, 4,64 4,	.86 .90				reusability possible reusability	
ight total	9,84 10,	,62				no reusability	
Building elements	Dimensions [m2]		Building materia	Dimensions [m2] Proc	luct Gradation	Reusability Comment	
ite	[m2]		Foundati	on			
oundation	86,69	_	Concrete piles Wooden piles	?		Dmv trekken hergebruiken?	
surface [[m2] NVO [m2] BV0		Floors				
.6 .e	71,77 86, 52,32 62,	.69 ,28	Channel plate floor Channel plate floor Concrett (landing)	71,77 37,46 14.85		Dekuloer verwijderen Dekuloer verwijderen In werk aestart met meefab scheidinaswand	
ie Jotal	37,12 45, 161,21 194	.66	Channel plate floor	37,12			
			Exterior w	alic			_
G north (i Vindow	[m2] # 1,50	1	Glass + frame	1,50	1	Product moet zelfde maat zijn	
oor iarage door avity wall garage	3,70 5,70 3,38	1	French doors Garage door Clickbrick	3,70 5,70 3,38	1	Product moet zelfde maat zijn Installatietechnisch niet haalbaar	
avity wall facade dwelling	12,99		Sand-lime brick Clickbrick	3,38 12,99			
otal facade dwelling	18,19		Sand-lime brick	12,99		-	
otal	27,27						
G south []	m2] # 3,70	1	Glass + frame	3,70	1	Product moet zelfde maat zijn	
oor avity wall facade	2,40 18,60	1	French doors Clickbrick Sand-lime brick	2,40 18,60 18.60	1	eroauct moet zeijde maat zijn	
otal	24,70	_				_	
Gwess arty wall (scheidingswand) avity wall facacle	28,01		Prefab Concrete Clickbrick	28,01 7 35		Prefab elements, otherwise downcycle	
iotal	35,36		Sand-lime brick	7,35			
G east ([m2] #	1	Glass + frame	1.50	1	Product most zelfde maat zijn	
boor Savity wall garage	2,40	î	Front door Clickbrick	2,40	î	Product moet zelfde maat zijn	
favity wall facade dwelling	43,43		Sand-lime brick Prefab Concrete	16,03 43,43			
otal facade dwelling fotal	47,33 63,36						
e north [[m2] #			2.52			
Vindow Sindow Savity wall facade	1,50 14,09	1	Glass + frame Glass + frame Clickbrick	2,60 1,50 14,09	1	Product moet zelfde maat zijn Product moet zelfde maat zijn	
otal	18,19		Sand-lime brick	14,09			
e south	(m2) # 2.60	1	Glass + frame	2.60	1	Product most zelfde maat zijn	
Vindow Javity wall facade	1,50 14,09	1	Glass + frame Clickbrick	1,50 14,09	1	Product moet zelfde maat zijn	
otal	18,19		Sand-lime brick	14,09		-	
e west arty wall (scheidingswand)	(m2] # 28,01		Prefab Concrete	28,01		Prefab elements, otherwise downcycle	
e cast Mintrue	[m2] #	1	Gloss + frame	0.40	1	Product most collida mont sin	
avity wall facade	27,61 28,01	_	Prefab Concrete	27,61	-	/ content inter a character	
e east	[m2] #		Drafab Concrete	20.02			
avity wan tacacat	20,33		Roof	20,02			
e norkh (i Nindow	m2] # 1,7	1	Glass + frame	1,7	1		
otal	38,80	_	Root bles	37,10		-	
e south avity wall facade	m2) # 38,80		Roof tiles	38,80		-	
			Interior w	alls			
G (panels)	m2] # 33,46	-	Prefab panels Ytong (cellenbeton) / Faay	26,26		Gedeeltelijk hergebruiken (uitzagen)	
oor avity wall facade rotal	7,20 26,26 33.46	3	Lorridor door	7,20	3		
le [(m2) #					_	
vall (penels) Joor Savity wall facade	43,95 9,60 34,35	4	Prelab panels Ytong (<i>cellenbeton</i>) / Faay Corridor door	34,35 9,60	4	Gedeeltelijk hergebruiken (uitzagen)	
otal	43,95						
-			Building material	information			
Building material (/ p	roduct)	51		Construction of Overview Fact	ors Reference	Soortelijk gewicht [kg/m3]	
leons	71, 37	,77		€ 36,80 € € 38,05 €	2.641,13 Channel plate floor 150mm 1.425,51 Channel plate floor 200mm	Beton	2100 k
nannet plate floor BG 1e	37	.12		€ 38,05 €	1.412,42 Channel plate floor 200mm		
Channel plate floor BG 1e 2e		.98 54		€ 109,80 € € 110,42 €	16.248,15 Prefab Beton 9.555,09 Clickbricks	Beton Keramiek	2100 kj 1800 k
numer plate floor BG 2e ze sterior walls refab Concrete lickbrick	147. 86.						
hannel plate floor BG Le 2e station wells refab Concrete lickbrick	147, 86, Dimensions total [m	2]		e	3 040 E3 Executin 22	Francis .	970 k
zhannel pilate filoor IKG 2c sterior walti "rehib Concrete licktrick Nordor walti Tehib panels (Yrong/I aay/other) of	147 86, Dimensions total [m 60,	2] _61 2]		€ 30,50 €	1.848,52 Faay VP 54	Faay	
channel pible filoor IKG 2c station walth "refab Concrete Liketrick Nordor walth refab panels (Yrong/r aay/other) <u>oof</u> oof likes	147 86 Dimensions total (m 60, Dimensions total (m 75,	2] 61 2] 89		€ 30,50 € € 48,07 €	1.848,52 Faay VP 54 3.648,34 Ceramic tiles	Faay Keramiek	3500 kj 38,6 kj
Tabane plate floor BG Lo 20 xtorio walts refab Concrete refab Cancrete refab panels foor float cont thes the formes	147 86 Dimensions total (m 60, Dimensions total (m 75, Dimensions total (m	2] 61 2] 89 2] Products [i	2	€ 30,50 € € 48,07 €	1.848,52 Faay VP 54 3.648,34 Ceramic tiles 3.751.86 Meranti (word) whether	Faay Keramiek Glas	3500 k 38,6 k 2500 k
Stanner pikle floor IKG 2c station walth Trafab Concrete Liketrick Nardo walth Trafab panels (Yrong/Law/other) Oof oof Loof tiles Vandows Jaca + from	147 86 Dimensions total (m 60, Dimensions total (m 75, Dimensions total (m 17,	2] 61 2] 89 2] Products [: 00	2	 ε 30,50 ε ε 48,07 ε ε 220,70 ε 	1.848,52 Faay VP 54 3.648,34 Ceramic tiles 3.751,86 Meranti (wood) window	Faay Keramlak Glas Hout	3500 kj 38,6 kj 2500 kj 500 kj
nonnel piate floor IGG 2c storio walib rafab Concrete Istrickrick (Yong/r asy/other) 50 60 60 61 103 + frame	147 86 Dimensions total (m 60, Dimensions total (m 75, Dimensions total (m 17, Dimensions total (m 2, c	2] 61 2] Products [: 2] Products [: 80	9 9 7	 ε 30,50 ε ε 48,07 ε ε 220,70 ε ε 595,00 ε ε 100,00 ε 	1.34352 Faxy VP 54 3.648,34 Ceramic tiles 3.751,86 Meranti (wood) window 595.00 Bossingpaneel	Faay Keramilak Glas Hout	3500 k 38,6 k 2500 k 500 k

163

11.7 APPENDIX G: OVERVIEW MATERIALS DETACHED HOUSE ANALYSIS

aba		Dwelling C: Detached Based on: Krijnen, M. (2015): Referentie	house Netherlands ewoning EPC 0,4 Vrijstaande w	oning		
		Dwelling info	ormation			
Dimensions (NVO)	[m] NVO [m] BVO	Gener	ral [m]	3] NVO [m3] BVO		
Width (BG) Width (1e) Width (2e)	6,02 6,84 6,02 6,84 4,48 5,30	BG 1e 2e		159,96 215,19 159,96 215,19 175,96 234,75		
Depth (BG)	10.22 11.00	Total		495,89 665,12		
Septh (1e) Septh (2e)	10,22 11,00 10,22 11,00				Reusability scale	
light BG light 1e	2,60 2,86 2,60 2,86				possible reusability no reusability	
light 2e light total	2,86 3,12 8,06 8,84					
Building eler	Dimensions Im21	Building materi	ial / product Dimensions [m2] Pro	duct Gradation	Reusability	
		Founda	tion			
oundation	75,24	Concrete piles Wooden piles	?	_	Dmy trekken hergebruiken?	
		Floor	75			
01660 3G	[m2] NVO [m2] BVO 61,52 75,24 61,52 75,24	Channel plate floor Channel plate floor	61,52 46,67		Dekvlaer verwijderen Dekvlaer vrowijderen	
e	61,52 75,24	Concrect (landing) Channel plate floor	14,85 61,52		In werk gestort met prefab scheidingswand	
otul	184,57 225,72					
G north	[m2] #	Exterior	walls			
/indow avity wall facade	4,50 3 26,96	Glass + frame Clickbrick	4,50 26,96	3	Product maet zelfde maat zijn	
otal	31,46	Sand-lime brick	26,96			
G south Vindow	(m2) (f 4,50 3	Glass + frame	4,50	3	Product moet zelfde moat zijn	
nndow (door) -oor avity wall facade	1,10 1 2,40 1 23,46	ulass + frame Front door Clickbrick	1,10 2,40 23.46	1	Product maet zelfde maat zijn Product maet zelfde maat zijn	
otal	31,46	Sand-lime brick	23,46			
3G west	(m2) #	Glass + frama	3.00	,	Product result tollide prost alia	
refab concrete	16,56 19,56	Prefab Concrete	3,00 16,56	-	, roauer moer zenjoe maar zijn	
G cast	[m2] #	etan dana			Desident mark of file	
refab concrete	9,12 1 10,44 19,56	Siass + Itame Prefab Concrete	9,12 10,44		roaust moet zeijne moat zijn	
e north	[m2] #					
Vindow Vindow avity wall facade	0,40 1 1,50 1 20.56	Glass + frame Glass + frame Clickbrick	0,40 1,50 29.56	1	Product maet zelfde maat zijn Product maet zelfde maat zijn	
otal	31,46	Sand-lime brick	29,56			
e south	[m2] #	Glass + feama	6.00	4	Seaduct ment selfets most vite	
avity wall facade	25,46	Clickbrick Sand-lime brick	25,46 25,46	4	Product moet zelfde mout zijn	
otal	31,46					
tawes Nindow Irefab concrete	3,00 2 16,56	Glass + frame Prefab Concrete	3,00 16,56	z	Product moet zelfde moat zijn	
rotal	19,56					
Nindow Prefab concrete	3,00 2 16,56	Glass + frame Prefab Concrete	3,00 16,56	2	Product maet zelfde maat zijn	
Fotal	19,56					
Andow Prefab concrete	1,50 1 9,17	Glass + frame Prefab Concrete	1,50 9,17	1		
Total	10,67					
Nindow Prefab concrete	1,50 1 9,17	Glass + frame Prefab Concrete	1,50 9,17	1		
rotal	10,67					
te north Nindow	[m2] # 1,68 1	Glass + frame	1,68	1		
Nindow Zavity wall facade	0,79 1 43,42	Glass + frame Roof tiles	0,79 43,42	1		
e south	45,10 [m2] #					
avity wall facade	45,10	Roof tiles	45,10			
10	[m2]#	Interior	walls			
Vall (panels) Door	39,84 7,20 3	Prefab panels Ytong (cellenbeton) / Faay Corridor door	32,64 7,20	з	Gedeeltelijk hergebruiken (uitzagen)	
Tavity wall facade	32,64 39,84					
e Vall (panels)	(m2) # 55,26	Prefab panels Ylong (cellenbelon) / Faay	43,26		Gedeeltelijk hergebruiken (uitzagen)	
avity wall facade	12,00 5 43,26	Corridor door	12,00	5		
otal	55,26 [m2] #					
Vall (panels) Door	12,78 2,40 1	Prefab panels Ytong (<i>cellenbeton</i>) / Faay Corridor door	10,38 2,40	1	Gedeeltelijk hergebruiken (uitzagen)	
avity wall facade otal	10,38 12,78					
		Building materia	l information			
Building ma	aterial (/ product)		Construction Overview Foo	costs (virgin) tors Reference	Soortelijk gewicht [kg/m3]	
Channel plate floor BG	Dimensions total [m2] 61,52 46,67		€ 36,80 € € 38.05 €	2.264,10 Channel plate floor 150n 1.775.88 Channel plate floor 200s	nm Beton	2100 kg
1e 2e	40,67 61,52		€ 38,05 €	2.341,00 Channel plate floor 200r	nm	
xterior walls refab concrete	Dimensions total [m2] 78,47		€ 109,80 €	8.615,99 Prefab Beton	Beton	2100 kg
nterior walls	105,44 Dimensions total [m2]		€ 110,42 €	11.042,33 Clickbricks	neramiek	1800 kg
refab panels (Ytong/Faay/other)	86,28		€ 30,50 €	2.631,43 Faay VP 54	Faay	970 kg
toof tiles	Dimensions total [m2] 88,52		€ 48,07 €	4.255,40 Ceramic tiles	Keramlek	3500 kg
Vindows	Dimensions total [m2]Produ	cts [#]				38,6 kg
Slass + frame	41,59	24	€ 220,70 €	9.178,81 Meranti (wood) window	Glas Hout	2500 kg 500 kg
ront deer	Dimensions total [m2] Produ	cts [#]	E 595.00 F	595.00 Bossinenannel	Hout	500 1-1
orridor door rench door	21,60 0,00	9 0	€ 100,00 € € 595,00 €	900,00 Opdekdeur Stapeldorpel 1/2	Hout	500 kg
Total	570,02 m2					

164

11.8 APPENDIX H: DISASSEMBLY FACTOR WEIGHTS AND ATTRIBUTE WEIGHTS

C	Factor	Attribute	Score
	weight		
Accessibility to	1.0	Accessible	1.0
connection		Accessible with an additional operation which causes no	0.8
		damage	
		Accessible with an additional operation which is reparable	0.6
		damage	
		Accessible with an additional operation which causes damage	0.4
		not accessible - total damage of elements	0.1
Type of	1.0	Accessory external connection or connection system	1.0
connection		Direct connection with additional fixing devices	0.8
		Direct integral connection with inserts (pin)	0.6
		Filled soft chemical connection	0.2
		Filled hard chemical connection	0.1
		Direct chemical connection	0.1
Assembly	1.0	Open linear	1.0
shape		Symmetrical overlapping	0.8
		Overlapping on one side	0.7
		Unsymmetrical overlapping	0.4
		Insert on one side	0.2
		Insert on two sides	0.1
Independency	1.0	Modular zoning	1.0
		Planned interpenetrating for different solutions (overcapacity)	0.8
		Planed for one solution	0.4
		Unplanned interpenetrating	0.2
		total dependence	0.1
Method of	1.0	Pre-made geometry	1.0
fabrication		Half standardized geometry	0.5
		Geometry made on the construction site	0.1
Assembly	1.0	Same level / Same level	1.0
Sequence		High level / Low level	0.5
		Low level / High level	0.1
Type of	1.0	One or two connections	1.0
relational		Three connections	0.6
pattern		Four connections	0.4
		Five or more connections	0.1

Table 11.9 Disassembly factor weights and attribute weights adopted from Durmisevic (2006) (van Vliet, 2018)

11.9 APPENDIX I: OVERVIEW COSTS VIRGIN BUILDING MATERIALS

aba					Construction costs Virgin building materic	ls			
Opmerking: In de vergelijking tuzsen de koo	sten zullen puur gegeken worde:	n naar de assumptie van de kosten voor d Spreifications	le bouwmateriolen. De cons	tructiekosten (in	cci montage benodigdheden en manuren) zullen voor s	owel virgin als reuse vergelijkbaar sij	in en daarom niet meegenomen in de berekenir	ıg.	
Materia: Roors	Type 285mm thick (smallest) 300mm thick 320mm thick	Thickness (mm) 285 300 320	Specs	€ / m2 €87 - €90 €90 - €93 €93 - €96	Reference https://www.offerteadviseur.nl/categorie/bouw/ve rbouwing/kosten-breedplaatvioer/	Comments Thorough estimation of total costs ; m2.	Explenation	Life cycle (years)	Reference
	340mm thick (largest) 200mm thick 200mm thick 260mm thick	340 200 200 260	<5m span 5m - 7,5m span <7,5m span	€95 - €95 € 41,34 € 43,75 € 48,54	http://dingspan.wis- studio3.wis.nt/Media/download/11180/Artikel_Bou	Only direct costs, no man hours or adjoining costs included.			
Channel plate floor	260mm thick Woningbouw	250 150 150 200	5m - 7,5m span Max span: 4,8m Max span: 5,7m Max span: 6,0m	€ 51,39 € 36,80 € 36,80 € 38,05	widtharie_teruiepangsvideren.oor		BG, based on SBR 101.0.1.02 1e / 7e, based on SBR 301.0.1.02		75 <u>https://www.nibe.info/nl/members#product-6802-102-28</u> 75 <u>https://www.nibe.info/nl/members#product-6802-102-28</u>
		200 200 260 260 260	Max span: 7,2m Max span: 8,1m Max span: 7,2m Max span: 8,1m Max span: 9,6m	€ 39,40 € 40,55 € 43,10 € 44,45 € 45,75	Linstallate, terrein/Systeenvloer, kanaelplaatvloe r/kostengegevens- Prijzen, Normen en Tarieven/2355278.htm				
Exterior walls					http://www.bouwkosten.nl/Proefsbonnement_R				
Prefab Beton	Binnenspouwbled	100	Prefab beton 100 mm	€ 109,80	amen/Utiliteitsbouw/Kostenkengetallen_21_Bui tenwanden, steen/kostengegevens- Kostenkengetallen/3986000.htm		Based on tab Recept Prefice Beton		75 https://www.nibo.info/ni/mombons#product-6094-268-27
	CS12 - F100 CS12 - C120 CS12 - E150 CS12 - E214 CS12 - E240	100 120 150 214 240	x 538 / 648 x 997 x 538 / 648 x 997	€ 24,50 € 23,25 € 30,95 € 42,70 € 43,50					
Sand-lime bricks	CS12 - E300 CS20 - E100 CS20 - E120 CS20 - E120 CS20 - E150	300 100 120 150	x 538 / 648 x 997 x 538 / 648 x 997	€ 52,85 € 26,20 € 25,15 € 33,30	http://www.bouwkosten.nl/Materiaalprijten/Bouw, _installatie,_terrein/Malkandsten,_elementen/kos	CS12 / CS20 / CS36 are different			
	CS20 - E175 CS20 - E214 CS20 - E240 CS20 - E300	175 214 240 300	X 538 / 648 X 997 X 538 / 648 X 997 X 538 / 648 X 997 X 538 / 648 X 997	€ 41,25 € 47,45 € 47,45 € 57,50	Prijzen, Normen en larieven/2433825.htm	Branario in conforcence such Bru		Dimension clickbrick Square meter	Height [mm] Width [mm] Surface [mm2] 240 100 24000 1000 1000 10000
	CS36 - EH175 CS36 - EH214 CS36 - EH250 CS36 - EH300	175 214 250 300	X 538 / 648 X 997 X 538 / 648 X 997 X 538 / 648 X 997 X 538 X 997	€ 49,75 € 62,55 € 63,35 € 79,50	Nie //www.be.wdestee.et/Materiasterij.co/Berr			Bricks / square meter	1000000 24000 41.6666667 Costs Pleces 6 7.6500 1000 € 110,42 41.6666667 41.6666667
Clickbricks	Diverse soorten	90	240mm X 100mm	€ 110,42	w, installatio, terrein/Baksteen, Click Brick/ko stendepavens: Phijzen, Normen_en_Taneven/2355256.htm	Calculation 6 / m2 clickbrick*	C		75 https://www.nibe.info/rid/members@product-8025-84-106
Prefab HSB	Buitenwanden (constructief) Binnenwanden (constructief)	HSB buitenwanden en muurplate HSB woningscheidende en	n. HBS pakket tuinkasten stabiliteitswanden	€ 347,33 € 276,57	http://www.bouw.osten.nl/yrojectanaryse/nieuw bouw_uitwerking/40_boussierectanaryse/nieuw <u>Zwalic/iostengogevens</u> Projectanalyse/234/154.htm#Kostenopniveau2Eie <u>mentendusters</u> }	Complet panels, in comparison to clickbricks and sand-lime bricks, which lacks the complete construction of the facade.			
Interior walls	Ytong G4/600 Blok	50 70 100 150	600mm X 200mm 500mm X 200mm 500mm X 200mm 600mm X 200mm	€ 14,76 € 18,37 € 76,65 € 47,72	https://www.online-	8,2 blocks / m2,€/block.	8,2		
Prefab Panels		50 70 100 100	600mm X 400mm 600mm X 400mm 600mm X 400mm 600mm X 250mm	€ 14,84 € 18,12 € 26,08 € 29,77	cellenseton/	4,1 blocks / m2, €/block. 6,6 blocks / m2, €/block.	4,1 6,6		
	raay wir on	57 57 57 57 57	2600mm X 400mm 2600mm X 400mm 2800mm X 400mm 3000mm X 400mm 3200mm X 400mm	€ 30,50 € 30,50 € 30,50 € 30,50 € 30,50	https://www.sleiderink.nl/faay- yp542child id=11598&ouantity=18&clid=CiwKCAjw knngNRIgEiwA- bZyto1eTog029J1X/UZ7MHX8Px608grl0tex6EH7rol b8ioTydDNxWIxoCNB00Ax0_EwE	Maat verschilt in afname panelen	isseed on price certainty		60 muss/www.ndo.mon/memoels#product-des/c-st-2/
Roof	Concrete tiles Ceramic tiles		Life cycle 30 years Life cycle 50 years	€ 50,00 € 50,00	https://www.dakstekker- weetjes.d/daktodekking/dakpannen/				
Roof tiles	Concrete tiles Ceramic tiles	20	Totaal	€ 56,75 € 48,07	http://www.bouwkosten.nl/Proefsbonnement_Ba men/Utilistisbouw/Kostenkengetalien_47_Dakafw crkingen/kostengegevens- Kostenkengetalien/3556003.htm		Based on 538 401.0.3.01 T1		75 https://www.nibo.info/n/mombers#product-6873-113-105
Windows	Plastic (kunststof) window Wooden window Aluminium window		Totaal (incl. manuren) incl. montage & HB++ incl. montage & HB++ incl. montage & HB++	€ 73,65 €700 - €800 €750 - €900 €800 - €950	https://www.kozlinen-weetjes.cl/ramen/				
Window	Meranti (wood) window Plastic (kunststof) window Aluminium window	115	Totaal Totaal (ink, Manuren)	€ 220,70 € 567,00 € 395,00 € 530,00	http://www.bouwkosten.n/Proefabonnement_R amen/Utiltetsbouw/Kostenkengetalien 31 Bui terwandopeningen_gevekozin/kostengegeve ms-Kostankengatalien/3866001.htm		Based on 538 201.0.3.01 Based on Tab <i>Recept Meranti Kurğ</i> ı		35 https://www.nibe.info/ri/members#product-6624-269-32
Doors Corridor door	Opdekdeur Opdekdeur	Speca Incl. montage Incl. montage	flight	€55-€75 € 100,00	https://www.werkspot.nl/binnendeur-plaatsen- vervangen/prizen-kosten		Rea listic		75 https://www.nibo.info/n/mombors#product-0125-138-33
	Achterdeur	No window Stapeldorpel 1/4 Stapeldorpel 1/7	2115mm 2315mm 2115mm 2315mm 2115mm	€ 521,00 € 606,00 € 526,00 € 643,00 € 595,00			Rased on drawing house		75
facels door	Vaardeur	4x bossingpaneel Verticale groeven	2315mm 2115mm 2315mm 2115mm	€ 688,00 € 1.315,00 € 1.330,00 € 987,00	http://www.bouwkosten.n/Materinalorijzen/Bou w. installatie, terrein/Builendeur, merbaukost				
- 9100° 0001		1x paneel Bossingpaneel en tussendorpel 1/2 Bossingpaneel en tussendorpel 1/2	2115mm 2315mm 2315mm 2315mm 2315mm	€ 1.440,00 € 1.455,00 € 1.055,00 € 1.100,00 € 1.340,00	engeoevens- Prijzen, Normen en Tarieven/4423242.htm				
		Bassingsaneel 4 ruits met roeden en bossingpanee	2315mm 2315mm 2315mm 2115mm 2315mm	€ 1.355,00 € \$09,00 € \$25,00 € 1.165,00 € 1.180,00			Standard cheapest front door		75

Table II.10 Overview costs virgin building materials (own table)



11.10 APPENDIX J: OVERVIEW DECONSTRUCTION / TRANSPORTATION COSTS REUSABLE BUILDING MATERIALS

Table II.II Overview deconstruction / transportation costs reusable building materials (own table)



11.11 APPENDIX K: OVERVIEW OF MATERIAL VALIDATION REUSABLE BUILDING MATERIALS

Table 11.12 Overview material validation reusable building materials (own table)

11.12 APPENDIX L: PROJECT-BASED MODEL INPUT

aba	Input						
Minimum and maximum amounts of dwellings per type							
	Dwelling type	A: Row-house	B: Semi-detached house	C: Detached house			
Dwellings	Minimum	10	20	12			
ľ	Maximum	40	50	60			

Amount of building material [m2] required per material and dwelling type (A,B & C) Dwelling type A: Row-house B: Semi-detached house C: Detached house

e floor 1e / 2e	67,64 113,13 110,97 34,73 145,70 52,80	74,58 146,35 147,98 86,54 234,52 60,61	108,20 169,72 78,47 105,44 183,91 86,28
/Faay/other)	113,13 110,97 34,73 145,70 52,80	146,35 147,98 86,54 234,52 60,61	169,72 78,47 105,44 183,91 86,28
/Faay/other)	110,97 34,73 145,70 52,80	147,98 86,54 234,52 60,61	78,47 105,44 183,91 86,28
/Faay/other)	34,73 145,70 52,80	86,54 234,52 60,61	105,44 183,91 86,28
/Faay/other)	145,70	23 4 ,52 60,61	183,91 86,28
/Faay/other)	52,80	60,61	86,28
	5140		
	64,48	75,89	88,52
	24,50	17,00	41,59
	1,00	1,00	1,00
	7,00	7,00	9,00
	0,00	2,00	0,00
	8,00	10,00	10,00
r	r	1,00 7,00 0,00 8,00	r 1,00 1,00 7,00 7,00 0,00 2,00 8,00 10,00

Amounts of material available per type and source (hypothetical)							
•		Source O	Source P				
Floors	Channel plate floor BG	5.000	4.000				
FIGUIS	Channel plate floor 1e / 2e	8.000	6.000				
Exterior wall	Prefab concrete	9.500	9.000				
Exterior wait	Clickbrick	10.000	8.000				
Interior wall	Panels (Ytong/Faay/other)	8.000	7.000				
Roof	Roof tiles	5.000	4.000				
Windows	Glass + frame	6.000	5.000				
	Front door	500	400				
Doors [#]	Corridor door	5.000	4.000				
	French door	600	500				
Geographical data (hypothetical)							
		Source O	Source P				
	Distance [km]	60	30				
Costs 'Reused' building materials			â				
		Source O	Source P				
Floors	Channel plate floor BG	€ 63,61	€ 62,26				
	Channel plate floor 1e / 2e	€ 67,93	€ 60,95				
Exterior wall	Prefab concrete	€ 115,21	€ 117,76				
	Clickbrick	€ 67,88	€ 73,58				
Interior wall	Panels (Ytong/Faay/other)	€ 22,16	€ 23,69				
Roof	Roof tiles	€ 28,25	€ 29,69				
Windows	Glass + frame	€ 88,67	€ 148,97				
	Front door	€ 334,21	€ 372,21				
Doors [#]	Corridor door	€ 39,01	€ 42,13				
	French door	€ 334,21	€ 372,21				

Budget € 5.000.000,00

		[eur/m	2]
Floors	Channel plate floor BG	€	36,8
	Channel plate floor 1e / 2e	€	38,0
Exterior wall	Prefab concrete	€	109,8
	Clickbrick	€	110,4
Interior wall	Panels (Ytong/Faay/other)	€	30,5
Roof	Roof tiles	€	48,0
Windows	Glass + frame	€	220,7
Doors [#]	Front door	€	595,0
	Corridor door	€	100,0
	French door	F	

	Total	€	31.679	€	43.016	€	44.200
	French door	€	-	€	1.190	€	-
	Corridor door	€	700	€	700	€	900
Doors [#]	Front door	€	595	€	595	€	595
Windows	Glass + frame	€	5.407	€	3.752	€	9.179
Roof	Roof tiles	€	3.100	€	3.648	€	4.255
Interior wall	Panels (Ytong/Faay/other)	€	1.610	€	1.849	€	2.631
	Clickbrick	€	3.835	€	9.555	€	11.642
Exterior wall	Prefab concrete	€	12.184	€	16.248	€	8.615
	Channel plate floor 1e / 2e	€	2.574	€	2.838	€	4.117
Floors	Channel plate floor BG	€	1.674	€	2.641	€	2.264
	Material	A: I	Row-house	B: Sem	i-detached house	C: D	etached house
Costs per dwelling type (Virgin	materials)						

Table 11.13 Input page project-based model (own table)

		input	Dwelling type A	Dwelling type B	Dwelling type C
alba		Minimum amount of dwellings	10	20	12
		Maximum amount of dwellings	40	50	60
		Budget	€ 5.000.000,00		
		Fourses	(aumaa 0	Course D	1
		Sources	Source U	30	(hypothetical)
			00	50	
		Construction year	Source O [1989]	Source P [1997]	(hypothetical)
		t [years]	30	22	
		Amounts of material available per type and source [m2]	Source O	Source D	(h
		Channel also floor PG	5 000	4 000	(nypotnetical)
Fla	Floor	S Channel plate filoso 16 (2)	8,000	6.000	-
		Channel plate floor 167 26	0.000	0.000	-
	Exterior wa	Pretab concrete	9.500	9.000	-
		Clickbrick	10.000	8.000	-
	Interior wa	Panels (Ytong/Faay/other)	8.000	7.000	-
	Roo	f Roof tiles	5.000	4.000	
	Window	S Glass + frame	6.000	5.000	
		Front door	500	400	
	Doors [#	Corridor door	5.000	4.000]
		French door	600	500]

Table 11.14 Solve page input Project-based model (own table)

11.13

APPENDIX M: PROJECT-BASED MODEL OUTPUT

Output Dwelling type Optimal A: Row-house B: Semi-detached house C: Detached house 10 50 52

Amounts of material used p	er type and dwelling type			
	Dwelling type	A: Row-house	B: Semi-detached house	C: Detached house
Floors	Channel plate floor BG	455	3.588	3.170
FIGORS	Channel plate floor 1e / 2e	676	3.729	5.574
Exterior wall	Prefab concrete	1.110	7.399	4.043
	Clickbrick	347	4.327	5.432
Interior wall	Panels (Ytong/Faay/other)	528	3.030	4.445
Roof	Roof tiles	645	3.795	4.561
Windows	Glass + frame	245	850	2.143
	Front door	10	50	52
Doors [#]	Corridor door	70	350	464
	French door	0	100	0

Amounts of Reused building material used per type and source

	Used	Source O	Source P
floor	Channel plate floor BG	3.213	4.000
HOOIS	Channel plate floor 1e / 2e	3.980	6.000
Eutonion well	Prefab concrete	9.500	3.052
Exterior wall	Clickbrick	10.000	106
Interior wall	Panels (Ytong/Faay/other)	8.000	3
Roof	Roof tiles	5.000	4.000
Windows	Glass + frame	3.238	0
	Front door	112	0
Doors [#]	Corridor door	884	0
	French door	100	0

Co

Dwelling type configuration

Dwellings

	Used	Source O)	Source P		Total	
Floors	Channel plate floor BG	€	204.387	€	249.059	€	453.446
HOORS	Channel plate floor 1e / 2e	€	270.354	€	365.730	€	636.084
Eutoxian wall	Prefab concrete	€	1.094.532	€	359.350	€	1.453.882
Exterior wait	Clickbrick	€	678.777	€	7.834	€	686.612
Interior wall	Panels (Ytong/Faay/other)	€	177.285	€	79	€	177.364
Roof	Roof tiles	€	141.263	€	118.761	€	260.025
Windows	Glass + frame	€	287.099	€	-	€	287.099
	Front door	€	37.272	€	-	€	37.272
Doors [#]	Corridor door	€	34.476	€	-	€	34.476
	French door	€	33.421	€	-	€	33.421

osts 'Virgin' building materials							
	Used	Total					
Flager	Channel plate floor BG	€	265.445				
FIGUES	Channel plate floor 1e / 2e	€	379.736				
Exterior wall	Prefab concrete	€	1.378.151				
	Clickbrick	€	1.115.924				
Interior wall	Panels (Ytong/Faay/other)	€	244.102				
Roof	Roof tiles	€	432.655				
Windows	Glass + frame	€	714.561				
	Front door	€	66.355				
Doors [#]	Corridor door	€	88.369				
	French door	€	59.500				

Comparison costs betwe ised / virgin build Channel plate floor BG Channel plate floor 1e / 2e Prefab concrete Clickbrick Panels (Ytong/Faay/other) Roof tiles Glass + frame Front door Corridor door French door 265.445 379.736 1.378.151 1.115.924 244.102 432.655 714.561 66.355 88.369 59.500 Virgi Vir 453.446 € 636.084 € 1.453.882 € 686.612 € 177.364 € 287.099 € 37.272 € 34.476 € 33.421 € Floors Exterior wall Interior wall Roof Windows Doors [#] Total

4.744.797 € 4.059.681 €

Comparison costs between reused / virgin building materials (/ m2) (source O)								
		Re-use		Virgin				
Floors	Channel plate floor BG	€	63,61	€	36,80			
noors	Channel plate floor 1e / 2e	€	67,93	€	38,05			
Free stars and	Prefab concrete	€	115,21	€	109,80			
Exterior wait	Clickbrick	€	67,88	€	110,42			
Interior wall	Panels (Ytong/Faay/other)	€	22,16	€	30,50			
Roof	Roof tiles	€	28,25	€	48,07			
Windows	Glass + frame	€	88,67	€	220,70			
	Front door	€	334,21	€	595,00			
Doors [#]	Corridor door	€	39,01	€	100,00			
	French door	€	334,21	€	595,00			
Comparison costs betwee	en reused / virgin building mate	rials (/ m2) (source P)				
		Re-use		Virgin				
Floore	Channel plate floor BG	€	62,26	€	36,80			
FIGUIS	Channel plate floor 1e / 2e	€	60,95	€	38,05			
Frank and a small	Prefab concrete	€	117,76	€	109,80			
Exterior wait	Clickbrick	€	73,58	€	110,42			
Interior wall	Panels (Ytong/Faay/other)	€	23,69	€	30,50			
Roof	Roof tiles	€	29,69	€	48,07			
Windows	Windows Glass + frame		148,97	€	220,70			
	Front door	€	372,21	€	595,00			
Doors [#]	Corridor door	€	42,13	€	100,00			
	French door	£	372.21	£	595.00			

 Table 11.15 Project-based model output (own table)



Table 11.16 Project-based model output Solve tab (own table)

4	en,		EALER EALER	4	En,		INT LEFT	4	Pin,		her affel
C una	ALE AL	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		a ma	Cores	11-1		CHEN	Direct of	A B C B B A	
1	AA	A FR	and the second second	C.	A		a transfer	N	A		and the second sec
03 03	and state	C.F.		20 5 T		C.F.	and the second se	27	the second	C.F.	BTB
TING TR				N A MARTIN		C B F		Hermitten.			
-		L and		the state			B	and the second s			
	P'A,		IN CHER	4	P N.		are pipel		P'A,		IN BIR
Sana	a starter			Cires.	No. of the second se			(inter	A REAL PROPERTY		
S.S.	9		A COLUMN	See.	6			See .	6 9	AL PRO	
pr a	and some first	· ·		211 E				an Ar	isting		
TI MILLA			۲	Wirman Ta			۲	A statute			
		AL.				all a		A trai		ALL A	
	AND APPEN	4	En,		ANY SPICE		e'x,		AT LAPE		P'A,
		() I I I	Ser.				No. of Concession, Name	Ser Print		(* 103)	a service
AL BAR		S	A	AL DER	in Province	- See	A A	Si pre		- Sta	AA
A.		and the	the second		B	Bu Bu		A STATE		En En	and starts
		traine.			۲	WII IN DI			۲	N in the	
Ques		A.							B	all a	
			P'A,		Are pitel		×14.		AN CHER		PIN,
	a	oa	No.	2. 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		C mar	New York	14		5° 17 138	Carles
	conc	epts.	6	A Free	and State and	N	6 4		JDe	əlft	A
and the second sec	E STA	27 27	tion seens			na Bu	tion state			are con	the state